# PROPOSED MEMORANDUM OF SETTLEMENT BETWEEN

## ALBERTA UNION OF PROVINCIAL EMPLOYEES

AND

## STRATHCONA COUNTY May 11, 2023

The attached documents constitute a full settlement of all issues and proposals remaining outstanding in the current round of negotiations. This proposed Memorandum of Settlement combined with the previously signed off Articles, and Letters of Understanding will form a Tentative Agreement between the parties, which will be subject to the normal ratification process of each party.

The Parties agree to recommend to their respective principals the terms and conditions contained in the Tentative Agreement.

Signed at Sherwood Park, this \_\_\_\_\_ day of May, 2023.

FOR STRATHCONA COUNTY

Swight

FOR AUPE LOCAL 118

Dr.

A.

## PROPOSED MEMORANDUM OF SETTLEMENT BETWEEN

#### ALBERTA UNION OF PROVINCIAL EMPLOYEES

#### AND

## STRATHCONA COUNTY May 11, 2023

The attached documents constitute a full settlement of all issues and proposals remaining outstanding in the current round of negotiations. This proposed Memorandum of Settlement combined with the previously signed off Articles, and Letters of Understanding will form a Tentative Agreement between the parties, which will be subject to the normal ratification process of each party.

The Parties agree to recommend to their respective principals the terms and conditions contained in the Tentative Agreement.

Signed at Sherwood Park, this	day of May, 2023.
FOR STRATHCONA COUNTY	FOR AUPE LOCAL 118
Slisha	
Sharla Mishna	

THIS AGREEMENT is made this 18th day of April, 2018, A.D. pursuant to the Labour Relations Code of Alberta (2000) as amended.

This Agreement is entered into by and between the Strathcona County, municipal corporation covered by the Collective Agreement (hereinafter referred to as "the County" or "the Employer")

and the Alberta Union of Provincial Employees Local 118/009 (hereinafter referred to as "the Union").

Whereas, the above parties are mutually desirous of entering into a Collective Agreement with the intent and purpose of:

- A. Promoting a harmonious relationship;
- B. Providing for conditions of employment, hours of work and rates of pay;
- C. Establishing the means for prompt and equitable settlement of disputes or grievances between the County and Employees represented by the Union, and;
- D. Encouraging efficiency in operation.

Now therefore, the parties hereto mutually agree as follows:

#### **ARTICLE 1 – DEFINITIONS**

- 1.01 In this Agreement:
  - (a) A word used in the singular applies also in the plural. A word used in the masculine applies also in the feminine.
  - (b) Employee: means a person employed by the Strathcona County in the Public Works Operations, Vehicle and Equipment Services, Transit Services, Materials Management, Water and Wastewater Operations, and as listed in Schedule A.
    - (i) Regular Employees: are Employees who occupy established positions and who have successfully completed the probationary period and work the fully scheduled hours.
    - (ii) Part-time Employees: are Employees who occupy established positions and have successfully completed the probationary period and work less than the fully scheduled hours of Regular Employees.
    - Temporary Employees: are Employees who are indefinite term (iii) Employees or Employees for specific projects, seasonal employment, replacements for Regular Employees who are on illness leave, General Illness, Long Term Disability, Workers' Compensation, Maternity Leave, Parental Leave and Adoption Leave, or other long term leaves of absence. Indefinite term Employees upon completion of more than fifteen (15) months of continuous service will be deemed to be Regular Employees, except if they are covering for an Employee on Long Term Disability, are moving from one Department to another Department for a temporary or seasonal position, or are covering off consecutive leaves, as referenced above. "Continuous service", for the purpose of this clause, means any period of employment that does not include a break in service of more than thirty (30) calendar days.
    - (iv) Apprenticed Employees: are Employees who are registered as Apprentices under the Apprenticeship and Industry Training Act.
    - (v) Probationary Employee: means an Employee who during his initial

- period of employment occupies an established position during a probationary period.
- (c) Grade: means the step assigned to a class within the pay grid.
- (d) Union Steward: means a person who is selected by the members of the Union to act on behalf of those Employees as empowered by the Union.
- (e) Union Representative: means a person authorized by the Union to act on behalf of Employees and includes a Union Steward, elected or appointed officer of the Union, or staff representative.
- (f) Liaison Committee: is the joint Liaison Committee established by the two parties as stipulated within this Agreement.
- (f) Casual Illness: is any illness which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less.
- (g) General Illness: is any illness which causes an Employee to be absent from work for a period of more than three (3) consecutive work days.
- (h) Layoff: includes a reduction in the work force as a result of shortage of work.
- (i) Branch: means
  - (i) Public Works Operations
  - (ii) Vehicle and Equipment Services
  - (iii) Transit Services
  - (iv) Water and Wastewater Services
  - (v) Materials Management
- (j) Classification: means each line in Schedule A.
- (k) Seniority: commences at date of hire, and shall include all continuous **and accumulated service** as a temporary Employee. provided there was no break in service greater than thirty (30) days.
- (l) Day: shall mean calendar day.
- (m) Continuous Service: any period of employment that does not include a break in service of more than thirty (30) calendar days.
- (n) Accumulated Service: Total number of regular hours worked, with no break in service of more than six (6) months.

#### **ARTICLE 2 – JURISDICTION**

- 2.01 During its term, this Agreement shall be applicable to any person employed in the Public Works Operations, Vehicle and Equipment Services, Transit Services, Materials Management and Water and Wastewater Services of Strathcona County, who are described in Schedule A attached hereto, or to positions that fall within the jurisdiction of the Union.
- 2.02 When additional work is required by the Employer, individuals who have paid Union dues to the Alberta Union of Provincial Employees, Local 118, Chapter 009 in the past twelve (12) months, will be called first and, if available, offered the additional work.

#### **ARTICLE 4 - APPLICATION**

- 4.01 Regular Employees shall be granted all the terms and conditions of this Agreement.
- 4.02 Part-time Employees shall be granted all the terms and conditions of this Agreement except those conditions which require a particular length of service, such benefits shall be granted on a pro-rata basis in relation to hours of work of full-time employment.
- 4.03 Apprenticed Employees as defined within Apprenticeship and Industry Training Act shall be granted all the terms and conditions of this Agreement.
- 4.04 Temporary Employees shall be entitled to the terms and conditions of this Agreement, except where Regular Employees are referenced. Seniority for employment periods of less than six (6) months shall not be accumulated.

#### **ARTICLE 6 - UNION RECOGNITION**

- 6.01 The County recognizes the Union as the sole bargaining agent of the Employees covered by this Agreement.
- No Union Employee shall be required or permitted to make any written or verbal agreement which may be in conflict with the terms of this Collective Agreement.
- 6.03 New Employees

The Employer shall supply all new Employees with a copy of the Collective Agreement.

#### 6.04 <u>Monthly Staffing Reports</u>

The County recognizes the need of the Union to have required information concerning Employees eligible for membership in the Union and hereby agrees to submit between the sixteenth (16th) day and the last day of each month, the name, address, and position filled by any new Employees assigned to Public Works Operations, Vehicle and Equipment Services, Materials Management, Water and Wastewater Operations and eligible for membership in the Union.

The Employer recognizes the importance of new Union Employees receiving an orientation to the Union. To ensure this occurs, the Chapter Executive will be given a list of all newly hired Employees and a representative of the Union shall, within thirty (30) calendar days of the Employees' date of hire, make a presentation to the Employee(s) with respect to the structure of the Union as well as the rights, responsibilities and benefits of membership in the Union.

The Employer will generally, provide the Union with one (1) weeks' advance notice of the Employee's scheduled orientation.

#### 6.04 **No Discrimination**

The parties agree that there shall be no discrimination or coercion exercised or practiced with respect to any Employee for reason of function, membership or legitimate activity in the Union.

The Employer shall exercise its rights in a fair and reasonable manner, nor shall these rights be used in a manner which would deprive any Employee of his

employment, except for just cause.

#### 6.05 **Bulletin Boards**

The County recognizes the right and need of the Union to be able to post notices of meetings and other legitimate Union activities at various sites wherein members of the Union are assigned or employed.

#### 6.06 <u>Seniority Lists</u>

County shall maintain a list showing the date upon which each Employee's service commenced. The County shall forthwith provide the Union, upon request, with an up-to-date list and shall, in any event, post an up-to-date list on all bulletin boards where the Employees work on July 1st and January 1st in each year.

#### 6.07 <u>Union Officers Stewards</u>

The County recognizes the right of the Union to appoint "Employees" as Union Stewards who are to be recognized as official representatives of the Union.

#### 6.08 <u>Union Membership</u>

Each person who becomes employed in positions covered by this Agreement shall automatically become a member of the Union as a condition of employment.

#### 6.09 **Monthly Staffing Reports**

- (a) The Employer agrees to remit to the central office of the Union, the amounts equal to the dues that have been deducted from the pay of all Employees by the first working day after the fifteenth (15th) calendar day in the following month. Where an accounting adjustment is necessary to correct an over or under payment of dues, it shall be effected in the succeeding month.
- (b) The County recognizes the need of the Union to have required information concerning Employees eligible for membership in the Union and hereby agrees to submit on a monthly basis between the sixteenth (16th) day and the last day of each month, the following information: the name, address, phone number, status (regular or temporary), Identification number, classification, starting date and position filled by any new Employees In the bargaining unit. assigned to Public Works Operations, Vehicle and Equipment Services, Materials Management, Water and Wastewater Operations and eligible for membership in the Union.

The Employer shall provide the Union with a monthly list identifying each Employee, including new or terminated Employees. The list will include Employee name, address, identification number, starting date, classification, and dues deducted for that calendar year.

(c) The Employer recognizes the importance of new Union Employees receiving an orientation to the Union. To ensure this occurs, the Chapter Executive will be given a list of all newly hired Employees and a representative of the Union shall, within thirty (30) calendar days of the Employees' date of hire, make a presentation to the Employee(s) with respect to the structure of the Union as well as the rights, responsibilities and benefits of membership in the Union.

#### ARTICLE 8 - JOB DESCRIPTIONS CLASSIFICATIONS & RECLASSIFICATIONS

8.01 The Employer shall make the Employee's job description available.

#### 8.02 New or Amended Classifications

The County Administration shall give written notice to the Union of the establishment of any new or amended job descriptions and/or salary classification levels (Schedule A). If it is established that a higher salary level is appropriate, the new salary shall be retroactive to the date of the classification was created. Where Union and management representatives through the Liaison Committee are unable to agree within thirty (30) days on either the correct classification of a revised job description or on the compensation level of a new job description, the matter may be advanced to arbitration for the establishment of the compensation for the individuals for which the job description has changed. If a wage change results, the effective date of change shall be determined by the Arbitrator. Both the Employer and the Union will share the cost of the Arbitrator equally. The Arbitrator will be guided by:

- (a) Evidence and submissions of the parties;
- (b) The salary levels in the Collective Agreement;
- (c) The complexity, working conditions and knowledge requirements of the other positions in the Collective Agreement at the time that the new position is established.

However, the Arbitrator shall not be restricted to the existing salary levels.

#### 8.02 Classification Reviews

An Employee may at any time bring to management for review, changes that may have occurred to his existing job duties. The Employee/Union, may bring such issues forward to Liaison Committee.

#### 8.03 <u>Reclassification</u>

When it is necessary to re-classify Regular Employees, the County may re-classify the affected Employees on the basis of Branch seniority, qualifications and ability.

- (a) An Employee to be re-classified may displace any Employee in a lower classification within that Branch that has less Branch seniority providing that the Employee being reclassified is qualified to perform the duties of that position;
- (b) Employees with special qualifications may be exempt from reclassification.
- 8.04-12.02 In those cases where re-classification according to Clause 12.01 8.03 have occurred and vacancies become available in the affected job classification, those Employees who were previously re-classified downward from that classification, shall be returned to that classification on the basis of their seniority so long as that Employee has maintained the ability and qualifications to be appointed to the position.
- 8.05 12.03 An apprentice who begins and completes their apprenticeship with the Employer shall be placed on level 2 of the appropriate trades rate upon confirmation of certification.

#### **ARTICLE 9 - POSTING OF POSITIONS, PROMOTIONS AND TRANSFERS**

- 9.01 When a position of a Regular or Temporary nature is to be filled, notice of the vacancy shall be posted as soon as reasonably possible for a minimum of one (1) week. The position can be filled by appointment if it is to be filled for a period of less than ninety (90) calendar days.
- 9.02 If a position with the same job code becomes vacant within six (6) months of the original closing date, the Employer may consider candidates from the original recruitment process, without re-posting. This is not applicable if the original posting was for a temporary position and the vacancy is for a permanent position.
- 9.02 3 Such notice will contain the following information: **date of the posting,** nature of position, qualifications, required knowledge and education, skills, shift, hours of work, experience, wage rate or range, consistent with the job description.
- 9.03 4 Where there are qualified Union candidates for a posted position, their applications shall be considered first.
- 9.04.5 Where qualified applicants have applied and a suitable candidate has been selected, the Employer shall appoint a candidate to the position within thirty (30) days of the competition closure.
- 9.05 6 Preference in the filling of positions shall be given to Employees of the County over all other persons, where the Employee has the required skills and ability to perform the tasks of the position. When filling positions or making promotions, qualifications, experience, skills, ability and seniority shall be the governing factors. When filling positions, employees of the County shall be given first preference.

#### 9.06 7 Trial Period

An Employee who is transferred or promoted to another position within this Agreement shall be confirmed in that other position within three (3) months unless the Employee wishes otherwise or proves unsatisfactory in the new position, in which case the Employee will be returned to his former position. Any other Employee promoted or transferred because of the displacement of positions shall also be returned to his former position. In any event, Employees promoted or transferred shall be paid at the rate applicable to the new position from commencement of employment in the position until returned to the previous position.

- (b) In making promotions and filling vacancies, appointments will be made on the basis of qualifications, experience, ability and seniority.
- 9.08 The posting requirements in this Article may be waived by mutual agreement in cases of medical accommodation.

#### **ARTICLE 10 – SENIORITY**

10.01 For the purpose of Clauses 10.01 to 10.03 **10.10**, 12.01 to 12.03 **12.02**, and 15.01 to 15.06 **15.05**, seniority shall be defined from the date of continuous **service** employment in any Branch covered by this Agreement.

In accordance with 1.01(k) seniority commences at date of hire, and shall include all continuous and accumulated service as a temporary Employee.

10.02(a) Employees who were employed by the County prior to January 1, 1979 and

entered in any classification covered by this Agreement on that date shall have their seniority established as the date of continuous **service** employment with the County.

Seniority shall be retained and accumulated during absences resulting from illness or an accident where the Employee is entitled to compensation, suspension with or without pay less than sixty-one (61) days, approved leave without pay less than sixty-one (61) days.

Seniority shall be retained during absences resulting from; layoff (up to eighteen (18) months), suspension with or without pay greater than sixty (60) days, approved leave without pay greater than sixty (60) days, strike or lockout.

10.045 Where Employees have been laid off and recalled, they shall retain the seniority held at the date of layoff. The Union shall be notified of all recalls.

10.056(d) An Employee who accepts an internal permanent position outside this Agreement shall retain the seniority accrued to that time for a period of six (6) months.

10.0210.06 Accumulation of seniority shall recommence from the date the Employee returns to a classification covered by this Agreement and their seniority date shall be adjusted by the period of their absence. This retained seniority shall not be effective in exercising any seniority rights until the Employee has been back in the scope of this Agreement for six (6) months.

An Employee shall lose their seniority and their name shall be removed from the seniority list for any of the following reasons: voluntary resignation, discharge for cause, layoff for more than eighteen (18) months, retirement, or when an Employee has been employed in a position outside this Agreement for a period of six (6) consecutive months **or more**.

#### **ARTICLE 11 - PERIOD OF PROBATION**

11.01 The probationary period for new Employees covered by this agreement in both permanent and temporary positions shall be five hundred (500) hours one thousand (1000) hours worked. , with the exception of Employees appointed to the positions listed below, whose probationary period will be one thousand (1000) hours worked:

3008	Public Works Technician	<del>3015</del>	<del>Operator I</del>
3009	Public Works Technician II	<del>3016</del>	<del>Operator II</del>
<del>3012</del>	Truck Driver I	<del>3017</del>	<del>Operator III</del>
<del>3013</del>	Truck Driver II	<del>3019</del>	Operations Foreman
3014	Truck Driver III		

<del>3025</del>	<del>Partsman I</del>	3030	<del>V&amp;E Chargehand</del>
<del>3026</del>	<del>Partsman II</del>	<del>3040</del>	<del>V&amp;E Leadhand</del>
<del>3027</del>	Welder	<del>3041</del>	Transit Chargehand
<del>3062</del>	Fleet Services Technician	<del>3042</del>	Transit Leadhand

<del>3021</del>	<del>W&amp;WW Foreman</del>	<del>3052</del>	Utilities Electrician
<del>3038</del>	W&WW Leadhand	<del>3053</del>	Utilities Instrument Technician
<del>3047</del>	Infrastructure Foreman	<del>3051</del>	Ut.Inst Tech/Electrician Dual
<del>3039</del>	Project Foreman	<del>3063</del>	Control Systems Technician
3043	Utility Operator I	<del>3020</del>	W&WW Operator (Collection and Distribution)
3044	Utility Operator II	<del>3049</del>	W&WW Operator (Transmission)
3045	Utility Operator III	3037	Wastewater Infrastructure Operator
<del>3046</del>	Utility Leadhand		

If the Employer identifies performance concerns, written feedback shall be provided to the Employee. Where circumstances warrant an extension, the probationary period may be extended up to an additional two hundred and fifty hours (250) hours upon mutual agreement of the Union and the Employer. may be extended up to an additional 500 hours worked, at the Employer's discretion. The Employee and the Union shall be notified of the extension and the cause for the extension.

- An Employee who has previously been employed in a position covered by this agreement will have such previous employment considered part of the probationary period provided that a break in service is one (1) year or less and the Employee is returning to the same job.
- 11.03 At any time, an Employee, categorized as probationary, may be released by the Department Head or the County Administration. Such notice of release will be formalized in writing and will give the reason for the Employee being released as soon as practicable following at the time of the notification of the release.
- 11.04 Employees shall provide at least ten (10) working days written notice of their intention to resign.
- In the event that an Employee terminates his employment with the County without submitting the written notice in compliance with Clause 11.04, the Employee shall not be entitled to receive his regular cheque and/or his final cheque until the next regular distribution of regular wage and salary cheques.
- 11.06 4 A Probationary Employee in accordance with Articles 11.01 shall be granted all of the terms and conditions of this Agreement. During the probationary period the County may terminate the employee. The employee or the Union may grieve the termination, but the decision at Level 3 shall be final and binding

and the matter shall not be referred to arbitration under the provisions of this Agreement.

however the termination of a Probationary Employee shall not be have access to the application of any grievance procedure for termination of employment as a result of:

- (a) Unsatisfactory probationary performance; or
- (b) The abolishment of the position occupied by the Probationary Employee

#### **ARTICLE 12 – RECLASSIFICATION**

- 12.01 When it is necessary to re-classify Regular Employees, the County may re-classify the affected Employees on the basis of Branch seniority, qualifications and ability.
  - (a) An Employee to be re-classified may displace any Employee in a lower classification within that Branch that has less Branch seniority providing that the Employee being reclassified is qualified to perform the duties of that position;
  - (b) Employees with special qualifications may be exempt from reclassification.
- In those cases where re-classification according to Clause 12.01 have occurred and vacancies become available in the affected job classification, those Employees who were previously re-classified downward from that classification, shall be returned to that classification on the basis of their seniority so long as that Employee has maintained the ability and qualifications to be appointed to the position.
- 12.03 An apprentice who begins and completes their apprenticeship with the Employer shall be placed on level 2 of the appropriate trades rate upon confirmation of certification.

#### **ARTICLE 14 - GRIEVANCES**

#### 14.01 **Statement of Intent**

The Employer and the Union recognize the desirability of resolving differences through joint consultation and **respectful and open** discussion. Both will attempt to resolve differences through **exploring interests and options together informally** means where possible, prior to proceeding to formal **process**.

#### 14.02 <u>Definitions and Scope</u>

- (a) A Personal grievance is a dispute regarding:
  - (i) Alleged unjust treatment or discrimination
  - (ii) Alleged unfair working conditions
  - (iii) Any disciplinary action
  - (iv) Any concern regarding the interpretation, application, operation or

alleged violation of this Agreement, including any questions as to the arbitral nature of the matter.

- (b) Grievances which involve a dismissal or a suspension will commence at level II and shall be initiated in writing within fourteen (14) days of the date when the Employee first became aware of the disciplinary action.
- (c) A Policy Grievance is a dispute between the parties which due to its nature is not properly subject of a Personal Grievance and concerns the interpretation, application, operation or alleged violation of this Agreement, including any questions as to the arbitral nature of the matter.
- (d) Formal personal grievances shall be initiated within fourteen (14) calendar days from the date of the incident prompting the grievance. Policy grievances shall be initiated either by the Union or the County in writing, within thirty (30) calendar days from the day that the matter giving rise to the grievance came to the attention of the Designated Officer of the Union or Official of the Employer.
- (d) Formal grievances shall be in writing (email will suffice) and must state the general nature of the grievance, identify the specific clauses in the collective agreement that are being violated, and provide details with respect to the redress or remedy being sought.
- (f) "Days" means calendar days.
- (g) A Union member submitting a grievance has the right to be assisted by an attending Union Representative or Union Steward.

#### 14.03 Grievance Process

A Personal Grievance shall be initiated by the Employee as follows:

#### Level I

The grievance shall be taken up with the Employee's manager or designate. an Employee wishes to pursue a Personal Grievance, the Employee and/or Union Representative shall advise the Employee's manager or designate that they wish to initiate a Level I grievance. The Employee and/or the Union representative will discuss the subject of the proposed grievance with the Employee's manager or designate within fourteen (14) calendar days of the receipt of the grievance. date upon which the subject of the grievance, in an attempt to resolve the matter. The manager or designate first level classified supervisor will provide a response to the Employee and/or Union Representative within seven (7) calendar days from when the proposed grievance was discussed.

#### Level II

If the Employee continues to be dissatisfied, he should submit the grievance in writing within fourteen (14) days of receiving the response from Level I, to the Department Director. The grievance shall state the general nature of the grievance, the Articles in the Collective Agreement alleged to have been breached and the redress or the remedy sought. The Director, in consultation with Human Resources, will hear the matter and will respond in writing to the Grievor within seven (7) calendar days.

#### Level III

If the Union does not accept the decision of the Department Director, they

may request in writing within seven (7) days of receiving the Department Directors decision, that the Chief Commissioner review the grievance. The Chief Commissioner shall hear the matter within ten (10) days and will advise the Union of his decision in writing within fourteen (14) days of the hearing;

The Chief Commissioner may designate an Associate Commissioner to hear and decide on Level III grievances;

If the grievance is not settled as a result of the Chief Commissioner's decision, the Union **or the Employer** may refer the matter to arbitration pursuant to the Labour Relations Code.

#### 14.04 **Arbitration Board**

In the event the matter being grieved is presented to an Arbitration Board, each Party shall bear the expense of its respective nominee and shall bear one-half (1/2) of the expenses of the Chairman of the Arbitration Board.

- 14.05 The Parties may mutually agree to have the matter heard by a single Arbitrator.
- 14.06 The Arbitration Board by its decision shall not alter, amend or otherwise change the terms of this Collective Agreement.
- 14.07 All Union members submitting a grievance have the right to be assisted by an attending Union Representative.

#### 14.08 Procedure and Time Limits

Extension of Time Limits: Time limitations set out in the above grievance procedure may be extended by mutual agreement between the parties and must be in writing.

Time Limits: Time limits and procedures contained in this grievance procedure are mandatory. Failure to pursue a grievance within the prescribed time limits and in accordance with the prescribed procedures shall result in abandonment of the grievance and all rights and recourse to the grievor.

Failure to reply to a grievance in a timely fashion shall advance the grievance to the next level. Grievances so advanced shall be subject to time limits as if a reply had been made on the last allowable day of the preceding level in the procedure.

- Policy Grievances may be initiated either by the Union or the County in the following manner:
  - (a) Policy Grievances shall be initiated either by the Union or the County in writing, within thirty (30) calendar days from the day that the matter giving rise to the grievance came to the attention of the Designated Officer of the Union or Official of the Employer. The written notice of the grievance shall be submitted to the President of the Union or the Director, Human–Resources of Strathcona County and shall stipulate the general particulars of the grievance together with reference to the allegedly violated clause of the Agreement;
  - (b) Grievances submitted in accordance with Clause 14.09 shall be referred to a Grievance Committee which shall endeavour to resolve the difference. This Committee will be composed of two representatives of the County and two members of the Union appointed by the Executive. This Committee shall meet within fifteen (15) calendar days of the submission of the written notice by either party;

- (c) After the parties have met in accordance with the preceding clause, each party will give any decision made in writing to the other party within fifteen (15) calendar days. In the event that the two parties reach a mutually acceptable agreement concerning the matter in question, a written and signed notice of mutual agreement will supersede the aforementioned notice of decision;
- (d) In the event that the parties do not reach a mutually acceptable agreement, or that either party does not accept the written notice of decision of the opposite party within thirty (30) calendar days of the Chief Commissioner's decision as per Clause 14.03 either party may notify the other party in writing of its intention to submit the grievance to arbitration pursuant to the provision of the Labour Relations Code.

#### ARTICLE 16 – HOURS OF WORK AND OVERTIME

16.01 The prescribed regular hours of work for an Employee, other than those referred to in Clause 16.04-5 following, shall be eight (8) hours in any one (1) day or forty (40) hours in any one (1) calendar week or five (5) working day period. Hours worked in excess of the above shall be calculated as overtime hours and shall be paid accordingly. Notwithstanding the above, when an established schedule provides a pattern that is irregular, the forty (40) hours in one (1) calendar week or five (5) working day period shall mean average over a two (2) week period.

#### 16.02 **Overtime**

The overtime pay rates shall be calculated at the rate of two (2.0) times the regular pay rate for all hours worked beyond eight (8) hours in a day.

- An Employee may bank overtime hours at the rate it is earned; up to the maximum of forty (40) regular hours within a calendar year. An Employee may request to be paid out for banked hours up to the time equivalent of forty (40) regular hours per calendar year. Banked hours may be taken as time off by Employees after receiving written authorization from the County designated official who shall consider the wishes and application of the Employee. Up to forty (40) banked regular hours may be carried forward into the next calendar year. Banked hours cannot exceed the time equivalent of eighty (80) regular hours at any one time.
- 16.04 For those Employees who are engaged in road construction, road maintenance or the removal of snow from roads or any of those undertakings, for the purpose of calculating overtime, their regular day shall consist of a twenty-four (24) hour period beginning at midnight. The maximum number of hours of work at regular rates of pay shall be those prescribed or allowed under the Employment Standards Code and Regulations thereunder, but shall not exceed:
  - (a) Ninety (90) hours bi-weekly, except where permitted under the Employment Standard Code;
  - (b) Eighty (80) hours bi-weekly during the period November 01 to April 30, inclusive;
- 16.05 Only hours actually worked, paid vacation (including general holidays) and banked time used to supplement hours are used in accumulation of overtime. Sick leave or other paid or unpaid absences are not included.
- 16.06 For those Employees specified in Clause 16.04 from May 1 to October 31, overtime rates shall be calculated at the rate of two (2.0) times the regular pay rate for all

hours worked beyond nine (9) hours in a day. This article only comes into effect when actually scheduled to work nine (9) hours a day.

#### 16.06 **Posting of Shift Schedules**

Barring unforeseen circumstances and emergency situations, shift schedules shall be posted no less than thirty (30) days before the expiry of the current schedule.

#### 16.07 **Shift Changes**

- (a) When an Employee is required to change from one shift to another shift, the Employer will provide at least twenty-four (24) hours written notice **in advance of the start time of the originally scheduled shift** and ten (10) hours off between shifts.
- (b) If the twenty-four (24) hours written notice and ten (10) hours off between shifts is not provided to the Employee, the Employee will be paid at the rate of two (2.0) times the regular rate of pay for the hours of the first shift worked.
- (c) If the Employee is on regular scheduled days off when the twenty-four (24) hours written notice is posted, the supervisor may contact the Employee directly to provide verbal twenty-four (24) hours notice change in shift.
- An Employee shall be paid for or expected to work for a minimum of three (3) hours if the regular shift is cancelled or rescheduled and the Employee was not notified of such cancellation or rescheduling on or before the day twelve (12) hours prior to the cancelled or rescheduled shift.
- When an Employee calls in ill for a work period affected by the preceding Clause 16.08, that Employee will be paid only for the minimum of three (3) hours that he would have been paid had he reported for work in the regular fashion. This clause shall not apply in cases in which the Employee provides proof of attendance at either medical, dental or optical appointments. In such cases, the Employee will be paid for either the half-day or full-day as appropriate to the circumstances and as provided in Clause 21.03 and 21.04.

#### 16.10 Early Reporting

(a) When an Employee is requested prior to the conclusion of a shift, to report or work more than two (2) or more hours earlier than his regularly scheduled shift, that Employee may (subject to operational requirements) be given a choice of leaving work prior to the end of their regularly scheduled shift or being paid the regular overtime rate for those hours worked prior to the time scheduled for the commencement of the regular shift. The total of the early reporting hours worked and regular shift hours worked will not be less than the number of regularly scheduled hours. Such early reporting arrangement may continue for two consecutive shifts.

If requested to report early on the second shift the early reporting hours will be at the overtime rate. If requested for a third consecutive shift, Clause 16.07 will apply.

(b) When an Employee is requested prior to the conclusion of a shift to report up to two (2) hours earlier or up to two (2) hours later for the next regular shift, that Employee shall not receive the overtime rate for the earlier or later reporting period. The total of the earlier reporting hours worked and regular shift hours worked will not be less than the number of regularly scheduled hours. If requested to report earlier or later on the second shift

the earlier or later reporting hours will be at the overtime rate. If requested for a third consecutive shift, Clause 16.07 will apply.

#### 16.11 Attendance

When an Employee is unable to report to work, the Employee shall follow the established protocol for their department to report such absence, and at a minimum shall report the absence one (1) hour prior to the start of the shift. within one hour prior to the scheduled shift start time. Failure to comply with this requirement shall result in loss of pay for the shift, and may result in discipline unless the Employee provides a reasonable explanation.

#### 16.121 Work Break

All Employees shall be entitled to one fifteen (15) minute work break in each half day worked, or in any overtime work period in excess of two (2) hours. Such work break shall be taken at the work site and the time of such break shall be approved by the immediate supervisor of each Employee. The work break shall be considered part of a work hour for the purposes of the preceding paragraphs.

#### 16.13 Sand Shift Workers and Sidewalk Crew

Notwithstanding Clause 16.01, when the needs of the operations require it, Sand Shift Workers and Sidewalk Crew may be required to work a modified shift schedule of up to twelve (12) hours. Modified hours of work shall commence annually on the first full pay period commencing on or after October 15 and shall continue until the first full pay period before or after April 15 based on operational need.

- (a) Overtime will be paid to Sand Shift Workers and Sidewalk Crew who exceed 80 hours of work in a bi-weekly pay period.
- (b) Employees working modified hours of work will have all benefits and entitlements which are expressed in terms of daily or weekly entitlement converted to produce the equivalent hours of benefits and entitlements as they would have had if the hours of work had not been modified. This will result in no loss or gain in Employee benefits and entitlements.

#### ARTICLE 17 - SHIFT AND WEEKEND PREMIUMS

17.01 (a) A shift premium of one dollar and forty cents (\$1.40) one dollar and ninety cents (\$1.90) per hour shall be paid to all Union Employees for those hours worked between seventeen hundred (1700) hours and zero seven hundred (0700) hours Monday to Friday and for all hours worked on Saturday and Sunday and holidays.

(b) At no time shall shift premium be included with the Employee's regular rate of pay for purposes of computing overtime payments, other premium payments, or any Employee benefits.

#### ARTICLE 20 - RELIEF AND ACTING INCUMBENT PAY

#### 20.01 Relief & Acting Pay

Any Employee who works in a classification (within the Bargaining Unit) with a higher rate of pay in any one (1) shift shall receive the higher rate of pay for each hour worked in the higher classification. If this time period exceeds ninety (90) days, effective the ninety-first (91st) day, the Employee will receive the higher rate of pay for all approved vacation, general holiday, illness, bereavement, and compassionate leave while working in the higher classification.

- 20.02 Notwithstanding the premium in Clause 20.09, An Employee appointed to Leadhand or Project Foreman shall not receive an hourly rate of pay that is less than the "County Department Union Employees" that he is responsible for.
- 20.03 Certified Technicians appointed to a seasonal Field Certified Technician assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- Vehicle Serviceman II appointed to a seasonal Field Vehicle Serviceman II assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- 20.05 The qualification or eligibility of an Employee to relieve or be assigned to a higher classification will be the responsibility of management.
- 20.03 An Employee who has become incapacitated by injury or illness may be employed in other work which he can do. Such Employee may not displace an Employee with more seniority.
- An Employee who refuses without good and satisfactory reason, to accept the alternate classification or position, may be suspended immediately. The supervisor issuing the suspension will report the suspension to the Department Head (Human Resources) as soon as possible who will notify the Union Executive.
- 20.05 Any Employee affected by the preceding Clauses 20.03 and 20.04 and not satisfied with the action recommended in accordance with Clause 20.04, may appeal the reclassification or suspension in keeping with Clause 14.03 of this Agreement.
- 20.06 Should an Employee who has completed the probationary period be unable to perform his usual duties by reason of the loss or suspension of his motor vehicle operator's license or a similar happening, that Employee may be placed in some other suitable position until the disability ends or is removed. The wages paid during the period of such disability shall be wages appropriate to the position in which the Employee is placed.
- 20.07 Notwithstanding the above clause, within fourteen (14) days of the suspension, an Employee may be allowed to book unused holidays at the wage prevailing prior to suspension of his operator's license.

#### **Acting Incumbent**

- 20.08 Any member of the Union may be appointed for one (1) or more days as a **Classified** supervisor and such appointment will not cause the member to be disqualified from the Union.
- 20.09 If any Employee is appointed temporary Classified Supervisor, they shall receive a premium of two dollars and twenty-five cents (\$2.25) per hour in addition to their regular wage rate for one (1) complete shift or longer.
- 20.10 Any member of the Union may be appointed for a period of time to the position of temporary Classified Supervisor and such appointment will not cause the

- member to be disqualified from the Union.
- 20.11 Notwithstanding the premium in Clause 20.09, an Employee appointed to Leadhand or Project Foreman shall not receive an hourly rate of pay that is less than the "County Department Union Employees" that he is responsible for.
- 20.12 Certified Technicians appointed to a seasonal Field Certified Technician assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- 20.13 Vehicle Serviceman II appointed to a seasonal Field Vehicle Serviceman II assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.

#### ARTICLE 21 - PROVISION FOR ILLNESS AND NORMAL ABSENCE

- 21.01 Sick leave benefits as herein stated are sponsored by the County to protect the Employee in the event of an unavoidable illness or injury not covered by the Alberta Workers' Compensation Act.
- 21.02 When an Employee is unable to report to work, the Employee shall follow the established protocol for their department to report such absence within one hour prior to the scheduled shift start time. Failure to comply with this requirement shall result in loss of pay for the shift, and may result in discipline unless the Employee provides a reasonable explanation. Moved to Article 16
- A Regular Employee, upon hire, shall be entitled to a maximum of **eighty (80)** hours ten (10) working days per calendar year of leave with pay for any casual illness, which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less. Up to **sixteen (16) hours** two (2) days of unused casual illness leave per calendar year shall be carried over and may be used in the next calendar year. The maximum paid casual illness leave in a calendar year is **ninety-six (96) hours** twelve (12) days including the carry over.
- A Temporary Employee after the completion of the standard probationary period, shall be entitled to a maximum of **thirty-two (32) hours** four (4) working days per calendar year of leave with pay for any casual illness, which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less. Temporary Employees with more than six (6) months of service in the current calendar year shall be entitled to an additional **sixteen (16) hours** two (2) working days in the current calendar year, of leave with pay for any casual illness. Up to two (2) days of allotment can be used to attend to illness of immediate family members.
- 21.05 Each day or portion of a day of absence due to casual illness used within a calendar year, shall be deducted from the remaining casual illness leave entitlement for that calendar year.
- NEW Where an employee terminates and is rehired in the same calendar year, the casual illness allotment will be the amount that they had in their available bank at the time of departure.
- 21.06 In any case, an Employee may be required to provide satisfactory proof of illness upon return to work, where it can be shown that reasonable doubt exists in respect

to the purpose of an absence claimed to be due to illness. **The employee will be provided the reasons for the request.** Such proof may take the form of a medical certificate signed by a qualified physician, dental practitioner or other professional as deemed appropriate by the Disability Management Supervisor/Human Resources.

- Upon approval (which will not be unreasonably denied), time off of two (2) hours or less for doctor appointments will result in no loss of casual illness or regular time. The Employer may require the Employee to submit proof of attendance at a medical, dental or optical appointment when time off work is granted to attend such appointments.
- 21.08 Each Regular Employee will be entitled to leave of absence with pay for General Illness in accord with the following scale. General Illness is defined as "any illness which causes an Employee to be absent from work for a period of more than three (3) consecutive work days".

To obtain General Illness benefits resulting from the inability of the Employee to perform the duties of his position, a proper medical certificate(s) signed by a qualified physician, dental practitioner or other professional as deemed appropriate by the Disability Management Supervisor/Human Resources is required.

#### 21.09 <u>Family Illness Leave</u>

A Regular Employee shall be entitled to a maximum of twenty-four (24) hours three (3) working days per calendar year with pay to attend to illness of immediate family members as per defined in Clause 22.06.

- 21.10 A Temporary Employee may use up to sixteen (16) hours two (2) days of their casual illness allotment, as per Clause 21.04 can be used to attend to illness of immediate family members.
- 21.0911 Effective the calendar year following ratification, Sick leave entitlement for illness longer than three (3) working days shall be as follows:
  - (a) After ninety (90) days three (3) months five hundred (500) hours of continuous employment the Employee shall receive one hundred sixty (160) hours twenty (20) working days at the base rate of pay;
  - (b) On the first day worked, in the next calendar year sick leave entitlement shall be increased to a maximum **two hundred eighty (280) hours** thirty-five (35) working days at the base rate of pay
  - (c) On the first day worked, in subsequent consecutive calendar years sick leave entitlement shall be increased to a maximum of **five hundred twenty** (520) hours sixty five (65) working days at the base rate of pay, plus any unused casual illness up to a maximum of six hundred (600) hours seventy-five (75) working days.
- 21.1012 When a day designated as a Paid Holiday under Clause 23.01 falls within a period

of General Illness, it will be considered as a paid holiday.

21.1213 Transfers, promotions, demotions and/or reclassification shall not affect sick leave entitlement of any Employee.

#### 21.1114 Workers Compensation

- a) An Employee is not eligible to receive sick leave benefits if the absence is due to an injury while in the employ of any other Employer that is covered by Workers' Compensation Benefits.
- b) Employee's will be eligible to receive benefits in accordance with the County's policies.
- 21.12 Transfers, promotions, demotions and/or reclassification shall not affect sick leave entitlement of any Employee.
- 21.1315 Sick leave entitlement will not be affected while the Employee is in receipt of Alberta Workers' Compensation benefits.
- 21.1416 Regular Employees who are eligible to receive Workers' Compensation benefits shall receive full pay from the County for a period not to exceed six (6) months provided that all Workers' Compensation payments are assigned to the County for that period.
- 21.1517 Expenses for medical certificates will be the responsibility of the employee, including medical certificates required to support eligibility for casual or general illness. Where the Employer has requested the employee to attend a medical examination and/or obtain additional medical documentation, the Employer shall reimburse the employee for the cost of the documentation (if any).
- 21.1618 An Employee who has become incapacitated by injury or illness may be employed in other work which he can do. Such Employee may not displace an Employee with more seniority.
- 21.1719 An Employee who refuses without good and satisfactory reason, to accept the alternate classification or position, may be suspended immediately. The supervisor issuing the suspension will report the suspension to the Department Head (Human Resources) as soon as possible who will notify the Union Executive.
- 21.1820 Any Employee affected by the preceding Clauses 21.176 and 21.187 and not satisfied with the action recommended in accordance with Clause 20.04, may appeal the re-classification or suspension in keeping with Clause 14.03 of this Agreement.
- 21.1921 Should an Employee who has completed the probationary period be unable to perform his usual duties by reason of the loss or suspension of his motor vehicle operator's license or a similar happening, that Employee may be placed in some other suitable position until the disability ends or is removed. The wages paid during the period of such disability shall be wages appropriate to the position in which the Employee is placed.
- 21.2022 Notwithstanding the above clause, within fourteen (14) days of the suspension *of* **their operator's license**, an Employee may be allowed to book unused holidays at the wage prevailing prior to suspension of his operator's license.

#### **ARTICLE 22 -LEAVES OF ABSENCE**

- Other Leaves of Absence, without pay, may be granted at the sole discretion of the County, and in accordance with the Alberta Employment Standards Code, upon written application by the affected Employee at least seven (7) days prior to the beginning date of the applied for Leave of Absence. To be eligible for such leave, an Employee shall have been employed with the County for ninety (90) days of continuous employment, with the exception of reservist leave which is twenty-six (26) consecutive weeks, and must have his application endorsed or approved by the Department Head. The application shall include the reason for the request as well as the dates of the requested Leave of Absence. Such approval should not be unreasonably withheld.
- In unusual circumstances or by mutual agreement, the requirement stated in Clause 22.01 for application to be made seven (7) days prior to the commencement of the requested leave of absence may be waived. Application for such waiver is to be made directly to the Department Head.
- In the case of an Employee who takes a leave of absence under the provisions of Clause 22.01, the benefits and conditions to which that Employee is entitled as a result of length of service in the County shall not be adversely affected as a result of having been granted such leave of absence.
- 22.04 An Employee on such approved leave may not engage in other employment for gain while on such leave without the express written consent of the County.

#### 22.05 <u>Court and Jury Duty</u>

Leave of Absence shall be provided to an Employee who is required to serve as a juror, or is subpoenaed as a witness **in a private capacity** in any court proceeding. The Employee shall receive their regular rate of pay and shall remit to the Employer any allowance they may receive for such duty or service.

If the Employee is subpoenaed as a witness to appear in court in their official capacity to give evidence or to produce Strathcona County records, the Employee shall receive their regular rate of pay during regular working hours and is eligible to receive overtime pay to appear during non-work hours. The Employee shall remit to the Employer any allowance they may receive for such duty or service.

The Employee shall present proof of such duty or service and the amount of remuneration or allowance received **in all cases.** 

Employees required to appear in court in a capacity other than subpoenaed as a witness or jurist will be granted leave without pay.

#### 22.06 Bereavement Leave

An Employee shall be granted Bereavement Leave of Absence with pay for not more than five (5) consecutive days because of the death of spouse (includes common-law), child (includes step, foster child or ward), parent, (includes step or foster parent), brother, sister, parent of spouse, guardian, son-in-law, daughter-in-law, grandchild or any other relative who is a member of the Employee's household, or the spouse of any of the foregoing. In addition to the above, the Employer may grant up to one (1) day travel to attend the funeral if the travel is five hundred (500) kilometres or more away.

22.07 Leave of absence for one (1) full day, with pay, shall be granted **at the time of** to attend the funeral **or death** of a grand-parent, grandparent-in-law, brother-in-law

or sister-in-law, aunt, uncle, niece or nephew, if the funeral is on a work day.

- 22.08 Upon written application, additional Compassionate or Bereavement Leave of Absence with pay, due to exceptional circumstances, may be granted an Employee at the discretion of the Chief Commissioner.
- One (1) day of leave may be granted without loss of pay or benefits to attend as a pallbearer or mourner for a deceased Employee. Where the family of a deceased Employee requests pallbearers from the Union, the County may grant the necessary leave with pay for up to six (6) pallbearers.

#### 22.1122.10 Family Illness Leave

A Regular Employee shall be entitled to a maximum of **twenty-four (24) hours** three (3) working days per calendar year with pay to attend to illness of immediate family members as per **defined in** Clause 22.06.

**A Temporary Employee may use** up to **sixteen (16) hours** two (2) days of **their casual illness** allotment, **as per Clause 21.04** can be used to attend to illness of immediate family members.

#### 22.10 <u>Compassionate Leave</u>

An Employee shall be granted Compassionate Leave of Absence with pay for not more than five (5) consecutive days because of the critical illness of a member of his immediate family as per **defined in** Clause 22.06 or any other relative who is a member of the Employee's household. The Employee shall provide the County with a letter from the attending physician concerning the fact of the critical illness.

Any additional Compassionate Care Leave shall be provided in accordance with *Alberta Employment Standards Code*.

- 22.11 A Regular Employee shall be entitled to a maximum of three (3) working days per calendar year with pay to attend to illness of immediate family members as per Clause 22.06.
- 22.12 One (1) day leave with pay will be allowed for an Employee to be present at the birth or adoption proceeding of an Employee's child, providing such occurrence is on a scheduled work day.

#### 22.1311 Maternity Leave

Maternity Leave will be granted to Employees in accordance with the Alberta Employment Standards Code.

- (a) Employees must have been employed with Strathcona County for ninety (90) days of continuous employment.
- (b) The maternity leave is not more than sixteen (16) weeks starting at any time during the thirteen (13) weeks immediately before the estimated date of delivery.
- (c) The Employee must give six (6) weeks' written notice of the date she intends to begin leave and four (4) weeks' notice of the date she intends to return to work.
- (d) If an Employee's pregnancy results in a miscarriage or stillbirth within sixteen (16) weeks of their due date, they are entitled to sixteen (16) weeks of maternity leave.

#### 22.124 Health Related Portion of the Maternity Leave

During the one-week Employment Insurance waiting period, Strathcona County will top-up salary to ninety-five percent (95%) of the Employee's regular earnings. Verification that Employees are in receipt of employment insurance benefits is necessary to provide the top-up benefit. A medical certificate must be provided to Human Resources, confirming the date the child was born and the duration of the leave that the physician indicates is health related. During the health-related portion of the maternity leave Strathcona County will top-up salary to ninety-five percent (95%) of the Employees regular earnings. The benefit coverage that the Employee has at the time of her leave will continue during the health related portion of the absence. At the end of this period, parental leave will commence, and benefit coverage will continue if the Employee pays both Employer and Employee portions of the cost of the premiums.

When a pregnancy interferes with the performance of duties or the Employer has a concern for the Employee's health, the Employer may reassign the Employee's duties at no loss of pay, or by notice in writing to the Employee, require them to commence sick leave under Clause 21.08.

#### 22.1416 Parental Leave

Employees who have been employed with Strathcona County for ninety (90) days of continuous employment are entitled to parental leave without pay for a maximum of sixty-two (62) weeks. The Employee must give at least six (6) weeks written notice of the date they intend to begin parental leave and four (4) weeks' notice of the date they intend to return to work. During that time they may receive benefit coverage by paying the Employee and Employer portions of the cost of the premiums.

22.1517 Employees wishing to return to work prior to the agreed return date may do so with two (2) weeks' notice.

#### 22.1618 Adoption Leave

One (1) day leave with pay will be allowed for an Employee to be present at the birth or adoption proceeding of an Employee's child, providing such occurrence is on a scheduled workday.

The Employer will grant adoption leave to Employees who are adopting a child. The Employee must have been employed with Strathcona County for ninety (90) days of continuous employment. Written notice of leave must be provided to the Employer at least two (2) weeks before Employees can reasonably expect to obtain custody of the child.

- 22.1719 If Employees cannot provide at least two (2) weeks written notice, they must notify the Employer immediately after receiving notice of the adoption.
- 22.1820 Adoption leave consists of not more than sixty-two (62) consecutive weeks of leave without pay within seventy-eight (78) weeks after the child is placed with the adoptive parents. During that time they may receive benefit coverage by paying the Employee and Employer portions of the cost of the premiums.
- 22.1921 Only one (1) parent of a child referred to in Clause 22.20 will be granted adoption leave under this clause.

#### 22.2022 Shared Leave

If both parents are Employees of Strathcona County parental or adoption leave may be taken wholly by one of the Employees or be shared by both. Only one (1)

parent will be granted parental or adoption leave at a time.

#### 22.2123 Anniversary Date

Employees who have been on maternity, parental or adoption leave will have their anniversary date adjusted for evaluation and increment purposes only. They will not earn vacation leave during this period, the exception being for maternity health related reasons.

#### 22.2224 Additional Leave

Leave of absence without pay beyond the approved maternity, parental or adoption leave may be granted upon written application up to a combined maximum of twelve (12) months at the discretion of the Employer.

#### 22.2325 <u>Domestic Violence Leave</u>

Domestic violence leave shall be granted in accordance with the *Alberta Employment Standards Code*.

#### 22.**242**6 Military Leave

Military leave shall be granted in accordance with the Alberta Employment Standards Code.

#### 22.2527 <u>Citizenship Ceremony Leave</u>

Citizenship Ceremony Leave shall be granted in accordance with the *Alberta Employment Standards Code*.

#### **ARTICLE 23 - HOLIDAYS**

Employees shall be entitled to one (1) day's paid leave for each of the following specified or general holidays:

New Year's Day Labour Day

Family Day
Good Friday
Remembrance Day
Easter Monday
Christmas Day
Victoria Day
Christmas Floater

Canada Day Boxing Day

1st Monday in August **Truth and Reconciliation Day** 

And any other day designated as a General Holiday by the Lieutenant Governor in Council under the Employment Standards Code.

Employees shall receive the recognized specified or general holidays for which they are eligible, with pay; or other days with pay in lieu of such holidays, or pay in lieu as approved by the County, providing they are available for work immediately preceding and following the days or an otherwise approved leave of absence, or they provide a medical certificate attesting to their illness on the day

prior or following the general holiday.

- 23.03 The provisions for pay in lieu of such specified or general holidays referred to in Clauses 23.01 and 23.02 shall be based on the number of regular hours which would have been worked had that day been a regular working day.
- In the event a specified or general holiday falls on an Employee's regular day off, an additional day with pay shall be granted in lieu of that day.
- 23.05 If any of the above specified or general holidays should occur during an Employee's vacation, such vacation shall be extended by one day or an additional day with pay shall be granted in lieu of that day.
- 23.06 Temporary Employees, upon hire, shall be eligible for specified or general holidays as per Clause 23.02.
- 23.07 With the exception of Sand Shift workers and Sidewalk Crew and those workers that are scheduled to work on Christmas day, the Christmas Floater holiday shall be observed to give five (5) consecutive days off including the weekend as follows:
  - (a) On December 24 when Christmas Day falls on a Tuesday, a Thursday, a Friday or Saturday;
  - (b) On December 27 when Christmas Day falls on a Monday or a Wednesday;
  - (c) On December 28 when Christmas Day falls on a Sunday.

For Sand Shift workers and Sidewalk Crew and those workers that are scheduled to work on Christmas day, the Christmas Floater Holiday shall be observed on December 24<sup>th</sup>.

#### **ARTICLE 24 – VACATION LEAVE**

- 24.01 Annual vacation entitlement with pay shall be as follows:
  - (a) An Employee who has completed less than six (6) years of service shall earn vacation leave at the rate of four point sixty-two (4.62) hours per biweekly period of continuous service [fifteen (15) days per year] or five point two (5.20) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
  - (b) An Employee who has completed six (6) years but less than fourteen (14) years of service shall earn vacation leave at the rate of six point fifteen (6.15) hours per bi-weekly period of continuous service [twenty (20) days per year] or six point ninety-two (6.92) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
  - (c) An Employee who has completed fourteen (14) years of service but less than twenty two (22) years shall earn vacation leave at the rate of seven point seven (7.70) hours per bi-weekly period of continuous service [twenty-five (25) days per year] or eight point sixty-seven (8.67) hours per bi-weekly period of continuous service if based on the forty-five (45) hour

work week;

- (d) An Employee who has completed twenty two (22) or more years of service shall earn vacation leave at the rate of nine point twenty-three (9.23) hours per bi-weekly period of continuous service [thirty (30) days per year] or ten point thirty-nine (10.39) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week.
- 24.02 If Employees become ill on an annual vacation leave for more than three (3) consecutive days paid vacation days, the applicable days will be deemed general illness, and they will have the applicable number of vacation days reinstated upon application and provision of a medical certificate.
- 24.03 The number of days of vacation leave which may be accumulated by an Employee is limited to the number he can earn in eighteen (18) months of continuous service.

The County will automatically pay out any excess vacation leave over the maximum accumulation at the beginning of each December.

When vacation pay is paid out rather than being taken as vacation, the payout shall be based on the Employee's rate of pay at the time of payout.

- 24.04 Vacation leave may be taken in the calendar year it is earned. However, vacation leave may not be taken until it has been earned. The vacation leave may be granted as follows:
  - (a) In one continuous period;
  - (b) As single days off or some other combination;
  - (c) If an Employee earns vacation at a rate of four point sixty-two (4.62) hours per bi-weekly period of continuous service, one (1) week vacation leave may be taken as single days off or some other combination provided the total does not exceed five work days in any calendar year;
  - (d) If an Employee earns vacation at a rate of six point fifteen (6.15) hours per bi-weekly period of continuous service or seven point seven (7.70) hours per bi-weekly period of continuous service, two (2) weeks' vacation leave may be taken as single days off or some other combination provided the total does not exceed ten (10) work days in any calendar year;
  - (e) If an Employee earns vacation at a rate of nine point twenty-three (9.23) hours per bi-weekly period of continuous service, three (3) weeks' vacation leave may be taken as single days off or some other combination provided the total does not exceed fifteen (15) work days in any calendar year.
- In addition to the automatic payout of excess vacation referred to in 24.03, an Employee may be permitted to receive a cash payment in lieu of vacation leave, provided any and all obligations have been met in accordance with Employment Standards legislation and the Employee has received written approval from the Department Head.

Temporary Employees shall have vacation pay added to their bi-weekly pay cheques and shall be entitled to unpaid vacation leave in accordance with the Employment Standards Code.

- 24.06 An Employee who does not complete one (1) full year of service shall be entitled to vacation pay in accordance with the Employment Standards Code and regulations thereunder.
  - (a) Temporary Employees will receive vacation pay at four percent (4%) of

their gross regular earnings during their first four thousand (4000) regular standard hours.

- (b) Temporary Employees will receive vacation pay at six percent (6%) of their gross regular earnings after completion of four thousand (4000) regular standard hours.
- (c) If a Temporary Employee has a break in service greater than eighteen (18) months then vacation pay shall be paid at four percent (4%) of their gross regular earnings as per Clause 24.076 a) above.
- Regular vacation leave shall be taken by Employees after receiving written authorization from the County or its designated official who shall consider the wishes and application of the Employee. Length of regular service Seniority with the County shall be the determining factor in resolving any dispute or conflict regarding preference for the time at which vacations are to be taken. Changes in vacation scheduling shall be authorized in writing.
- Vacation hours will be paid out at the Employee's base rate in place at the time the vacation is taken.

#### **Summer Vacation Schedules**

#### 24.09 Summer Vacation Schedules

Employees must submit summer vacation applications to the Director or his designate by March 15th of each year. The Employer will post schedules, as approved by the Director, by May 1st of each year.

#### **ARTICLE 25 - BENEFIT AND PENSION PLANS**

#### 25.01 Benefit Plans

The County shall contribute on behalf of all participating Employees in group insurance and benefit plans, which provide minimum benefit levels agreed upon by the County and the Union at rates outlined in the following clauses:

- (a) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Strathcona County Benefit Plan Life, Accidental Death and Dismemberment and Long-Term Disability Insurance. All members of the Union hired after July 1, 1978, shall be members of the said plan as a condition of employment;
- (b) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Extended Health Care option of the Strathcona County Employee Benefit Plan;
- (c) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Dental option of the Strathcona County Employee Benefit Plan;
- (d) The County shall pay seventy five percent (75%) of the total premium for each eligible Union member enrolled in the Alberta Health Care Insurance Plan through the Strathcona County.

The County's liability shall be limited to the payment of premiums, where required, for the various plans and the County shall be relieved of any further

liability with respect to the plans.

In the event that the County should undertake to transfer group insurance and benefit plans from one carrier to another, it is agreed that the Union will be consulted prior to the negotiation of such coverage.

#### **Benefits for Temporary Employees**

#### 25.03 **Benefits for Temporary Employees**

To be eligible to participate in the Strathcona County Employee Benefit plans, Temporary Employees must meet the following conditions:

- (a) All Temporary Employees who have been employed for twelve (12) consecutive or more, shall receive the same benefits as Regular Employees.
- (b) An exception to the above shall be made for Temporary Employees replacing Regular Employees on long term disability. A Temporary Employee in this situation will be eligible for benefits after six (6) consecutive months of employment.

#### Pension Plan

#### 25.04 **Pension Plan**

The County participates in the Local Authorities Pension Plan. Participation in this plan is compulsory for "Regular Employees" after they have completed one (1) year of service or immediately if transferring from another Local Authority Board Employer.

25.05 The County shall make payroll deductions as follows:

- (a) The County shall deduct legislated Revenue Canada deductions and Local Authorities Pension Plan contributions as required from each bi-weekly pay cheque;
- (b) The current authorized deductions for Canada Savings Bonds will be taken each pay cheque;
- (c) The Employee contribution to the Strathcona County Employee Benefit Plan will be deducted from the first bi-weekly pay cheque. each month.

#### **ARTICLE 29 - TEMPORARY EMPLOYEES**

29.01 Temporary Employees shall be entitled to the terms and conditions of this Agreement, except where Regular Employees are referenced. Seniority for employment periods of less than six (6) months shall not be accumulated.

A Temporary Employee shall be considered probationary as per Article 11.01 and shall not serve probation for subsequent years of employment unless a break of service of greater than one (1) year has occurred or the Employee is returning to a different position. Temporary Employees shall have access to the application of the grievance procedure except in connection with termination of employment as

a result of either:

- (a) Unsatisfactory performance, or;
- (b) The abolishment of the position occupied by the Temporary Employee.

29.03 For the purpose of determination of Pay Grade and Step a Temporary Employee returning for subsequent employment within one (1) year may be required to undertake the appropriate probationary period unless he is returning to a position in which he has previously fulfilled the probationary requirements.

29.04 When Temporary Employees are recalled, it shall be based on:

- (a) A satisfactory performance evaluation from the previous year of employment;
- (b) The Employee having qualifications to do the job for which he is recalled;
- (c) The Employee's accumulated service.

The opportunity for recall shall exist for six (6) months within the Branch.

This clause will not apply to Employees participating in the NAIT or other post-secondary Water and Wastewater Training Program.

#### **ARTICLE 30 - GENERAL WAGE ADMINISTRATION**

The rates of pay for all Employee members of the Union excluding Apprentices, covered by the Agreement shall be as specified in the attached Schedule A.

Temporary Employees returning for subsequent employment within one (1) year shall have their prior service recognized for the purposes of determining Pay Grade and Step.

30.02 An Employee is entitled to the rates indicated in Schedule A in the following way:

- (a) To Step One (1) upon hiring;
- (b) To Step Two (2) upon completion of the five hundred (500) hours of work in the classification and recommendation of the appropriate supervisor;
- (c) To Step Three (3) upon completion of an additional fifteen hundred (1,500) hours of work in the classification and the recommendation of the appropriate supervisor
- (d) Water & Wastewater Operators step progression is as stated in the Water and Wastewater Job Summary attached to the Collective Agreement. Water and Wastewater Operators step progression shall be:
  - (i) To Step One (1) upon hiring;
  - (ii) To Step Two (2) upon completion of five hundred (500) hours of work in the classification, additional qualifications (in accordance with the job description), and recommendation of the appropriate supervisor;
  - (iii) To Step Three (3) upon completion of an additional one thousand eight

hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;

- (iv) To Step Four (4) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (v) To Step Five (5) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (vi) To Step Six (6) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;

#### (e) Construction Operator step progression shall be:

- (i) To Step One (1) upon hiring;
- (ii) To Step Two (2) upon completion of two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (iii) To Step Three (3) upon completion of an additional two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (iv) To Step Four (4) upon completion of an additional two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the description), and the recommendation of the appropriate supervisor.
- (f) Water and Wastewater Leadhand step progression:

Water & Wastewater Operator, Steps 1 to 4	Water & Wastewater Leadhand, Step 1
Water & Wastewater Operator, Step 5	Water & Wastewater Leadhand, Step 2
Water & Wastewater Operator, Step 6	Water & Wastewater Leadhand, Step 3

(g) Water and Wastewater Infrastructure Repair Technician step progression:

Water	&	Wastewater	Operator,	W&WW	Infrastructure	Repair

Steps 1 to 4	Technician, Step 1
Water & Wastewater Operator, Step 5	W&WW Infrastructure Repair Technician, Step 2
Water & Wastewater Operator, Step 6	W&WW Infrastructure Repair Technician, Step 3

(h) An appropriate form will be completed and placed In the Employee's personnel file when the wage rate is amended.

30.03 Should an Employee not be recommended by the appropriate supervisor for advancement as per Clause 30.02 b), c), d) or e), the Administration shall advise that Employee of the reasons in writing, and in addition:

- (a) What action might be taken in order for the wage to be advanced.
- (b) In the event that the wage will not be advanced, what alternate action the administration intended to take.
- 30.04 An Employee affected by Article 30.03 may file a grievance for a failure to be recommended for an increase.
- 30.05 The rate of pay established in Schedule A shall apply during the term of this Agreement. Employees shall be paid bi-weekly. On each pay day, each Employee shall be provided with an itemized statement of his wages, overtime and other earnings and deductions.

#### 30.06 Flag Person

The Employer may hire Temporary Employees as Flagperson at a rate in accordance with Schedule A. Should such employees be assigned to any other work they shall be paid in accordance with Clause 20.01. No one shall be regularly assigned duties as Flagperson until they have successfully completed an approved Flagperson's training course. Employees taking the Flagperson training shall be paid the appropriate rate of pay while training.

#### ARTICLE X - DIRECT RESPONSILBE PAY (DRP)

X.01 When a Water and Wastewater Operator or Water and Wastewater Leadhand is assigned to the Utilities On-Call Schedule as On-Call #1, On-Call #2 or Transmission Operator (for emergent calls) and deemed to be in direct responsible charge (DRC), the Employee shall be paid an additional, two dollars and twenty-five cents (\$2.25) per hour on the base rate of pay, for all hours worked between sixteen hundred and thirty (1630) hours and zero eight hundred (0800) hours Monday to Friday and for all hours worked on Saturday and Sunday and Holidays.

- (a) At no time shall Operators called out to assist the On-Call #1, On-Call #2 or Transmission Operator with emergency service or other similar callouts receive DRP.
- X.02 Where applicable, an Employee shall be eligible to receive DRP in addition to overtime, shift and weekend premiums.
- X.03 At no time shall DRP be included with the Employee's regular rate of pay for the purpose of calculating overtime payments, other premium payments, or any Employee benefits.

#### **WAGE INCREASES:**

January 1, 2020 – 0.5% - Retro to be paid out within 3 months following ratification

**January 1, 2021 – 0%** 

January 1, 2022 - 1% - Retro will be paid out within three months following ratification

January 1, 2023 – 2.5% - Retro will be paid out within three months following ratification.

The following is included for clarity, but will not form part of the collective agreement:

- Shift Premiums are not retroactive and will be effective on the date of ratification.
- Unless otherwise specified, all changes are effective on the date of ratification.
- Premiums referenced in Article 17 will be included in pensionable earnings.
- The updated grids reflecting agreed upon increases will be provided by May 18, 2023.

The following articles have no amendments and will remain as current, except for any consequential amendments that may arise as a result of other agreed to amendments:

**Article 5 – Management Rights** 

Article 7 - Time off for Union Business

Article 13 – Discipline

Article 18 - Call out & Standby

### **SCHEDULE A**

\*\*To be provided my May 18, 2023.

#### **SCHEDULE B**

EQUIPMENT	MINIMUM OF CLASSIFICATION RATE PAID
Non-Finishing Packers (Rubber Roller, Pad Foot Operator, and combination for non- finishing work)	Equipment Operator I – Step 1
Finishing Packer (Smooth Drum Rollers and combination of finishing work)	Equipment Operator I
Skid Steer / Sidewalk Machine	Operator II TBD
Single Axle Truck	Truck Driver I Equipment Operator I
Sign Truck (Single Axle with Manlift)	Truck Driver II Equipment Operator I
<del>Tracked Dozer</del>	Equipment Operator II
Wheel Loader	Equipment Operator II
Tandem Axle Truck	Truck Driver II Equipment Operator II
Street Sweepers	Equipment Operator II
Truck Plow	Equipment Operator II
Sander	Truck Driver II Equipment Operator II
RPM Snow Blower	Equipment Operator II
Road Reclaimer	Equipment Operator II
PB Patcher Patch Truck	Equipment Operator II
Rubber Tire Backhoe	Equipment Operator II
Graders	<del>Operator II</del>

	Equipment Operator III
Track Hoe	Equipment Operator III + \$0.20 / hour
Paver	Equipment Operator III
Tractor-Trailer	Truck Driver III Class 1
Directional Driller	Utility Operator III

Letter of Understanding

#### **Between**

#### **Strathcona County**

and

The Alberta Union of Provincial Employees, Local 118, Chapter 009

Subject: Classification of the Skid Steer/Sidewalk Machine

- 1. The parties have agreed to Implement new classifications and associated rates of pay, however there is a dispute regarding the classification of the Skid Steer/Sidewalk Machine outlined in Schedule B.
- 2. As a result, the parties have agreed that the appropriate classification, of the Skid Steer/Sidewalk Machine, shall be advanced to arbitration (the "Matter").
- 3. The parties agree that Dev Chankasingh will act as arbitrator.
- 4. The arbitrators' jurisdiction shall be restricted to deciding whether the Skid Steer/Sidewalk Machine Is properly classified as an Equipment Operator I or an Equipment Operator II.
- 5. The Matter shall be arbitrated by September 30, 2023. The arbitrator shall Issue a decision within thirty (30) days from the date of the arbitration.
- 6. The decision of the arbitrator shall be final and binding.
- 7. The arbitrator shall not alter, amend, or otherwise change any term of the Collective Agreement, Including any of the pay rates or classifications in Schedule A, or any other classifications in Schedule B.
- 8. Both the Employer and the Union will share the cost of the arbitrator equally.

This letter of Understanding expires December 31, 2023.

Letter of Understanding

Between

Strathcona County

and

The Alberta Union of Provincial Employees, Local 118, Chapter 009

#### Subject: Implementation of New Classifications and Grid Placements

- 1. The new classifications and corresponding rates of pay (new Schedule A & B) shall be implemented within three (3) months of the arbitrator rendering their decision regarding the classification of the Skid Steer/Sidewalk Machine.
- 2. Upon implementation of the new Schedule A, if an Employee's wage rate falls between 2 steps, the Employee's rate of pay will be adjusted to the next higher rate of pay on the grid.
- 3. Employees with hours worked at the Operator II classification prior to implementation of the new Schedule A ratification, including such hours worked on the Skid Steer/Sidewalk Machine, shall continue to have such hours recognized.
- 4. The following positions will be reclassified as follows:
  - Truck Driver 1 and Operator 1 positions will be reclassified as Equipment Operator 1;
  - Truck Driver 2 and Operator 2 positions will be reclassified as Equipment Operator 2;
  - Truck Driver 3 positions will be reclassified as Truck Driver Class 1;
  - Operator 3 positions will be reclassified as Equipment Operator 3;
  - Utility Operator 1, 2 and 3 positions will be reclassified as Construction Operator;
  - Labourer 1, 2 & 3 positions will be reclassified as PW Labourer.

This letter of Understanding expires December 31, 2023.

# PROPOSED MEMORANDUM OF SETTLEMENT BETWEEN

## ALBERTA UNION OF PROVINCIAL EMPLOYEES

AND

# STRATHCONA COUNTY

December 15, 2021

The attached documents constitute a full settlement of all issues and proposals remaining outstanding in the current round of negotiations. This proposed Memorandum of Settlement combined with the previously signed off Articles, and Letters of Understanding will form a Tentative Agreement between the parties, which will be subject to the normal ratification process of each party.

The Parties agree to recommend to their respective principals the terms and conditions contained in the Tentative Agreement.

Signed at Sherwood Park, this \_\_\_\_\_ day of December, 2021.

FOR STRATHCONA COUNTY

Slisha

**Sharla Mish** 

**FOR AUPE LOCAL 118** 

25-



Devl Pud.

THIS AGREEMENT is made this 18th day of April, 2018, A.D. pursuant to the Labour Relations Code of Alberta (2000) as amended.

This Agreement is entered into by and between the Strathcona County, municipal corporation covered by the Collective Agreement (hereinafter referred to as "the County" or "the Employer") and the Alberta Union of Provincial Employees Local 118/009 (hereinafter referred to as "the Union").

Whereas, the above parties are mutually desirous of entering into a Collective Agreement with the intent and purpose of:

- Α. Promoting a harmonious relationship;
- В. Providing for conditions of employment, hours of work and rates of pay;
- Establishing the means for prompt and equitable settlement of disputes or grievances C. between the County and Employees represented by the Union, and;
- D. Encouraging efficiency in operation.

Now therefore, the parties hereto mutually agree as follows:

# **ARTICLE 1 – DEFINITIONS**

- 1.01 In this Agreement:
  - A word used in the singular applies also in the plural. A word used in the (a) masculine applies also in the feminine.
  - Employee: means a person employed by the Strathcona County in the (b) Public Works Operations, Vehicle and Equipment Services, Transit Services, Materials Management, Water and Wastewater Operations, and as listed in Schedule A.
    - (i) Regular Employees: are Employees who occupy established positions and who have successfully completed the probationary period and work the fully scheduled hours.
    - (ii) Part-time Employees: are Employees who occupy established positions and have successfully completed the probationary period

- and work less than the fully scheduled hours of Regular Employees.
- Temporary Employees: are Employees who are indefinite term (iii) Employees or Employees for specific projects, seasonal employment, replacements for Regular Employees who are on illness leave, General Illness, Long Term Disability, Workers' Compensation, Maternity Leave, Parental Leave and Adoption Leave, or other long term leaves of absence. Indefinite term Employees upon completion of more than fifteen (15) months of continuous service will be deemed to be Regular Employees, except if they are covering for an Employee on Long Term Disability, are moving from one Department to another Department for a temporary or seasonal position, or are covering off consecutive leaves, as referenced above. "Continuous service", for the purpose of this clause, means any period of employment that does not include a break in service of more than thirty (30) calendar days.
- (iv) Apprenticed Employees: are Employees who are registered as Apprentices under the Apprenticeship and Industry Training Act.
- (v) Probationary Employee: means an Employee who during his initial period of employment occupies an established position during a probationary period.
- (c) Grade: means the step assigned to a class within the pay grid.
- (d) Union Steward: means a person who is selected by the members of the Union to act on behalf of those Employees as empowered by the Union.
- (e) Union Representative: means a person authorized by the Union to act on behalf of Employees and includes a Union Steward, elected or appointed officer of the Union, or staff representative.
- (f) Liaison Committee: is the joint Liaison Committee established by the two parties as stipulated within this Agreement.
- (f) Casual Illness: is any illness which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less.
- (g) General Illness: is any illness which causes an Employee to be absent from work for a period of more than three (3) consecutive work days.
- (h) Layoff: includes a reduction in the work force as a result of shortage of work.
- (i) Branch: means
  - (i) Public Works Operations
  - (ii) Vehicle and Equipment Services

- (iii) Transit Services
- (iv) Water and Wastewater Services
- (v) Materials Management
- (j) Classification: means each line in Schedule A.
- (k) Seniority: commences at date of hire, and shall include all continuous **and accumulated service** as a temporary Employee. <del>provided there was no break in service greater than thirty (30) days.</del>
- (I) Day: shall mean calendar day.
- (m) Continuous Service: any period of employment that does not include a break in service of more than thirty (30) calendar days.
- (n) Accumulated Service: Total number of regular hours worked, with no break in service of more than six (6) months.

#### **ARTICLE 2 – JURISDICTION**

- 2.01 During its term, this Agreement shall be applicable to any person employed in the Public Works Operations, Vehicle and Equipment Services, Transit Services, Materials Management and Water and Wastewater Services of Strathcona County, who are described in Schedule A attached hereto, or to positions that fall within the jurisdiction of the Union.
- 2.02 When additional work is required by the Employer, individuals who have paid Union dues to the Alberta Union of Provincial Employees, Local 118, Chapter 009 in the past twelve (12) months, will be called first and, if available, offered the additional work.

## **ARTICLE 4 - APPLICATION**

- 4.01 Regular Employees shall be granted all the terms and conditions of this Agreement.
- 4.02 Part-time Employees shall be granted all the terms and conditions of this Agreement except those conditions which require a particular length of service, such benefits shall be granted on a pro-rata basis in relation to hours of work of full-time employment.
- 4.03 Apprenticed Employees as defined within Apprenticeship and Industry Training Act shall be granted all the terms and conditions of this Agreement.
- 4.04 Temporary Employees shall be entitled to the terms and conditions of this Agreement, except where Regular Employees are referenced. Seniority for employment periods of less than six (6) months shall not be accumulated.

#### **ARTICLE 6 - UNION RECOGNITION**

- 6.01 The County recognizes the Union as the sole bargaining agent of the Employees covered by this Agreement.
- 6.02 No Union Employee shall be required or permitted to make any written or verbal agreement which may be in conflict with the terms of this Collective Agreement.

## 6.03 New Employees

The Employer shall supply all new Employees with a copy of the Collective Agreement.

# 6.04 Monthly Staffing Reports

The County recognizes the need of the Union to have required information concerning Employees eligible for membership in the Union and hereby agrees to submit between the sixteenth (16th) day and the last day of each month, the name, address, and position filled by any new Employees assigned to Public Works Operations, Vehicle and Equipment Services, Materials Management, Water and Wastewater Operations and eligible for membership in the Union.

The Employer recognizes the importance of new Union Employees receiving an orientation to the Union. To ensure this occurs, the Chapter Executive will be given a list of all newly hired Employees and a representative of the Union shall, within thirty (30) calendar days of the Employees' date of hire, make a presentation to the Employee(s) with respect to the structure of the Union as well as the rights, responsibilities and benefits of membership in the Union.

The Employer will generally, provide the Union with one (1) weeks' advance notice of the Employee's scheduled orientation.

#### 6.04 No Discrimination

The parties agree that there shall be no discrimination or coercion exercised or practiced with respect to any Employee for reason of function, membership or legitimate activity in the Union.

The Employer shall exercise its rights in a fair and reasonable manner, nor shall these rights be used in a manner which would deprive any Employee of his employment, except for just cause.

#### 6.05 Bulletin Boards

The County recognizes the right and need of the Union to be able to post notices of meetings and other legitimate Union activities at various sites wherein members of the Union are assigned or employed.

## 6.06 <u>Seniority Lists</u>

County shall maintain a list showing the date upon which each Employee's service commenced. The County shall forthwith provide the Union, upon request, with an up-to-date list and shall, in any event, post an up-to-date list on all bulletin boards

where the Employees work on July 1st and January 1st in each year.

## 6.07 Union Officers-Stewards

The County recognizes the right of the Union to appoint "Employees" as Union Stewards who are to be recognized as official representatives of the Union.

# 6.08 Union Membership

Each person who becomes employed in positions covered by this Agreement shall automatically become a member of the Union as a condition of employment.

## 6.09 Monthly Staffing Reports

- (a) The Employer agrees to remit to the central office of the Union, the amounts equal to the dues that have been deducted from the pay of all Employees by the first working day after the fifteenth (15th) calendar day in the following month. Where an accounting adjustment is necessary to correct an over or under payment of dues, it shall be effected in the succeeding month.
- (b) The County recognizes the need of the Union to have required information concerning Employees eligible for membership in the Union and hereby agrees to submit on a monthly basis between the sixteenth (16th) day and the last day of each month, the following information: the name, address, phone number, status (regular or temporary), Identification number, classification, starting date and position filled by any new Employees In the bargaining unit. assigned to Public Works Operations, Vehicle and Equipment Services, Materials Management, Water and Wastewater Operations and eligible for membership in the Union.

The Employer shall provide the Union with a monthly list identifying each Employee, including new or terminated Employees. The list will include Employee name, address, identification number, starting date, classification, and dues deducted for that calendar year.

(c) The Employer recognizes the importance of new Union Employees receiving an orientation to the Union. To ensure this occurs, the Chapter Executive will be given a list of all newly hired Employees and a representative of the Union shall, within thirty (30) calendar days of the Employees' date of hire, make a presentation to the Employee(s) with respect to the structure of the Union as well as the rights, responsibilities and benefits of membership in the Union.

# ARTICLE 8 - JOB DESCRIPTIONS CLASSIFICATIONS & RECLASSIFICATIONS

8.01 The Employer shall make the Employee's job description available.

## 8.02 New or Amended Classifications

The County Administration shall give written notice to the Union of the establishment of any new or amended job descriptions and/or salary classification

6

levels (Schedule A). If it is established that a higher salary level is appropriate, the new salary shall be retroactive to the date of the classification was created. Where Union and management representatives through the Liaison Committee are unable to agree within thirty (30) days on either the correct classification of a revised job description or on the compensation level of a new job description, the matter may be advanced to arbitration for the establishment of the compensation for the individuals for which the job description has changed. If a wage change results, the effective date of change shall be determined by the Arbitrator. Both the Employer and the Union will share the cost of the Arbitrator equally. The Arbitrator will be guided by:

- (a) Evidence and submissions of the parties;
- (b) The salary levels in the Collective Agreement;
- (c) The complexity, working conditions and knowledge requirements of the other positions in the Collective Agreement at the time that the new position is established.

However, the Arbitrator shall not be restricted to the existing salary levels.

## 8.02 Classification Reviews

An Employee may at any time bring to management for review, changes that may have occurred to his existing job duties. The Employee/Union, may bring such issues forward to Liaison Committee.

#### 8.03 Reclassification

When it is necessary to re-classify Regular Employees, the County may re-classify the affected Employees on the basis of Branch seniority, qualifications and ability.

- (a) An Employee to be re-classified may displace any Employee in a lower classification within that Branch that has less Branch seniority providing that the Employee being reclassified is qualified to perform the duties of that position;
- (b) Employees with special qualifications may be exempt from re-classification.
- 8.04-12.02 In those cases where re-classification according to Clause 12.01 8.03 have occurred and vacancies become available in the affected job classification, those Employees who were previously re-classified downward from that classification, shall be returned to that classification on the basis of their seniority so long as that Employee has maintained the ability and qualifications to be appointed to the position.
- 8.05 12.03 An apprentice who begins and completes their apprenticeship with the Employer shall be placed on level 2 of the appropriate trades rate upon confirmation of certification.

## ARTICLE 9 - POSTING OF POSITIONS, PROMOTIONS AND TRANSFERS

- 9.01 When a position of a Regular or Temporary nature is to be filled, notice of the vacancy shall be posted as soon as reasonably possible for a minimum of one (1) week. The position can be filled by appointment if it is to be filled for a period of less than ninety (90) calendar days.
- 9.02 If a position with the same job code becomes vacant within six (6) months of the original closing date, the Employer may consider candidates from the original recruitment process, without re-posting. This is not applicable if the original posting was for a temporary position and the vacancy is for a permanent position.
- 9.02 3 Such notice will contain the following information: date of the posting, nature of position, qualifications, required knowledge and education, skills, shift, hours of work, experience, wage rate or range, consistent with the job description.
- 9.03 4 Where there are qualified Union candidates for a posted position, their applications shall be considered first.
- 9.04-5 Where qualified applicants have applied and a suitable candidate has been selected, the Employer shall appoint a candidate to the position within thirty (30) days of the competition closure.
- 9.05 6 Preference in the filling of positions shall be given to Employees of the County over all other persons, where the Employee has the required skills and ability to perform the tasks of the position. When filling positions or making promotions, qualifications, experience, skills, ability and seniority shall be the governing factors. When filling positions, employees of the County shall be given first preference.

## 9.06 7 Trial Period

An Employee who is transferred or promoted to another position within this Agreement shall be confirmed in that other position within three (3) months unless the Employee wishes otherwise or proves unsatisfactory in the new position, in which case the Employee will be returned to his former position. Any other Employee promoted or transferred because of the displacement of positions shall also be returned to his former position. In any event, Employees promoted or transferred shall be paid at the rate applicable to the new position from commencement of employment in the position until returned to the previous position.

- (b) In making promotions and filling vacancies, appointments will be made on the basis of qualifications, experience, ability and seniority.
- 9.08 The posting requirements in this Article may be waived by mutual agreement in cases of medical accommodation.

## **ARTICLE 10 – SENIORITY**

10.01 For the purpose of Clauses 10.01 to 10.03 **10.10**, 12.01 to 12.03 **12.02**, and 15.01 to 15.06 **15.05**, seniority shall be defined from the date of continuous service employment in any Branch covered by this Agreement.

In accordance with 1.01(k) seniority commences at date of hire, and shall include all continuous and accumulated service as a temporary Employee.

- 10.02(a) Employees who were employed by the County prior to January 1, 1979 and entered in any classification covered by this Agreement on that date shall have their seniority established as the date of continuous service employment with the County.
- 10.023(b) Seniority shall be retained and accumulated during absences resulting from illness or an accident where the Employee is entitled to compensation, suspension with or without pay less than sixty-one (61) days, approved leave without pay less than sixty-one (61) days.
- 10.034 (c)——Seniority shall be retained during absences resulting from; layoff (up to eighteen (18) months), suspension with or without pay greater than sixty (60) days, approved leave without pay greater than sixty (60) days, strike or lockout.
- 10.045 Where Employees have been laid off and recalled, they shall retain the seniority held at the date of layoff. The Union shall be notified of all recalls.
- 10.056(d) An Employee who accepts an internal permanent position outside this Agreement shall retain the seniority accrued to that time for a period of six (6) months.
- Accumulation of seniority shall recommence from the date the Employee returns to a classification covered by this Agreement and their seniority date shall be adjusted by the period of their absence. This retained seniority shall not be effective in exercising any seniority rights until the Employee has been back in the scope of this Agreement for six (6) months.
- 40.03-10.07 An Employee shall lose their seniority and their name shall be removed from the seniority list for any of the following reasons: voluntary resignation, discharge for cause, layoff for more than eighteen (18) months, retirement, or when an Employee has been employed in a position outside this Agreement for a period of six (6) consecutive months or more.

#### **ARTICLE 11 - PERIOD OF PROBATION**

11.01 The probationary period for new Employees covered by this agreement in both permanent and temporary positions shall be five hundred (500) hours one thousand (1000) hours worked. , with the exception of Employees appointed to the positions listed below, whose probationary period will be one thousand (1000) hours worked:

3008	Public Works Technician	<del>3015</del>	<del>Operator I</del>
------	-------------------------	-----------------	-----------------------

3009	Public Works Technician II	3016	Operator II
3012	Truck Driver I	3017	Operator III
3013	Truck Driver II	3019	Operations Foreman
3014	Truck Driver III		

3025	<del>Partsman I</del>	3030	<del>V&amp;E Chargehand</del>	
<del>3026</del>	<del>Partsman II</del>	3040	V&E Leadhand	
3027	Welder	3041	Transit Chargehand	
3062	Fleet Services Technician	<del>3042</del>	Transit Leadhand	

3021	W&WW Foreman	<del>3052</del>	Utilities Electrician		
3038	W&WW Leadhand	<del>3053</del>	Utilities Instrument Technician		
3047	Infrastructure Foreman	<del>3051</del>	Ut.Inst Tech/Electrician Dual		
<del>3039</del>	Project Foreman	<del>3063</del>	Control Systems Technician		
3043	Utility Operator I	3020	W&WW Operator (Collection and Distribution)		
3044	Utility Operator II	3049	W&WW Operator (Transmission)		
3045	Utility Operator III	3037	Wastewater Infrastructure Operator		
3046	Utility Leadhand				

If the Employer identifies performance concerns, written feedback shall be provided to the Employee. Where circumstances warrant an extension, the probationary period may be extended up to an additional two hundred and fifty hours (250) hours upon mutual agreement of the Union and the Employer. may be extended up to an additional 500 hours worked, at the Employer's discretion. The Employee and the Union shall be notified of the extension and the cause for the extension.

An Employee who has previously been employed in a position covered by this agreement will have such previous employment considered part of the probationary period provided that a break in service is one (1) year or less and the Employee is returning to the same job.

11.03 At any time, an Employee, categorized as probationary, may be released by the Department Head or the County Administration. Such notice of release will be formalized in writing and will give the reason for the Employee being released as

{00139306} 10

11.02

soon as practicable following at the time of the notification of the release.

- 11.04 Employees shall provide at least ten (10) working days written notice of their intention to resign.
- In the event that an Employee terminates his employment with the County without submitting the written notice in compliance with Clause 11.04, the Employee shall not be entitled to receive his regular cheque and/or his final cheque until the next regular distribution of regular wage and salary cheques.
- 11.06 4 A Probationary Employee in accordance with Articles 11.01 shall be granted all of the terms and conditions of this Agreement. During the probationary period the County may terminate the employee. The employee or the Union may grieve the termination, but the decision at Level 3 shall be final and binding and the matter shall not be referred to arbitration under the provisions of this Agreement.

however the termination of a Probationary Employee shall not be have access to the application of any grievance procedure for termination of employment as a result of:

- (a) Unsatisfactory probationary performance; or
- (b) The abolishment of the position occupied by the Probationary Employee

#### **ARTICLE 12 - RECLASSIFICATION**

- 12.01 When it is necessary to re-classify Regular Employees, the County may re-classify the affected Employees on the basis of Branch seniority, qualifications and ability.
  - (a) An Employee to be re-classified may displace any Employee in a lower classification within that Branch that has less Branch seniority providing that the Employee being reclassified is qualified to perform the duties of that position;
  - (b) Employees with special qualifications may be exempt from re-classification.
- In those cases where re-classification according to Clause 12.01 have occurred and vacancies become available in the affected job classification, those Employees who were previously re-classified downward from that classification, shall be returned to that classification on the basis of their seniority so long as that Employee has maintained the ability and qualifications to be appointed to the position.
- An apprentice who begins and completes their apprenticeship with the Employer shall be placed on level 2 of the appropriate trades rate upon confirmation of certification.

#### **ARTICLE 14 - GRIEVANCES**

## 14.01 Statement of Intent

The Employer and the Union recognize the desirability of resolving differences through joint consultation and **respectful and open** discussion. Both will attempt to resolve differences through **exploring interests and options together informally** means where possible, prior to proceeding to formal **process**.

## 14.02 <u>Definitions and Scope</u>

- (a) A Personal grievance is a dispute regarding:
  - (i) Alleged unjust treatment or discrimination
  - (ii) Alleged unfair working conditions
  - (iii) Any disciplinary action
  - (iv) Any concern regarding the interpretation, application, operation or alleged violation of this Agreement, including any questions as to the arbitral nature of the matter.
- (b) Grievances which involve a dismissal or a suspension will commence at level II and shall be initiated in writing within fourteen (14) days of the date when the Employee first became aware of the disciplinary action.
- (c) A Policy Grievance is a dispute between the parties which due to its nature is not properly subject of a Personal Grievance and concerns the interpretation, application, operation or alleged violation of this Agreement, including any questions as to the arbitral nature of the matter.
- (d) Formal personal grievances shall be initiated within fourteen (14) calendar days from the date of the incident prompting the grievance. Policy grievances shall be initiated either by the Union or the County in writing, within thirty (30) calendar days from the day that the matter giving rise to the grievance came to the attention of the Designated Officer of the Union or Official of the Employer.
- (d) Formal grievances shall be in writing (email will suffice) and must state the general nature of the grievance, identify the specific clauses in the collective agreement that are being violated, and provide details with respect to the redress or remedy being sought.
- (f) "Days" means calendar days.
- (g) A Union member submitting a grievance has the right to be assisted by an attending Union Representative or Union Steward.

## 14.03 <u>Grievance Process</u>

A Personal Grievance shall be initiated by the Employee as follows:

#### Level I

The grievance shall be taken up with the Employee's manager or designate.

an Employee wishes to pursue a Personal Grievance, the Employee and/or Union Representative shall advise the Employee's manager or designate that they wish to initiate a Level I grievance. The Employee and/or the Union representative will discuss the subject of the proposed grievance with the Employee's manager or designate within fourteen (14) calendar days of the receipt of the grievance. date upon which the subject of the grievance, in an attempt to resolve the matter. The manager or designate first level classified supervisor will provide a response to the Employee and/or Union Representative within seven (7) calendar days from when the proposed grievance was discussed.

#### Level II

If the Employee continues to be dissatisfied, he should submit the grievance in writing within fourteen (14) days of receiving the response from Level I, to the Department Director. The grievance shall state the general nature of the grievance, the Articles in the Collective Agreement alleged to have been breached and the redress or the remedy sought. The Director, in consultation with Human Resources, will hear the matter and will respond in writing to the Grievor within seven (7) calendar days.

#### Level III

If the Union does not accept the decision of the Department Director, they may request in writing within seven (7) days of receiving the Department Directors decision, that the Chief Commissioner review the grievance. The Chief Commissioner shall hear the matter within ten (10) days and will advise the Union of his decision in writing within fourteen (14) days of the hearing;

The Chief Commissioner may designate an Associate Commissioner to hear and decide on Level III grievances;

If the grievance is not settled as a result of the Chief Commissioner's decision, the Union **or the Employer** may refer the matter to arbitration pursuant to the Labour Relations Code.

# 14.04 <u>Arbitration Board</u>

In the event the matter being grieved is presented to an Arbitration Board, each Party shall bear the expense of its respective nominee and shall bear one-half (1/2) of the expenses of the Chairman of the Arbitration Board.

- 14.05 The Parties may mutually agree to have the matter heard by a single Arbitrator.
- 14.06 The Arbitration Board by its decision shall not alter, amend or otherwise change the terms of this Collective Agreement.
- 14.07 All Union members submitting a grievance have the right to be assisted by an attending Union Representative.

## 14.08 Procedure and Time Limits

Extension of Time Limits: Time limitations set out in the above grievance procedure may be extended by mutual agreement between the parties and must be in writing.

Time Limits: Time limits and procedures contained in this grievance procedure are mandatory. Failure to pursue a grievance within the prescribed time limits and in accordance with the prescribed procedures shall result in abandonment of the grievance and all rights and recourse to the grievor.

Failure to reply to a grievance in a timely fashion shall advance the grievance to the next level. Grievances so advanced shall be subject to time limits as if a reply had been made on the last allowable day of the preceding level in the procedure.

14.09 Policy Grievances may be initiated either by the Union or the County in the following manner:

- (a) Policy Grievances shall be initiated either by the Union or the County in writing, within thirty (30) calendar days from the day that the matter giving rise to the grievance came to the attention of the Designated Officer of the Union or Official of the Employer. The written notice of the grievance shall be submitted to the President of the Union or the Director, Human Resources of Strathcona County and shall stipulate the general particulars of the grievance together with reference to the allegedly violated clause of the Agreement;
- (b) Grievances submitted in accordance with Clause 14.09 shall be referred to a Grievance Committee which shall endeavour to resolve the difference. This Committee will be composed of two representatives of the County and two members of the Union appointed by the Executive. This Committee shall meet within fifteen (15) calendar days of the submission of the written notice by either party;
- (c) After the parties have met in accordance with the preceding clause, each party will give any decision made in writing to the other party within fifteen (15) calendar days. In the event that the two parties reach a mutually acceptable agreement concerning the matter in question, a written and signed notice of mutual agreement will supersede the aforementioned notice of decision;
- (d) In the event that the parties do not reach a mutually acceptable agreement, or that either party does not accept the written notice of decision of the opposite party within thirty (30) calendar days of the Chief Commissioner's decision as per Clause 14.03 either party may notify the other party in writing of its intention to submit the grievance to arbitration pursuant to the provision of the Labour Relations Code.

14

#### ARTICLE 16 – HOURS OF WORK AND OVERTIME

The prescribed regular hours of work for an Employee, other than those referred to in Clause 16.04-5 following, shall be eight (8) hours in any one (1) day or forty (40) hours in any one (1) calendar week or five (5) working day period. Hours worked in excess of the above shall be calculated as overtime hours and shall be paid accordingly. Notwithstanding the above, when an established schedule provides a pattern that is irregular, the forty (40) hours in one (1) calendar week or five (5) working day period shall mean average over a two (2) week period.

## 16.02 Overtime

The overtime pay rates shall be calculated at the rate of two (2.0) times the regular pay rate for all hours worked beyond eight (8) hours in a day.

- An Employee may bank overtime hours at the rate it is earned; up to the maximum of forty (40) regular hours within a calendar year. An Employee may request to be paid out for banked hours up to the time equivalent of forty (40) regular hours per calendar year. Banked hours may be taken as time off by Employees after receiving written authorization from the County designated official who shall consider the wishes and application of the Employee. Up to forty (40) banked regular hours may be carried forward into the next calendar year. Banked hours cannot exceed the time equivalent of eighty (80) regular hours at any one time.
- 16.04 For those Employees who are engaged in road construction, road maintenance or the removal of snow from roads or any of those undertakings, for the purpose of calculating overtime, their regular day shall consist of a twenty-four (24) hour period beginning at midnight. The maximum number of hours of work at regular rates of pay shall be those prescribed or allowed under the Employment Standards Code and Regulations thereunder, but shall not exceed:
  - (a) Ninety (90) hours bi-weekly, except where permitted under the Employment Standard Code;
  - (b) Eighty (80) hours bi-weekly during the period November 01 to April 30, inclusive;
- Only hours actually worked, paid vacation (including general holidays) and banked time used to supplement hours are used in accumulation of overtime. Sick leave or other paid or unpaid absences are not included.
- 16.06 For those Employees specified in Clause 16.04 from May 1 to October 31, overtime rates shall be calculated at the rate of two (2.0) times the regular pay rate for all hours worked beyond nine (9) hours in a day. This article only comes into effect when actually scheduled to work nine (9) hours a day.

15

## 16.06 <u>Posting of Shift Schedules</u>

Barring unforeseen circumstances and emergency situations, shift schedules shall be posted no less than thirty (30) days before the expiry of the current schedule.

# 16.07 Shift Changes

- (a) When an Employee is required to change from one shift to another shift, the Employer will provide at least twenty-four (24) hours written notice in advance of the start time of the originally scheduled shift and ten (10) hours off between shifts.
- (b) If the twenty-four (24) hours written notice and ten (10) hours off between shifts is not provided to the Employee, the Employee will be paid at the rate of two (2.0) times the regular rate of pay for the hours of the first shift worked.
- (c) If the Employee is on regular scheduled days off when the twenty-four (24) hours written notice is posted, the supervisor may contact the Employee directly to provide verbal twenty-four (24) hours notice change in shift.
- An Employee shall be paid for or expected to work for a minimum of three (3) hours if the regular shift is cancelled or rescheduled and the Employee was not notified of such cancellation or rescheduling on or before the day twelve (12) hours prior to the cancelled or rescheduled shift.
- 16.09 When an Employee calls in ill for a work period affected by the preceding Clause 16.08, that Employee will be paid only for the minimum of three (3) hours that he would have been paid had he reported for work in the regular fashion. This clause shall not apply in cases in which the Employee provides proof of attendance at either medical, dental or optical appointments. In such cases, the Employee will be paid for either the half-day or full-day as appropriate to the circumstances and as provided in Clause 21.03 and 21.04.

## 16.10 Early Reporting

- (a) When an Employee is requested prior to the conclusion of a shift, to report or work more than two (2) or more hours earlier than his regularly scheduled shift, that Employee may (subject to operational requirements) be given a choice of leaving work prior to the end of their regularly scheduled shift or being paid the regular overtime rate for those hours worked prior to the time scheduled for the commencement of the regular shift. The total of the early reporting hours worked and regular shift hours worked will not be less than the number of regularly scheduled hours. Such early reporting arrangement may continue for two consecutive shifts.
  - If requested to report early on the second shift the early reporting hours will be at the overtime rate. If requested for a third consecutive shift, Clause 16.07 will apply.
- (b) When an Employee is requested prior to the conclusion of a shift to report up to two (2) hours earlier or up to two (2) hours later for the next regular shift, that Employee shall not receive the overtime rate for the earlier or later reporting period. The total of the earlier reporting hours worked and regular shift hours worked will not be less than the number of regularly scheduled hours. If requested to report earlier or later on the second shift the earlier or later reporting hours will be at the overtime rate. If requested

for a third consecutive shift, Clause 16.07 will apply.

# 16.11 <u>Attendance</u>

When an Employee is unable to report to work, the Employee shall follow the established protocol for their department to report such absence, and at a minimum shall report the absence one (1) hour prior to the start of the shift. within one hour prior to the scheduled shift start time. Failure to comply with this requirement shall result in loss of pay for the shift, and may result in discipline unless the Employee provides a reasonable explanation.

#### 16.124 Work Break

All Employees shall be entitled to one fifteen (15) minute work break in each half day worked, or in any overtime work period in excess of two (2) hours. Such work break shall be taken at the work site and the time of such break shall be approved by the immediate supervisor of each Employee. The work break shall be considered part of a work hour for the purposes of the preceding paragraphs.

## 16.13 Sand Shift Workers and Sidewalk Crew

Notwithstanding Clause 16.01, when the needs of the operations require it, Sand Shift Workers and Sidewalk Crew may be required to work a modified shift schedule of up to twelve (12) hours. Modified hours of work shall commence annually on the first full pay period commencing on or after October 15 and shall continue until the first full pay period before or after April 15 based on operational need.

- (a) Overtime will be paid to Sand Shift Workers and Sidewalk Crew who exceed 80 hours of work in a bi-weekly pay period.
- (b) Employees working modified hours of work will have all benefits and entitlements which are expressed in terms of daily or weekly entitlement converted to produce the equivalent hours of benefits and entitlements as they would have had if the hours of work had not been modified. This will result in no loss or gain in Employee benefits and entitlements.

#### ARTICLE 17 - SHIFT AND WEEKEND PREMIUMS

- 17.01 (a) A shift premium of one dollar and forty cents (\$1.40) one dollar and ninety cents (\$1.90) per hour shall be paid to all Union Employees for those hours worked between seventeen hundred (1700) hours and zero seven hundred (0700) hours Monday to Friday and for all hours worked on Saturday and Sunday and holidays.
  - (b) At no time shall shift premium be included with the Employee's regular rate of pay for purposes of computing overtime payments, other premium payments, or any Employee benefits.

#### ARTICLE 20 - RELIEF AND ACTING INCUMBENT PAY

## 20.01 Relief & Acting Pay

Any Employee who works in a classification (within the Bargaining Unit) with a higher rate of pay in any one (1) shift shall receive the higher rate of pay for each hour worked in the higher classification. If this time period exceeds ninety (90) days, effective the ninety-first (91st) day, the Employee will receive the higher rate of pay for all approved vacation, general holiday, illness, bereavement, and compassionate leave while working in the higher classification.

- 20.02 Notwithstanding the premium in Clause 20.09, An Employee appointed to Leadhand or Project Foreman shall not receive an hourly rate of pay that is less than the "County Department Union Employees" that he is responsible for.
- 20.03 Certified Technicians appointed to a seasonal Field Certified Technician assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- Vehicle Serviceman II appointed to a seasonal Field Vehicle Serviceman II assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- The qualification or eligibility of an Employee to relieve or be assigned to a higher classification will be the responsibility of management.
- 20.03 An Employee who has become incapacitated by injury or illness may be employed in other work which he can do. Such Employee may not displace an Employee with more seniority.
- 20.04 An Employee who refuses without good and satisfactory reason, to accept the alternate classification or position, may be suspended immediately. The supervisor issuing the suspension will report the suspension to the Department Head (Human Resources) as soon as possible who will notify the Union Executive.
- 20.05 Any Employee affected by the preceding Clauses 20.03 and 20.04 and not satisfied with the action recommended in accordance with Clause 20.04, may appeal the reclassification or suspension in keeping with Clause 14.03 of this Agreement.
- 20.06 Should an Employee who has completed the probationary period be unable to perform his usual duties by reason of the loss or suspension of his motor vehicle operator's license or a similar happening, that Employee may be placed in some other suitable position until the disability ends or is removed. The wages paid during the period of such disability shall be wages appropriate to the position in which the Employee is placed.
- 20.07 Notwithstanding the above clause, within fourteen (14) days of the suspension, an Employee may be allowed to book unused holidays at the wage prevailing prior to suspension of his operator's license.

## **Acting Incumbent**

- 20.08 Any member of the Union may be appointed for one (1) or more days as a **Classified** supervisor and such appointment will not cause the member to be disqualified from the Union.
- 20.09 If any Employee is appointed temporary Classified Supervisor, they shall receive a premium of two dollars and twenty-five cents (\$2.25) per hour in addition to their regular wage rate for one (1) complete shift or longer.
- 20.10 Any member of the Union may be appointed for a period of time to the position of temporary Classified Supervisor and such appointment will not cause the member to be disqualified from the Union.
- 20.11 Notwithstanding the premium in Clause 20.09, an Employee appointed to Leadhand or Project Foreman shall not receive an hourly rate of pay that is less than the "County Department Union Employees" that he is responsible for.
- 20.12 Certified Technicians appointed to a seasonal Field Certified Technician assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- 20.13 Vehicle Serviceman II appointed to a seasonal Field Vehicle Serviceman II assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.

#### ARTICLE 21 - PROVISION FOR ILLNESS AND NORMAL ABSENCE

- 21.01 Sick leave benefits as herein stated are sponsored by the County to protect the Employee in the event of an unavoidable illness or injury not covered by the Alberta Workers' Compensation Act.
- 21.02 When an Employee is unable to report to work, the Employee shall follow the established protocol for their department to report such absence within one hour prior to the scheduled shift start time. Failure to comply with this requirement shall result in loss of pay for the shift, and may result in discipline unless the Employee provides a reasonable explanation. Moved to Article 16
- A Regular Employee, upon hire, shall be entitled to a maximum of eighty (80) hours ten (10) working days per calendar year of leave with pay for any casual illness, which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less. Up to sixteen (16) hours two (2) days of unused casual illness leave per calendar year shall be carried over and may be used in the next calendar year. The maximum paid casual illness leave in a calendar year is ninety-six (96) hours twelve (12) days including the carry over.
- 21.04 A Temporary Employee after the completion of the standard probationary period,

shall be entitled to a maximum of thirty-two (32) hours four (4) working days per calendar year of leave with pay for any casual illness, which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less. Temporary Employees with more than six (6) months of service in the current calendar year shall be entitled to an additional sixteen (16) hours two (2) working days in the current calendar year, of leave with pay for any casual illness. Up to two (2) days of allotment can be used to attend to illness of immediate family members.

- 21.05 Each day or portion of a day of absence due to casual illness used within a calendar year, shall be deducted from the remaining casual illness leave entitlement for that calendar year.
- NEW Where an employee terminates and is rehired in the same calendar year, the casual illness allotment will be the amount that they had in their available bank at the time of departure.
- In any case, an Employee may be required to provide satisfactory proof of illness upon return to work, where it can be shown that reasonable doubt exists in respect to the purpose of an absence claimed to be due to illness. The employee will be provided the reasons for the request. Such proof may take the form of a medical certificate signed by a qualified physician, dental practitioner or other professional as deemed appropriate by the Disability Management Supervisor/Human Resources.
- 21.07 Upon approval (which will not be unreasonably denied), time off of two (2) hours or less for doctor appointments will result in no loss of casual illness or regular time. The Employer may require the Employee to submit proof of attendance at a medical, dental or optical appointment when time off work is granted to attend such appointments.
- 21.08 Each Regular Employee will be entitled to leave of absence with pay for General Illness in accord with the following scale. General Illness is defined as "any illness which causes an Employee to be absent from work for a period of more than three (3) consecutive work days".

To obtain General Illness benefits resulting from the inability of the Employee to perform the duties of his position, a proper medical certificate(s) signed by a qualified physician, dental practitioner or other professional as deemed appropriate by the Disability Management Supervisor/Human Resources is required.

# 21.09 <u>Family Illness Leave</u>

A Regular Employee shall be entitled to a maximum of twenty-four (24) hours three (3) working days per calendar year with pay to attend to illness of immediate family

members as per defined in Clause 22.06.

- 21.10 A Temporary Employee may use up to sixteen (16) hours two (2) days of their casual illness allotment, as per Clause 21.04 can be used to attend to illness of immediate family members.
- 21.0911 Effective the calendar year following ratification, Sick leave entitlement for illness longer than three (3) working days shall be as follows:
  - (a) After ninety (90) days three (3) months five hundred (500) hours of continuous employment the Employee shall receive one hundred sixty (160) hours twenty (20) working days at the base rate of pay;
  - (b) On the first day worked, in the next calendar year sick leave entitlement shall be increased to a maximum **two hundred eighty (280) hours thirty**-five (35) working days at the base rate of pay
  - (c) On the first day worked, in subsequent consecutive calendar years sick leave entitlement shall be increased to a maximum of **five hundred twenty** (520) hours sixty five (65) working days at the base rate of pay, plus any unused casual illness up to a maximum of **six hundred** (600) hours seventy five (75) working days.
- 21.<del>1012</del> When a day designated as a Paid Holiday under Clause 23.01 falls within a period of General Illness, it will be considered as a paid holiday.
- 21.1213 Transfers, promotions, demotions and/or reclassification shall not affect sick leave entitlement of any Employee.
- 21,<del>1114 Workers Compensation</del>
  - a) An Employee is not eligible to receive sick leave benefits if the absence is due to an injury while in the employ of any other Employer that is covered by Workers' Compensation Benefits.
  - b) Employee's will be eligible to receive benefits in accordance with the County's policies.
- 21.12 Transfers, promotions, demotions and/or reclassification shall not affect sick leave entitlement of any Employee.
- 21.1315 Sick leave entitlement will not be affected while the Employee is in receipt of Alberta Workers' Compensation benefits.
- 21.1416 Regular Employees who are eligible to receive Workers' Compensation benefits shall receive full pay from the County for a period not to exceed six (6) months provided that all Workers' Compensation payments are assigned to the County for that period.

- 21.4517 Expenses for medical certificates will be the responsibility of the employee, including medical certificates required to support eligibility for casual or general illness. Where the Employer has requested the employee to attend a medical examination and/or obtain additional medical documentation, the Employer shall reimburse the employee for the cost of the documentation (if any).
- 21.<del>16</del>18 An Employee who has become incapacitated by injury or illness may be employed in other work which he can do. Such Employee may not displace an Employee with more seniority.
- An Employee who refuses without good and satisfactory reason, to accept the alternate classification or position, may be suspended immediately. The supervisor issuing the suspension will report the suspension to the Department Head (Human Resources) as soon as possible who will notify the Union Executive.
- Any Employee affected by the preceding Clauses 21.176 and 21.187 and not satisfied with the action recommended in accordance with Clause 20.04, may appeal the re-classification or suspension in keeping with Clause 14.03 of this Agreement.
- 21.1921 Should an Employee who has completed the probationary period be unable to perform his usual duties by reason of the loss or suspension of his motor vehicle operator's license or a similar happening, that Employee may be placed in some other suitable position until the disability ends or is removed. The wages paid during the period of such disability shall be wages appropriate to the position in which the Employee is placed.
- 21.<del>2022</del> Notwithstanding the above clause, within fourteen (14) days of the suspension *of* **their operator's license**, an Employee may be allowed to book unused holidays at the wage prevailing prior to suspension of his operator's license.

#### **ARTICLE 22 - LEAVES OF ABSENCE**

- Other Leaves of Absence, without pay, may be granted at the sole discretion of the County, and in accordance with the Alberta Employment Standards Code, upon written application by the affected Employee at least seven (7) days prior to the beginning date of the applied for Leave of Absence. To be eligible for such leave, an Employee shall have been employed with the County for ninety (90) days of continuous employment, with the exception of reservist leave which is twenty-six (26) consecutive weeks, and must have his application endorsed or approved by the Department Head. The application shall include the reason for the request as well as the dates of the requested Leave of Absence. Such approval should not be unreasonably withheld.
- 22.02 In unusual circumstances or by mutual agreement, the requirement stated in Clause 22.01 for application to be made seven (7) days prior to the commencement of the requested leave of absence may be waived. Application for such waiver is

to be made directly to the Department Head.

In the case of an Employee who takes a leave of absence under the provisions of Clause 22.01, the benefits and conditions to which that Employee is entitled as a result of length of service in the County shall not be adversely affected as a result of having been granted such leave of absence.

22.04 An Employee on such approved leave may not engage in other employment for gain while on such leave without the express written consent of the County.

# 22.05 Court and Jury Duty

Leave of Absence shall be provided to an Employee who is required to serve as a juror, or is subpoenaed as a witness in a private capacity in any court proceeding. The Employee shall receive their regular rate of pay and shall remit to the Employer any allowance they may receive for such duty or service.

If the Employee is subpoenaed as a witness to appear in court in their official capacity to give evidence or to produce Strathcona County records, the Employee shall receive their regular rate of pay during regular working hours and is eligible to receive overtime pay to appear during non-work hours. The Employee shall remit to the Employer any allowance they may receive for such duty or service.

The Employee shall present proof of such duty or service and the amount of remuneration or allowance received in all cases.

Employees required to appear in court in a capacity other than subpoenaed as a witness or jurist will be granted leave without pay.

## 22.06 Bereavement Leave

An Employee shall be granted Bereavement Leave of Absence with pay for not more than five (5) consecutive days because of the death of spouse (includes common-law), child (includes step, foster child or ward), parent, (includes step or foster parent), brother, sister, parent of spouse, guardian, son-in-law, daughter-in-law, grandchild or any other relative who is a member of the Employee's household, or the spouse of any of the foregoing. In addition to the above, the Employer may grant up to one (1) day travel to attend the funeral if the travel is five hundred (500) kilometres or more away.

- 22.07 Leave of absence for one (1) full day, with pay, shall be granted at the time of to attend-the funeral or death of a grand-parent, grandparent-in-law, brother-in-law or sister-in-law, aunt, uncle, niece or nephew, if the funeral is on a work day.
- 22.08 Upon written application, additional Compassionate or Bereavement Leave of Absence with pay, due to exceptional circumstances, may be granted an Employee at the discretion of the Chief Commissioner.
- 22.09 One (1) day of leave may be granted without loss of pay or benefits to attend as a pallbearer or mourner for a deceased Employee. Where the family of a deceased Employee requests pallbearers from the Union, the County may grant the

necessary leave with pay for up to six (6) pallbearers.

# 22.1122.10 Family Illness Leave

A Regular Employee shall be entitled to a maximum of twenty-four (24) hours three (3) working days per calendar year with pay to attend to illness of immediate family members as per defined in Clause 22.06.

22.11 A Temporary Employee may use up to sixteen (16) hours two (2) days of their casual illness allotment, as per Clause 21.04 can be used to attend to illness of immediate family members.

# 22.10 Compassionate Leave

An Employee shall be granted Compassionate Leave of Absence with pay for not more than five (5) consecutive days because of the critical illness of a member of his immediate family as per defined in Clause 22.06 or any other relative who is a member of the Employee's household. The Employee shall provide the County with a letter from the attending physician concerning the fact of the critical illness.

Any additional Compassionate Care Leave shall be provided in accordance with *Alberta Employment Standards Code*.

- 22.11 A Regular Employee shall be entitled to a maximum of three (3) working days per calendar year with pay to attend to illness of immediate family members as per Clause 22.06.
- 22.12 One (1) day leave with pay will be allowed for an Employee to be present at the birth or adoption proceeding of an Employee's child, providing such occurrence is on a scheduled work day.

## 22.<del>13</del>11 Maternity Leave

Maternity Leave will be granted to Employees in accordance with the Alberta Employment Standards Code.

- (a) Employees must have been employed with Strathcona County for ninety (90) days of continuous employment.
- (b) The maternity leave is not more than sixteen (16) weeks starting at any time during the thirteen (13) weeks immediately before the estimated date of delivery.
- (c) The Employee must give six (6) weeks' written notice of the date she intends to begin leave and four (4) weeks' notice of the date she intends to return to work.
- (d) If an Employee's pregnancy results in a miscarriage or stillbirth within sixteen (16) weeks of their due date, they are entitled to sixteen (16) weeks of maternity leave.

# 22.124 <u>Health Related Portion of the Maternity Leave</u>

During the one-week Employment Insurance waiting period, Strathcona County will top-up salary to ninety-five percent (95%) of the Employee's regular earnings. Verification that Employees are in receipt of employment insurance benefits is necessary to provide the top-up benefit. A medical certificate must be provided to Human Resources, confirming the date the child was born and the duration of the leave that the physician indicates is health related. During the health-related portion of the maternity leave Strathcona County will top-up salary to ninety-five percent (95%) of the Employees regular earnings. The benefit coverage that the Employee has at the time of her leave will continue during the health related portion of the absence. At the end of this period, parental leave will commence, and benefit coverage will continue if the Employee pays both Employer and Employee portions of the cost of the premiums.

22.135 When a pregnancy interferes with the performance of duties or the Employer has a concern for the Employee's health, the Employer may reassign the Employee's duties at no loss of pay, or by notice in writing to the Employee, require them to commence sick leave under Clause 21.08.

## 22.1416 Parental Leave

Employees who have been employed with Strathcona County for ninety (90) days of continuous employment are entitled to parental leave without pay for a maximum of sixty-two (62) weeks. The Employee must give at least six (6) weeks written notice of the date they intend to begin parental leave and four (4) weeks' notice of the date they intend to return to work. During that time they may receive benefit coverage by paying the Employee and Employer portions of the cost of the premiums.

22-1517 Employees wishing to return to work prior to the agreed return date may do so with two (2) weeks' notice.

## 22.1618 Adoption Leave

One (1) day leave with pay will be allowed for an Employee to be present at the birth or adoption proceeding of an Employee's child, providing such occurrence is on a scheduled workday.

The Employer will grant adoption leave to Employees who are adopting a child. The Employee must have been employed with Strathcona County for ninety (90) days of continuous employment. Written notice of leave must be provided to the Employer at least two (2) weeks before Employees can reasonably expect to obtain custody of the child.

- 22.**17**19 If Employees cannot provide at least two (2) weeks written notice, they must notify the Employer immediately after receiving notice of the adoption.
- 22.1820 Adoption leave consists of not more than sixty-two (62) consecutive weeks of leave without pay within seventy-eight (78) weeks after the child is placed with the

adoptive parents. During that time they may receive benefit coverage by paying the Employee and Employer portions of the cost of the premiums.

22.1921 Only one (1) parent of a child referred to in Clause 22.20 will be granted adoption leave under this clause.

## 22.2022 Shared Leave

If both parents are Employees of Strathcona County parental or adoption leave may be taken wholly by one of the Employees or be shared by both. Only one (1) parent will be granted parental or adoption leave at a time.

# 22.2123 Anniversary Date

Employees who have been on maternity, parental or adoption leave will have their anniversary date adjusted for evaluation and increment purposes only. They will not earn vacation leave during this period, the exception being for maternity health related reasons.

## 22.2224 Additional Leave

Leave of absence without pay beyond the approved maternity, parental or adoption leave may be granted upon written application up to a combined maximum of twelve (12) months at the discretion of the Employer.

# 22.23<del>25</del> <u>Domestic Violence Leave</u>

Domestic violence leave shall be granted in accordance with the *Alberta Employment Standards Code*.

# 22.2426 Military Leave

Military leave shall be granted in accordance with the *Alberta Employment Standards Code.* 

## 22.25<del>27</del> <u>Citizenship Ceremony Leave</u>

Citizenship Ceremony Leave shall be granted in accordance with the *Alberta Employment Standards Code*.

#### **ARTICLE 23 - HOLIDAYS**

23.01 Employees shall be entitled to one (1) day's paid leave for each of the following specified or general holidays:

New Year's Day

Family Day

Good Friday

Easter Monday

Labour Day

Thanksgiving Day

Remembrance Day

Christmas Day

Victoria Day Christmas Floater

Canada Day Boxing Day

1st Monday in August Truth and Reconciliation Day

And any other day designated as a General Holiday by the Lieutenant Governor in Council under the Employment Standards Code.

- 23.02 Employees shall receive the recognized specified or general holidays for which they are eligible, with pay; or other days with pay in lieu of such holidays, or pay in lieu as approved by the County, providing they are available for work immediately preceding and following the days or an otherwise approved leave of absence, or they provide a medical certificate attesting to their illness on the day prior or following the general holiday.
- 23.03 The provisions for pay in lieu of such specified or general holidays referred to in Clauses 23.01 and 23.02 shall be based on the number of regular hours which would have been worked had that day been a regular working day.
- In the event a specified or general holiday falls on an Employee's regular day off, an additional day with pay shall be granted in lieu of that day.
- 23.05 If any of the above specified or general holidays should occur during an Employee's vacation, such vacation shall be extended by one day or an additional day with pay shall be granted in lieu of that day.
- 23.06 Temporary Employees, upon hire, shall be eligible for specified or general holidays as per Clause 23.02.
- 23.07 The Christmas Floater holiday shall be observed to give five (5) consecutive days off including the weekend as follows:
  - (a) On December 24 when Christmas Day falls on a Tuesday, a Thursday, a Friday or Saturday;
  - (b) On December 27 when Christmas Day falls on a Monday or a Wednesday;
  - (c) On December 28 when Christmas Day falls on a Sunday.

#### **ARTICLE 24 – VACATION LEAVE**

- 24.01 Annual vacation entitlement with pay shall be as follows:
  - (a) An Employee who has completed less than six (6) years of service shall earn vacation leave at the rate of four point sixty-two (4.62) hours per bi-weekly period of continuous service [fifteen (15) days per year] or five point two (5.20) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
  - (b) An Employee who has completed six (6) years but less than fourteen (14) years of service shall earn vacation leave at the rate of six point fifteen (6.15) hours per bi-weekly period of continuous service [twenty (20) days per year]

- or six point ninety-two (6.92) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
- (c) An Employee who has completed fourteen (14) years of service but less than twenty two (22) years shall earn vacation leave at the rate of seven point seven (7.70) hours per bi-weekly period of continuous service [twenty-five (25) days per year] or eight point sixty-seven (8.67) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
- (d) An Employee who has completed twenty two (22) or more years of service shall earn vacation leave at the rate of nine point twenty-three (9.23) hours per bi-weekly period of continuous service [thirty (30) days per year] or ten point thirty-nine (10.39) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week.
- If Employees become ill on an annual vacation leave for more than three (3) consecutive days-paid vacation days, the applicable days will be deemed general illness, and they will have the applicable number of vacation days reinstated upon application-and provision of a medical certificate.
- 24.03 The number of days of vacation leave which may be accumulated by an Employee is limited to the number he can earn in eighteen (18) months of continuous service.

The County will automatically pay out any excess vacation leave over the maximum accumulation at the beginning of each December.

When vacation pay is paid out rather than being taken as vacation, the payout shall be based on the Employee's rate of pay at the time of payout.

- 24.04 Vacation leave may be taken in the calendar year it is earned. However, vacation leave may not be taken until it has been earned. The vacation leave may be granted as follows:
  - (a) In one continuous period;
  - (b) As single days off or some other combination;
  - (c) If an Employee earns vacation at a rate of four point sixty-two (4.62) hours per bi-weekly period of continuous service, one (1) week vacation leave may be taken as single days off or some other combination provided the total does not exceed five work days in any calendar year;
  - (d) If an Employee earns vacation at a rate of six point fifteen (6.15) hours per bi-weekly period of continuous service or seven point seven (7.70) hours per bi-weekly period of continuous service, two (2) weeks' vacation leave may be taken as single days off or some other combination provided the total does not exceed ten (10) work days in any calendar year;
  - (e) If an Employee earns vacation at a rate of nine point twenty-three (9.23) hours per bi-weekly period of continuous service, three (3) weeks' vacation

leave may be taken as single days off or some other combination provided the total does not exceed fifteen (15) work days in any calendar year.

In addition to the automatic payout of excess vacation referred to in 24.03, an Employee may be permitted to receive a cash payment in lieu of vacation leave, provided any and all obligations have been met in accordance with Employment Standards legislation and the Employee has received written approval from the Department Head.

Temporary Employees shall have vacation pay added to their bi-weekly pay cheques and shall be entitled to unpaid vacation leave in accordance with the Employment Standards Code.

- 24.06 An Employee who does not complete one (1) full year of service shall be entitled to vacation pay in accordance with the Employment Standards Code and regulations thereunder.
  - (a) Temporary Employees will receive vacation pay at four percent (4%) of their gross regular earnings during their first four thousand (4000) regular standard hours.
  - (b) Temporary Employees will receive vacation pay at six percent (6%) of their gross regular earnings after completion of four thousand (4000) regular standard hours.
  - (c) If a Temporary Employee has a break in service greater than eighteen (18) months then vacation pay shall be paid at four percent (4%) of their gross regular earnings as per Clause 24.076 a) above.
- 24.07 Regular vacation leave shall be taken by Employees after receiving written authorization from the County or its designated official who shall consider the wishes and application of the Employee. Length of regular service Seniority with the County shall be the determining factor in resolving any dispute or conflict regarding preference for the time at which vacations are to be taken. Changes in vacation scheduling shall be authorized in writing.
- Vacation hours will be paid out at the Employee's base rate in place at the time the vacation is taken.

## **Summer Vacation Schedules**

## 24.09 Summer Vacation Schedules

Employees must submit summer vacation applications to the Director or his designate by March 15th of each year. The Employer will post schedules, as approved by the Director, by May 1st of each year.

#### **ARTICLE 25 - BENEFIT AND PENSION PLANS**

## 25.01 <u>Benefit Plans</u>

The County shall contribute on behalf of all participating Employees in group insurance and benefit plans, which provide minimum benefit levels agreed upon by the County and the Union at rates outlined in the following clauses:

- (a) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Strathcona County Benefit Plan Life, Accidental Death and Dismemberment and Long-Term Disability Insurance. All members of the Union hired after July 1, 1978, shall be members of the said plan as a condition of employment;
- (b) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Extended Health Care option of the Strathcona County Employee Benefit Plan;
- (c) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Dental option of the Strathcona County Employee Benefit Plan;
- (d) The County shall pay seventy five percent (75%) of the total premium for each eligible Union member enrolled in the Alberta Health Care Insurance Plan through the Strathcona County.

The County's liability shall be limited to the payment of premiums, where required, for the various plans and the County shall be relieved of any further liability with respect to the plans.

In the event that the County should undertake to transfer group insurance and benefit plans from one carrier to another, it is agreed that the Union will be consulted prior to the negotiation of such coverage.

## **Benefits for Temporary Employees**

#### 25.03 Benefits for Temporary Employees

To be eligible to participate in the Strathcona County Employee Benefit plans, Temporary Employees must meet the following conditions:

(a) All Temporary Employees who have been employed for twelve (12) consecutive cumulative months or more, provided there has been no break in service for longer than six (6) months, shall receive the same benefits as Regular Employees, with the exception of pension and the LTD benefit entitlement will be limited to a maximum of twenty-four months.

For the purposes of calculating the twelve (12) cumulative months, leaves of absence of more than thirty (30) calendar days shall be excluded.

(b) An exception to the above shall be made for Temporary Employees replacing Regular Employees on long term disability. A Temporary

30

Employee in this situation will be eligible for benefits after six (6) consecutive months of employment.

## Pension Plan

#### 25.04 Pension Plan

The County participates in the Local Authorities Pension Plan. Participation in this plan is compulsory for "Regular Employees" after they have completed one (1) year of service or immediately if transferring from another Local Authority Board Employer.

- 25.05 The County shall make payroll deductions as follows:
  - (a) The County shall deduct legislated Revenue Canada deductions and Local Authorities Pension Plan contributions as required from each bi-weekly pay cheque;
  - (b) The current authorized deductions for Canada Savings Bonds will be taken each pay cheque;
  - (c) The Employee contribution to the Strathcona County Employee Benefit Plan will be deducted from the first bi weekly pay cheque. each month.

#### **ARTICLE 29 - TEMPORARY EMPLOYEES**

- 29.01 Temporary Employees shall be entitled to the terms and conditions of this Agreement, except where Regular Employees are referenced. Seniority for employment periods of less than six (6) months shall not be accumulated.
- A Temporary Employee shall be considered probationary as per Article 11.01 and shall not serve probation for subsequent years of employment unless a break of service of greater than one (1) year has occurred or the Employee is returning to a different position. Temporary Employees shall have access to the application of the grievance procedure except in connection with termination of employment as a result of either:
  - (a) Unsatisfactory performance, or;
  - (b) The abolishment of the position occupied by the Temporary Employee.
- 29.03 For the purpose of determination of Pay Grade and Step a Temporary Employee returning for subsequent employment within one (1) year may be required to undertake the appropriate probationary period unless he is returning to a position in which he has previously fulfilled the probationary requirements.
- 29.04 When Temporary Employees are recalled, it shall be based on:
  - (a) A satisfactory performance evaluation from the previous year of

employment;

- (b) The Employee having qualifications to do the job for which he is recalled;
- (c) The Employee's accumulated service.

The opportunity for recall shall exist for six (6) months within the Branch.

This clause will not apply to Employees participating in the NAIT or other post-secondary Water and Wastewater Training Program.

#### ARTICLE 30 - GENERAL WAGE ADMINISTRATION

30.01 The rates of pay for all Employee members of the Union excluding Apprentices, covered by the Agreement shall be as specified in the attached Schedule A.

Temporary Employees returning for subsequent employment within one (1) year shall have their prior service recognized for the purposes of determining Pay Grade and Step.

30.02 An Employee is entitled to the rates indicated in Schedule A in the following way:

- (a) To Step One (1) upon hiring;
- (b) To Step Two (2) upon completion of the five hundred (500) hours of work in the classification and recommendation of the appropriate supervisor;
- (c) To Step Three (3) upon completion of an additional fifteen hundred (1,500) hours of work in the classification and the recommendation of the appropriate supervisor
- (d) Water & Wastewater Operators step progression is as stated in the Water and Wastewater Job Summary attached to the Collective Agreement. Water and Wastewater Operators step progression shall be:
  - (i) To Step One (1) upon hiring;
  - (ii) To Step Two (2) upon completion of five hundred (500) hours of work in the classification, additional qualifications (in accordance with the job description), and recommendation of the appropriate supervisor;
  - (iii) To Step Three (3) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
  - (iv) To Step Four (4) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
  - (v) To Step Five (5) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the

recommendation of the appropriate supervisor;

(vi) To Step Six (6) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;

# (e) Construction Operator step progression shall be:

- (i) To Step One (1) upon hiring;
- (ii) To Step Two (2) upon completion of two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (iii) To Step Three (3) upon completion of an additional two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (iv) To Step Four (4) upon completion of an additional two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the description), and the recommendation of the appropriate supervisor.

## (f) Water and Wastewater Leadhand step progression:

Water & Wastewater Operator, Steps 1 to 4	Water & Wastewater Leadhand, Step 1
Water & Wastewater Operator, Step 5	Water & Wastewater Leadhand, Step 2
Water & Wastewater Operator, Step 6	Water & Wastewater Leadhand, Step 3

#### (g) Water and Wastewater Infrastructure Repair Technician step progression:

Water & Wastewater Operator, Steps 1 to 4	W&WW Infrastructure Repair Technician, Step 1
Water & Wastewater Operator, Step 5	W&WW Infrastructure Repair Technician, Step 2
Water & Wastewater Operator, Step 6	W&WW Infrastructure Repair Technician, Step 3

- (h) An appropriate form will be completed and placed In the Employee's personnel file when the wage rate is amended.
- 30.03 Should an Employee not be recommended by the appropriate supervisor for advancement as per Clause 30.02 b), c), d) or e), the Administration shall advise that Employee of the reasons in writing, and in addition:
  - (a) What action might be taken in order for the wage to be advanced.
  - (b) In the event that the wage will not be advanced, what alternate action the administration intended to take.
- 30.04 An Employee affected by Article 30.03 may file a grievance for a failure to be recommended for an increase.
- 30.05 The rate of pay established in Schedule A shall apply during the term of this Agreement. Employees shall be paid bi-weekly. On each pay day, each Employee shall be provided with an itemized statement of his wages, overtime and other earnings and deductions.

# 30.06 Flag Person

The Employer may hire Temporary Employees as Flagperson at a rate in accordance with Schedule A. Should such employees be assigned to any other work they shall be paid in accordance with Clause 20.01. No one shall be regularly assigned duties as Flagperson until they have successfully completed an approved Flagperson's training course. Employees taking the Flagperson training shall be paid the appropriate rate of pay while training.

#### ARTICLE X - DIRECT RESPONSILBE PAY (DRP)

- X.01 When a Water and Wastewater Operator or Water and Wastewater Leadhand is assigned to the Utilities On-Call Schedule as On-Call #1, On-Call #2 or Transmission Operator (for emergent calls) and deemed to be in direct responsible charge (DRC), the Employee shall be paid an additional, five per cent (5%) per hour on the base rate of pay, for all hours worked between sixteen hundred and thirty (1630) hours and zero eight hundred (0800) hours Monday to Friday and for all hours worked on Saturday and Sunday.
  - (a) At no time shall Operators called out to assist the On-Call #1, On-Call #2 or Transmission Operator with emergency service or other similar callouts receive DRP.
- X.02 Where applicable, an Employee shall be eligible to receive DRP in addition to overtime, shift and weekend premiums.
- X.03 At no time shall DRP be included with the Employee's regular rate of pay for

the purpose of calculating overtime payments, other premium payments, or any Employee benefits.

## COLA:

January 1, 2020 – 1% LUMP SUM

January 1, 2021 - 1% LUMP SUM

January 1, 2022 – 1%

February 1, 2022 New Compensation structure and pay grid

January 1, 2023 – 1.5%

The following is included for clarity, but will not form part of the collective agreement:

- January 1, 2022 COLA increase will be retroactive to January 1, 2022 and payable within 3 months of ratification;
- Lump sums will be payable within 3 months of ratification;
- The new compensation structure and pay grid will be effective February 1, 2022;
- Temporary Employees will be eligible for benefits March 1, 2022;
- All other agreed to amendments will be effective February 1, 2022.
- Premiums referenced in Article 17 will be included in pensionable earnings.

The following articles have no amendments and will remain as current, except for any consequential amendments that may arise as a result of other agreed to amendments:

Article 5 - Management Rights

Article 7 - Time off for Union Business

Article 13 - Discipline

Article 18 - Call out & Standby

# **SCHEDULE A**

## February 1, 20<del>20</del>22

Job Code	Position	Hourly Rate		е
3001	Flag Person	<del>16.81</del>	17.42	
3002	<del>Labourer I</del>	20.78	21.14	

3003	<del>Labourer II</del>	23.41	<del>24.18</del>	
3004	Labourer III	25.77	<del>26.45</del>	<del>26.90</del>
TBD	PW Labourer	21.14	24.18	26.90
3005	Pickup/Courier Driver	24.85	25.40	
	,			[
3006	Survey Aide I	24.52	25.23	
3007	Survey Aide II	26.43	27.05	27.74
3008	PW Tech I	31.71	32.46	33.24
3009	PW Tech II	36.60	37.08	37.46
				-
3032	Gravel Checker	29.10	29.76	30.23
3010	Concrete Worker I	30.54	31.20	31.92
3011	Concrete Worker II	32.05	32.95	33.85
				4
3012	Truck Driver I	<del>27.95</del>	<del>28.55</del>	<del>29.14</del>
<del>3013</del>	Truck Driver II	<del>29.60</del>	30.33	31.03
3014	Truck Driver III	32.32	32.97	<del>33.76</del>
TBD	Truck Driver - Class 1	32.32	32.97	33.76
<del>3015</del>	<del>Operator I</del>	<del>27.95</del>	<del>28.67</del>	<del>29.08</del>
3016	<del>Operator II</del>	31.47	32.20	32.84
3017	Operator III	33.52	34.37	35.25
TBD	Equipment Operator I	27.95	29.53	30.01
TBD	Equipment Operator II	31.47	32.20	32.84
TBD	Equipment Operator III	33.52	34.37	35.25

3031	Sign Person	33.26	34.12	35.00
3018	Yardperson	33.26	34.12	35.00
TBD	North Yardperson	36.76	37.64	
3036	PW Leadhand	36.76	37.64	
3035	PW Project <del>Foreman</del> <b>Foreperson</b>	38.32	39.36	
3019	Operations Foreman Foreperson	41.48	42.36	43.30
3034	W&WW Labourer	20.78	24.18	26.90
3037	Wastewater Infrastructure Operator	30.27	32.84	35.25
3020	W&WW Operator (Collection & Distribution)	30.27	32.77	34.26
		36.72	38.45	40.91
3049	W&WW Operator (Transmission)	30.27	32.77	34.26
		36.72	38.45	40.91
3038	W&WW Leadhand	38.27	40.00	42.46
3039	W&WW Project Foreman	39.66	41.36	43.87
TBD	W&WW Infrastructure Repair Technician	39.66	41.36	43.87
3021	W&WW <del>Foreman</del> Foreperson	42.42	44.08	45.61
3022	W&WW Electrician	44.11	45.78	47.28
3033	Utilities Facilities Maintenance Technician	52.47	54.23	56.09
	1			
3023	Vehicle <del>Serviceman I</del> <b>Serviceperson I</b>	33.06	33.52	33.97
3024	Vehicle <del>Serviceman II</del> Serviceperson II	36.18	36.92	37.61
			_	
TBD	Shipper-Receiver, Materials Management	20.78	24.18	26.90
3025	Partsman   Parts Person	31.78	32.61	33.25
3026	Partsman II Parts Person II	34.14	35.08	35.95
3027	Welder	42.04	43.58	45.22

3028	Small Engine Mechanic	30.75	31.59	32.44
3040	V&E Leadhand	47.02	48.42	
3030	V&E Chargehand	45.91	46.34	48.42
3042	Transit Leadhand	47.02	48.42	
3062	Fleet Services Technician	42.04	43.58	45.22
			<u> </u>	
3043	Utility Operator I	27.95	<del>28.67</del>	29.08
3044	Utility Operator II	31.47	32.20	32.84
3045	Utility Operator III	33.62	35.72	<del>37.81</del>
TBD	Construction Operator	29.14	32.20	35.72
		37.81		
3046	Utility Leadhand	37.13	39.25	41.42
3047	7 Infrastructure Foreman Foreperson		43.31	44.90
3048	Utility Labourer	20.78	24.18	26.90
3050	Traffic Support Aid			32.84
	Utilities Instrument Technician/Electrician			
3051	Dual	46.30	48.05	49.66
3052	Utilities Electrician	44.11	45.78	47.28
3053	Utilities Instrument Technician	44.11	45.78	47.28
3063	Control System Technician	44.11	45.78	47.28
			1	
3101	Apprentice Auto Mechanic, 1st year (55%)	24.87		
3102	Apprentice Auto Mechanic, 2nd year (70%)	31.66		
3103	Apprentice Auto Mechanic, 3rd year (80%)	36.17		
3104	Apprentice Auto Mechanic, 4th year (90%)	40.69		

3111	Apprentice Heavy Duty Mechanic, 1st year (60%)	27.13		
3112	Apprentice Heavy Duty Mechanic, 2nd year (70%)			
3113	Apprentice Heavy Duty Mechanic, 3rd year (80%)	36.17		
3114	Apprentice Heavy Duty Mechanic, 4th year (90%)	40.69		
3029	Certified Technician (HD)	42.04	43.58	45.22
3131	Apprentice Transit Mechanic, 1st year (60%)	27.13		
3132	Apprentice Transit Mechanic, 2nd year (70%)	31.66		
3133	133 Apprentice Transit Mechanic, 3rd year (80%)			
3134	Apprentice Transit Mechanic, 4th year (90%)	40.69		

Hourly rates for apprentice technicians are a percentage of step 3 of Fleet Services
Technician.

# **SCHEDULE B**

EQUIPMENT	MINIMUM OF CLASSIFICATION RATE PAID
Non-Finishing Packers (Rubber Roller, Pad Foot Operator, and combination for non- finishing work)	Equipment Operator I – Step 1
Finishing Packer (Smooth Drum Rollers and combination of finishing work)	Equipment Operator I
Skid Steer / Sidewalk Machine	Operator II Equipment Operator I
Single Axle Truck	Truck Driver I Equipment Operator I
Sign Truck (Single Axle with Manlift)	Truck Driver II

	Equipment Operator I
<del>Tracked Dozer</del>	Equipment Operator II
Wheel Loader	Equipment Operator II
Tandem Axle Truck	Truck Driver II Equipment Operator II
Street Sweepers	Equipment Operator II
Truck Plow	Equipment Operator II
Sander	Truck Driver II Equipment Operator II
RPM Snow Blower	Equipment Operator II
Road Reclaimer	Equipment Operator II
PB Patcher Patch Truck	Equipment Operator II
Rubber Tire Backhoe	Equipment Operator II
Graders	Operator II Equipment Operator III
Track Hoe	Equipment Operator III + \$0.20 / hour
Paver	Equipment Operator III
Tractor-Trailer	Truck Driver <b>III Class 1</b>
<del>Directional Driller</del>	Utility Operator III

# Letter of Understanding

Between

Strathcona County

and

The Alberta Union of Provincial Employees, Local 118, Chapter 009

Subject: New Grid Placement and Recognition of Operator II hours

1. Employees with hours worked gained at the Operator II classification prior to ratification, including such hours worked gained on the Skid Steer/Sidewalk Machine, shall continue to have such hours recognized for the purposes of:

- a. Determining step levels on the new wage grid for Operator I and II classifications; and
- **b.** For those employed at date of ratification, the skid steer/sidewalk machines will qualify as one of the pieces of equipment when applying for an Equipment Operator II position, for the term of this agreement.

This letter of understanding expires on December 31, 2023.

#### and

#### Alberta Union of Provincial Employees (AUPE Local 118 Chapter 9)

Employer Proposal February 18, 2021

#### **ARTICLE 15 - LAYOFFS AND RECALLS**

- 15.01 Prior to any layoff affecting Regular Employees of the County, the parties will meet and endeavour to reduce the affect effect of such layoffs on Union members. Consultation shall occur prior to the issuance of any layoff notice to Union members. The Union shall be represented at such meetings by the following Chapter Officers; Chair, Secretary, Chief Union Steward and the AUPE Staff Representative or such other persons as designated by the Union, but in any event there shall be no more than four (4) representatives from the Union. Union members attending such meetings shall not suffer any loss of pay.
- 15.02 If a layoff is to be effected within a Branch, Temporary Employees will be laid off before any Regular Employees. If the staff of any Branch is to be reduced, the County shall first determine the number of positions to be reduced within each classification within that Branch. Those Employees with the least amount of seniority within a Branch shall be the first Employees removed from those classifications that are to be reduced. Those Employees who are to be removed from a position may displace any Employee in a lower classification within that Branch that has less Branch seniority providing that the Employee notified of layoff is qualified to perform the duties of the lower position.
- 15.03 The County will notify, or pay in lieu of such notice, the Employees who are to be laid-off according to the Employment Standards Code. The Union shall be advised at the same time as the Employee is notified.
- 15.04 Regular Employee(s) shall be recalled in the order of their seniority (the most senior Employee laid off shall be the first to be recalled) providing they have the necessary qualifications and ability to do the work. The opportunity for reemployment shall exist for eighteen (18) months.
- 15.05 Recall notice shall be by either text or email to the Employee's last contact information on record with the County. If the Employee fails to respond within

forty-eight (48) hours, the Employer shall send the recall notice by registered mail to the Employee's last address on record with the County. It is the responsibility of each Employee to notify the County promptly in writing of any change of address or contact information. When recalled, an Employee who fails to report to work within fifteen (15) days of the date of recall notice shall forfeit his claim to re-employment.

15.06

Where Employees have been laid off and recalled, they shall retain the seniority held at the date of layoff. The Union shall be notified of all recalls. NOT TO BE DELETED BUT RATHER MOVED TO SENIORITY ARTICLE (ARTICLE NUMBER TO BE DETERMINED).



Sharla Mishna Strathcona County Merryn Edwards

#### and

# Alberta Union of Provincial Employees (AUPE Local 118 Chapter 9)

Employer Proposal February 18, 2021

#### **ARTICLE 19 - EMPLOYEE ALLOWANCES**

#### 19.01 Safety Equipment Allowance

- (a) Each Regular Employee who has completed the initial period of probation as required in Clause 11.01 of this Agreement shall be reimbursed by the County to a maximum of two hundred dollars (\$200) per calendar year as an allowance for the purchase of safety equipment or upgraded safety equipment not supplied by the County which must include, but is not limited to, C.S.A. approved safety footwear. A valid receipt(s) as evidence of the purchase of C.S.A. approved safety footwear and safety equipment must be presented. The Employee may carry over into the next calendar year any unused amount of his current years boot allowance. The maximum carry over in any one year cannot exceed two hundred dollars (\$200). The Department Head, if he feels it is necessary, may approve for reimbursement a second pair of boots in a calendar year up to the value of the annual boot / safety equipment allowance.
- (b) Each Temporary Employee, after the completion of the probationary period, shall be reimbursed by the County to a maximum of one hundred and fifty dollars (\$150) per calendar year as an allowance for the purchase of safety equipment or upgraded safety equipment not supplied by the County which must include, but is not limited to, C.S.A. approved safety footwear. A valid receipt(s) as evidence of the purchase of C.S.A. approved safety footwear and safety equipment must be presented. The Temporary Employee may carry over into the next calendar year any unused amount of his current years boot allowance. The maximum carry over in any one year cannot exceed one hundred and fifty dollars (\$150).
- (c) Upgraded safety equipment may include but is not limited to:
  - Molded ear plugs
  - Prescription safety lenses/glasses
  - Upgraded gloves

Welding (individual) shields

Approval will not be unreasonably denied.

## 19.02 <u>Meal Allowance</u>

If an Employee is authorized to work overtime of more than two (2) consecutive hours immediately following his scheduled daily hours on a normal working day and purchases a meal for consumption during a break in the overtime period or immediately following completion of the overtime period, a dinner meal allowance may be claimed in accordance with the County Employee Business Expense Policy (HUM-001-026) County policies.

19.03 If an Employee is authorized to work more than four (4) consecutive hours on a call out situation a meal allowance may be claimed in accordance with the County Employee Business Expense Policy (HUM-001-026) County policies.

#### 19.04 Tool Allowance

The Employer shall pay a six hundred dollar (\$600) per year tool allowance to Certified Technicians and Vehicle Serviceman. The Employee may carry over into the next calendar year any unused amount of the current year's tool allowance. The maximum carry over in any one year cannot exceed six hundred dollars (\$600). Reimbursement shall be made once proof of purchase is provided and approved.

#### 19.05 Mileage

When an Employee reports to his first designated work place of the day and then is directed to use his personal vehicle to move from one job site to another during his work shift he shall be reimbursed for mileage as follows: between his first designated workplace to his last designated workplace of the day. Mileage reimbursement shall be claimed in accordance with the County Employee Business Expense Policy (HUM-001-026) County policies.

Slisha

Sharla Mishna Strathcona County Merryn Edwards

#### and

### Alberta Union of Provincial Employees (AUPE Local 118 Chapter 9)

February 19, 2021

The parties agree on clauses 24.01, 24.02, 24.03, 24.04, 24.05, 24.06, 24.08, and 24.09. The Union proposal for clause 24.07 is still outstanding.

#### **ARTICLE 24 - VACATION LEAVE**

- 24.01 Annual vacation entitlement with pay shall be as follows:
  - (a) An Employee who has completed less than six (6) years of service shall earn vacation leave at the rate of four point sixty-two (4.62) hours per bi-weekly period of continuous service [fifteen (15) days per year] or five point two (5.20) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
  - (b) An Employee who has completed six (6) years, but less than fourteen (14) years of service shall earn vacation leave at the rate of six point fifteen (6.15) hours per bi-weekly period of continuous service [twenty (20) days per year] or six point ninety-two (6.92) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
  - (c) An Employee who has completed fourteen (14) years of service, but less than twenty-two (22) years shall earn vacation leave at the rate of seven point seven (7.70) hours per bi-weekly period of continuous service [twenty-five (25) days per year] or eight point sixty-seven (8.67) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
  - (d) An Employee who has completed twenty-two (22) or more years of service shall earn vacation leave at the rate of nine point twenty-three (9.23) hours per bi-weekly period of continuous service [thirty (30) days per year] or ten point thirty-nine (10.39) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week.
- 24.02 If Employees become ill on an annual vacation leave for more than three (3) consecutive days paid vacation days, the applicable days will be deemed general

illness, and they will have the applicable number of vacation days reinstated upon application and provision of a medical certificate.

24.03 The number of days of vacation leave which may be accumulated by an Employee is limited to the number he can earn in eighteen (18) months of continuous service.

The County will automatically pay out any excess vacation leave over the maximum accumulation at the beginning of each December.

When vacation pay is paid out rather than being taken as vacation, the payout shall be based on the Employee's rate of pay at the time of payout.

- 24.04 Vacation leave may be taken in the calendar year it is earned. However, vacation leave may not be taken until it has been earned. The vacation leave may be granted as follows:
  - (a) In one continuous period;
  - (b) As single days off or some other combination;
  - (c) If an Employee earns vacation at a rate of four point sixty-two (4.62) hours per bi-weekly period of continuous service, one (1) week vacation leave may be taken as single days off or some other combination provided the total does not exceed five work days in any calendar year;
  - (d) If an Employee earns vacation at a rate of six point fifteen (6.15) hours per bi-weekly period of continuous service or seven point seven (7.70) hours per bi-weekly period of continuous service, two (2) weeks' vacation leave may be taken as single days off or some other combination provided the total does not exceed ten (10) work days in any calendar year;
  - (e) If an Employee earns vacation at a rate of nine point twenty-three (9.23) hours per bi-weekly period of continuous service, three (3) weeks' vacation leave may be taken as single days off or some other combination provided the total does not exceed fifteen (15) work days in any calendar year.
- In addition to the automatic payout of excess vacation referred to in 24.03, an Employee may be permitted to receive a cash payment in lieu of vacation leave, provided any and all obligations have been met in accordance with Employment Standards legislation and the Employee has received written approval from the Department Head.

Temporary Employees shall have vacation pay added to their bi-weekly pay cheques and shall be entitled to unpaid vacation leave in accordance with the

### **Employment Standards Code.**

- 24.06 An Employee who does not complete one (1) full year of service shall be entitled to vacation pay in accordance with the Employment Standards Code and regulations thereunder.
  - (a) Temporary Employees will receive vacation pay at four percent (4%) of their gross regular earnings during their first four thousand (4000) regular standard hours.
  - (b) Temporary Employees will receive vacation pay at six percent (6%) of their gross regular earnings after completion of four thousand (4000) regular standard hours.
  - (c) If a Temporary Employee has a break in service greater than eighteen (18) months then vacation pay shall be paid at four percent (4%) of their gross regular earnings as per Clause 24.07 6a) above.

# 24.07 **[OUTSTANDING]**

24.08 Vacation hours will be paid out at the Employee's base rate in place at the time the vacation is taken.

# **Summer Vacation Schedules**

#### 24.09 Summer Vacation Schedules

Employees must submit summer vacation applications to the Director or his designate by March 15th of each year. The Employer will post schedules, as approved by the Director, by May 1st of each year.

Sharla Mishna
Strathcona County

Merryn Edwards
AUPE

#### and

### Alberta Union of Provincial Employees (AUPE Local 118 Chapter 9)

Employer Proposal February 18, 2021

#### **ARTICLE 26 - LIAISON COMMITTEE**

- 26.01 The parties agree that there shall be a Liaison Committee established for the purpose of discussing and dealing with matters of mutual concern. The Committee shall consist of not less than two (2) representatives of both the County and the Union. 26.02 On the part of the County, one of the members will be the Department Head (Human Resources) or his designate; and, on the part of the Union, one of the members will be the assigned Membership Services Officer or his designate. 26.03 Where the Liaison Committee determines, or accepts a recommendation from the Safety Committee, that coveralls or other such items should be provided for the protection of the Employee's personal garments, such items should be provided, and replaced, upon approval of the designated official. 26.04 The Liaison Committee shall meet not less than once each month, but such meetings may be waived by mutual consent. 26.05 The rules of procedure for the Liaison Committee meetings shall be developed and agreed upon by the Committee itself and the recorded minutes of each meeting will be signed by a representative of each group. 26.06 The Chairmanship of the Committee will alternate in presiding over the meetings between the officials referred to in Clause 26.02.
- 26.07 The Liaison Committee shall not have jurisdiction over wages or any matter of collective bargaining including the amending or administering of this Agreement. The Committee shall also not supersede the lawful activities of any other committee of the County or the Union and shall not have the power or authority to bind either Party to any decisions reached in their discussions. The Committee shall have only the power to make recommendations to the County or to the Union with respect to its decisions.

26.08 Notwithstanding Clause 26.07, the Liaison Committee may review job descriptions and/or classifications pursuant to Clauses 8.01 and 8.02.

Slisha

Sharla Mishna Strathcona County Merryn Edwards

#### and

# Alberta Union of Provincial Employees (AUPE Local 118 Chapter 9)

Employer Proposal February 18, 2021

#### ARTICLE X - NOTICE OF RESIGNATION

X.XX Employees shall provide at least ten (10) working days written notice of their

intention to resign.

X.XX In the event that an Employee terminates his employment with the County

without submitting the written notice in compliance with Clause 11.04X.XX, the Employee shall not be entitled to receive his regular cheque and/or his final cheque until the next regular distribution of regular wage and salary cheques.

Slisha

Sharla Mishna Strathcona County Merryn Edwards

#### and

# Alberta Union of Provincial Employees (AUPE Local 118 Chapter 9)

Employer Proposal February 19, 2021

**Delete LOU – Level 1 Grievance Procedure** 

**Delete LOU – Casual Vehicle Transporter Classification** 

Delete LOU – Administrative Changes to the Collective Agreement

Delete LOU - Vacation

Mesha

Sharla Mishna Strathcona County Merryn Edwards

#### and

#### Alberta Union of Provincial Employees (AUPE Local 118 Chapter 9)

Partial sign off March 15, 2021

The parties agree on clauses 3.01 (with the exception of the effective date, as noted below), 3.02, 3.03, 3.04, and 3.05. The term of the agreement in 3.01 is outstanding. Both parties have proposed a three-year term ending December 31, 2022. However final agreement on term shall be subject to agreement on other terms and conditions in this agreement.

#### **ARTICLE 3 – TERMS OF AGREEMENT**

- 3.01 This Agreement shall be binding and remain in effect from the date upon which both parties exchange notice of ratification by their principals from January 1, 2017 until December 31, 2019 [EFFECTIVE DATE IS OUTSTANDING] and shall continue from year to year thereafter unless either party gives the other notice in writing in accordance with the Labour Relations Code in effect at the time.
- 3.02 Where notice to negotiate amendments for a new Agreement has been given, this Agreement shall continue in force and effect during the period that the parties bargain for a new Collective Agreement or in accordance with the Labour Relations Code in effect at the time.

Either party may give to the other not less than sixty (60) nor more than one hundred and twenty (120) calendar days prior to the termination of this agreement notice in writing of its intention to commence collective bargaining. Collective bargaining shall be conducted in accordance with the provisions of the *Labour Relations Code*.

Where notice is served by either party under the Labour Relations Code, provisions of this Collective Agreement shall continue until either:

- (a) a settlement is agreed upon and a new Collective Agreement signed; or
- (b) a settlement is not agreed upon and then this Collective Agreement shall remain in effect until a strike or lockout commences as provided in the Labour Relations Code.

Any notices shall be served in accordance with the Labour Relations Code.

- 3.03 Notwithstanding anything in this Article, any portion of this Agreement may be opened for negotiations between the County and the Union at any time provided that both parties agree.
- 3.04 Notwithstanding the foregoing, the parties may mutually agree to amend the terms and/or condition of employment.

In such cases, the agreement must be in writing and executed by authorized representatives of the parties to the Agreement.

- 3.05 Within one (1) month of the ratification of this Agreement, the Department Head (Human Resources) shall convene a meeting of all supervisory personnel to review the contract in an attempt to achieve consistency and fairness. The Union will be invited to be represented at this meeting.
- 3.06 The Employer and the Union shall equally cost share the printing of this Agreement. Printing of this agreement shall be the responsibility of the Union.

Slisha

Sharla Mishna Strathcona County Merryn Edwards
AUPE

#### and

# Alberta Union of Provincial Employees (AUPE Local 118 Chapter 9)

Sign off March 15, 2021

#### ARTICLE 27 - OCCUPATIONAL HEALTH AND SAFETY

- 27.01 All Employees must comply with the approved departmental and OH&S dress code as a condition of employment.
- 27.02 Each Employee and each supervisor shall take reasonable care for the protection of public and Employee health and safety in the operation of equipment and the storage or handling of materials and substances.
- 27.03 The Employer and the Union recognize the importance of a safe and healthy workplace. The Employer and the Union will cooperate in the promotion of occupational health and safety and the prevention of incidents. The Employer and Employees will take reasonable steps to eliminate, reduce or minimize workplace hazards.

The Employee shall immediately notify his/her supervisor in the event that an incident either results in a "near miss" or a workplace accident resulting in a loss or injury.

- 27.04 Where the Employer requires an Employee to undergo compulsory medical examination(s), the cost of such examination(s) shall be paid for by the Employer.
- 27.05 Personal protective clothing and safety equipment shall be provided by the Employer, and worn by the Employee, as required by the Occupational Health and Safety Act and Regulations thereto at no cost to the Employee. All protective clothing and safety equipment supplied by the Employer shall be maintained by the Employer and remain the property of the Employer.
- 27.06 The Employer shall provide on an as needed basis to Employees the following: coveralls, winter gear (which includes coveralls or winter jacket/pants), rain coat and rain pants. Temporary Employees shall return all clothing to the Employer at the termination of their temporary employment. The clothing as outlined in this clause will be provided, repaired or replaced at the discretion of the Employer.

### 27.07 Joint OHS Committee

The Joint Occupational Health and Safety Committee shall be composed of representatives of the Employer and Employees, one of whom is appointed by the Union as a representative of the local Chapter. The Committee shall meet regularly to discuss, communicate and make recommendations on the implementation of the health and safety program.

An Employee who attends a Joint OHS Committee meeting that is scheduled outside of their regular shift, shall be paid at their base hourly rate of pay for a minimum of two (2) hours, or the length of the meeting whichever is greater. There shall be no minimum guaranteed compensation if the meeting is contiguous with a normal working period.

Slisha

Sharla Mishna Strathcona County Merryn Edwards AUPE

and

### Alberta Union of Provincial Employees (AUPE Local 118 Chapter 9)

Employer Proposal April 12, 2021

#### **ARTICLE 30 - OVERPAYMENT**

30.xx In the event of an overpayment, the County shall notify the employee that an overpayment has been made within six (6) months of the date of the last overpayment. The County is authorized to recover any overpayment from an Employee and will discuss options with the Employee regarding how that overpayment is repaid.

30.xx In the event that the parties cannot agree on a repayment plan, the Employer shall recover the overpayment by deducting up to ten percent (10%) of the Employee's gross earnings, per pay period, until the overpayment has been fully repaid.

30.xx If the Employee is terminated for just cause or resigns before full repayment is made, the remainder of the repayment shall be recovered from the Employee's final pay.

Slisha

Sharla Mishna

**Strathcona County** 

Merryn Edwards

#### and

### Alberta Union of Provincial Employees (AUPE Local 118 Chapter 9)

Employer Proposal April 12, 2021

### Letter of Understanding

#### Between

### Strathcona County

and

The Alberta Union of Provincial Employees, Local 118, Chapter 009

### Subject: Registered Apprenticeship Program (RAP)

The Registered Apprenticeship Program is an apprenticeship program for high school students, providing students with the opportunity to start learning and practicing their future trade while in high school. The Employer and the Union jointly support the RAP program with the understanding that no current Employee will be laid off as a direct result of participation in the program. The parties agree that employees employed under the RAP program are included in the bargaining unit.

- 1. Employment under the RAP program will be on a casual basis for the duration of the student's participation in the program. Hours of work will be as agreed to between the Employer, student, and educational institution as outlined in the RAP program contract.
- Hours worked while participating in the RAP program are not considered continuous employment toward permanent status.
- 3. The rate of pay while participating in the program, and all other specific terms and conditions of employment, will be determined by the employer and the educational institution. The following terms will continue to apply as outlined in the Collective Agreement:
  - Union Recognition Article 6 Union Membership Clauses 5.08 and 5.09
  - Occupational Health and Safety Article 13
  - Grievances Article 7
    - o Application of the grievance procedure shall apply, except for termination of employment as a result of unsatisfactory probationary employment, or failure to

comply with the terms and conditions of the RAP program contract.

4.	RAP students will be I	aid off upon completion of participation in the RAP program.
Guy Smitl	n, President	Debra Smith, Human Resources Director Strathcona County
	Date	Date
5	lisha	Lon
Sharla Mishi Strathcona (		Merryn Edwards AUPE

Sharla Mishna	Merryn Edwards
Strathcona County	AUPE

From: Merryn Edwards < m.edwards@aupe.org>
Sent: Wednesday, May 18, 2022 9:23 AM

To: Chantel Kassongo

Cc: Sharla Mishna; Sharry Sowiak; Sheron M. Ward

**Subject:** Re: Strathcona County & AUPE

Attachments: AUPE Strathcona County settlement proposal .docx

Hi Chantel,

Thanks for your patience while we have consulted with our membership.

The main concerns we heard were:

- The sense that the inclusion of seasonals in benefits at the same time as reducing pay due to the downgrading of the sidewalk equipment (and reduction in callout pay for utilities) was an unacceptable "give with one hand/take with the other" approach
- The insistence that wage increases be on the grid, not as lump sums
- Ongoing frustrations with scheduling for TAS staff over Christmas

The results of that discussion have culminated in the attached proposal, which includes the following changes from the previous TA:

- 1% on the grid instead of lump sums for 2020 and 2021 (maintained 1% for 2022 and 1.5% for 2023)
- Sidewalk machine/skid steer maintained at Equipment Operator II
  - LOU re Recognition of Operator II hours on sidewalk/skid steer not necessary
- Rates for Equipment Operator I and new merged Labourer classification reduced from TA
  - Labourer steps correspond to existing LI step 1, LII step 2, and LIII step 3 (vs. LI step 2, LII step 2, LIII step 3)
  - Equipment Operator I steps same as current Operator I steps (vs. increased steps 2 and 3)
- Flat rate (\$2.25) vs per cent (5%) premium for Direct Responsible Pay (current Working Alone) and addition of application to Holidays
- Removed deletion of "five (5) consecutive days" from Christmas Day/Christmas floater/Boxing Day and weekend (23.07)

We are available June 27 and/or 28 to reconvene in bargaining.

Thank you, Merryn

Merryn Edwards

Negotiations
Alberta Union of Provincial Employees
phone: 780-930-5287 (temporarily unavailable)

cell: 780-952-1951 fax: 780-930-3392

toll-free: 1-800-232-7284 email: m.edwards@aupe.org

El respeto al derecho ajeno es la paz. Respect for the rights of others is peace. --Benito Juarez

From: Chantel Kassongo < ckassongo@ntlaw.ca>

Sent: May 16, 2022 12:22 PM

To: Merryn Edwards < m.edwards@aupe.org>

Cc: Sharla Mishna <Sharla.Mishna@strathcona.ca>; Sharry Sowiak <Sharry.Sowiak@strathcona.ca>; Sheron M. Ward

<sward@ntlaw.ca>

Subject: Strathcona County & AUPE

CAUTION: This email has been sent from an external source. Treat links and attachments in this email with care.

Hi Merryn:

Four months have passed since you advised that the TA did not ratify.

The County would like to schedule dates to bargain and/or at the very least discuss next steps.

Please let us know of your availability over the next couple of months.

Thanks.

Chantel



# Chantel Kassongo Barrister & Solicitor

#301, 550 – 91 Street SW, Edmonton, AB T6X 0V1
D: 780-733-6563 | C: 780-938-8040 | F: 780-488-

Email | Website | LinkedIn | Employment Law Alliance



This email and any attachments may contain privileged and confidential information. Review, use or retention of this information by someone other than the recipient is prohibited. If you have received this message in error, please contact the sender and delete all references to the email from your system.

As the CMO has implemented mandatory restrictions, we ask that all members and staff comply with those restrictions. It is expected that members continue to hold all meetings virtually. Meeting space at AUPE offices will not be available as the offices will remain closed until further notice.

Disclaimer

The information contained in this communication from the sender is confidential. It is intended solely for use by the recipient and others authorized to receive it. If you are not the recipient, you are hereby notified that any disclosure, copying, distribution or taking action in relation of the contents of this information is strictly prohibited and may be unlawful.

THIS AGREEMENT is made this 18th day of April, 2018, A.D. pursuant to the Labour Relations Code of Alberta (2000) as amended.

This Agreement is entered into by and between the Strathcona County, municipal corporation covered by the Collective Agreement (hereinafter referred to as "the County" or "the Employer") and the Alberta Union of Provincial Employees Local 118/009 (hereinafter referred to as "the Union").

Whereas, the above parties are mutually desirous of entering into a Collective Agreement with the intent and purpose of:

- A. Promoting a harmonious relationship;
- B. Providing for conditions of employment, hours of work and rates of pay;
- C. Establishing the means for prompt and equitable settlement of disputes or grievances between the County and Employees represented by the Union, and;
- D. Encouraging efficiency in operation.

Now therefore, the parties hereto mutually agree as follows:

#### **ARTICLE 1 – DEFINITIONS**

- 1.01 In this Agreement:
  - (a) A word used in the singular applies also in the plural. A word used in the masculine applies also in the feminine.
  - (b) Employee: means a person employed by the Strathcona County in the Public Works Operations, Vehicle and Equipment Services, Transit Services, Materials Management, Water and Wastewater Operations, and as listed in Schedule A.
    - (i) Regular Employees: are Employees who occupy established positions and who have successfully completed the probationary period and work the fully scheduled hours.
    - (ii) Part-time Employees: are Employees who occupy established positions and have successfully completed the probationary period and work less than the fully scheduled hours of Regular Employees.
    - (iii) Temporary Employees: are Employees who are indefinite term Employees or Employees for specific projects, seasonal employment, replacements for Regular Employees who are on illness leave, General Illness, Long Term Disability, Workers' Compensation, Maternity Leave, Parental Leave and Adoption Leave, or other long term leaves of absence. Indefinite term Employees upon completion of more than

fifteen (15) months of continuous service will be deemed to be Regular Employees, except if they are covering for an Employee on Long Term Disability, are moving from one Department to another Department for a temporary or seasonal position, or are covering off consecutive leaves, as referenced above. "Continuous service", for the purpose of this clause, means any period of employment that does not include a break in service of more than thirty (30) calendar days.

- (iv) Apprenticed Employees: are Employees who are registered as Apprentices under the Apprenticeship and Industry Training Act.
- (v) Probationary Employee: means an Employee who during his initial period of employment occupies an established position during a probationary period.
- (c) Grade: means the step assigned to a class within the pay grid.
- (d) Union Steward: means a person who is selected by the members of the Union to act on behalf of those Employees as empowered by the Union.
- (e) Union Representative: means a person authorized by the Union to act on behalf of Employees and includes a Union Steward, elected or appointed officer of the Union, or staff representative.
- (f) Liaison Committee: is the joint Liaison Committee established by the two parties as stipulated within this Agreement.
- (f) Casual Illness: is any illness which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less.
- (g) General Illness: is any illness which causes an Employee to be absent from work for a period of more than three (3) consecutive work days.
- (h) Layoff: includes a reduction in the work force as a result of shortage of work.
- (i) Branch: means
  - (i) Public Works Operations
  - (ii) Vehicle and Equipment Services
  - (iii) Transit Services
  - (iv) Water and Wastewater Services
  - (v) Materials Management
- (j) Classification: means each line in Schedule A.
- (k) Seniority: commences at date of hire, and shall include all continuous **and accumulated service** as a temporary Employee. <del>provided there was no</del>

- break in service greater than thirty (30) days.
- (l) Day: shall mean calendar day.
- (m) Continuous Service: any period of employment that does not include a break in service of more than thirty (30) calendar days.
- (n) Accumulated Service: Total number of regular hours worked, with no break in service of more than six (6) months.

#### **ARTICLE 2 – JURISDICTION**

- 2.01 During its term, this Agreement shall be applicable to any person employed in the Public Works Operations, Vehicle and Equipment Services, Transit Services, Materials Management and Water and Wastewater Services of Strathcona County, who are described in Schedule A attached hereto, or to positions that fall within the jurisdiction of the Union.
- When additional work is required by the Employer, individuals who have paid Union dues to the Alberta Union of Provincial Employees, Local 118, Chapter 009 in the past twelve (12) months, will be called first and, if available, offered the additional work.

#### **ARTICLE 4 - APPLICATION**

- 4.01 Regular Employees shall be granted all the terms and conditions of this Agreement.
- 4.02 Part-time Employees shall be granted all the terms and conditions of this Agreement except those conditions which require a particular length of service, such benefits shall be granted on a pro-rata basis in relation to hours of work of full-time employment.
- 4.03 Apprenticed Employees as defined within Apprenticeship and Industry Training Act shall be granted all the terms and conditions of this Agreement.
- 4.04 Temporary Employees shall be entitled to the terms and conditions of this Agreement, except where Regular Employees are referenced. Seniority for employment periods of less than six (6) months shall not be accumulated.

#### ARTICLE 6 - UNION RECOGNITION

- 6.01 The County recognizes the Union as the sole bargaining agent of the Employees covered by this Agreement.
- No Union Employee shall be required or permitted to make any written or verbal agreement which may be in conflict with the terms of this Collective Agreement.

#### 6.03 New Employees

The Employer shall supply all new Employees with a copy of the Collective

# 6.04 <u>Monthly Staffing Reports</u>

The County recognizes the need of the Union to have required information concerning Employees eligible for membership in the Union and hereby agrees to submit between the sixteenth (16th) day and the last day of each month, the name, address, and position filled by any new Employees assigned to Public Works Operations, Vehicle and Equipment Services, Materials Management, Water and Wastewater Operations and eligible for membership in the Union.

The Employer recognizes the importance of new Union Employees receiving an orientation to the Union. To ensure this occurs, the Chapter Executive will be given a list of all newly hired Employees and a representative of the Union shall, within thirty (30) calendar days of the Employees' date of hire, make a presentation to the Employee(s) with respect to the structure of the Union as well as the rights, responsibilities and benefits of membership in the Union.

The Employer will generally, provide the Union with one (1) weeks' advance notice of the Employee's scheduled orientation.

# 6.04 <u>No Discrimination</u>

The parties agree that there shall be no discrimination or coercion exercised or practiced with respect to any Employee for reason of function, membership or legitimate activity in the Union.

The Employer shall exercise its rights in a fair and reasonable manner, nor shall these rights be used in a manner which would deprive any Employee of his employment, except for just cause.

### 6.05 <u>Bulletin Boards</u>

The County recognizes the right and need of the Union to be able to post notices of meetings and other legitimate Union activities at various sites wherein members of the Union are assigned or employed.

### 6.06 <u>Seniority Lists</u>

County shall maintain a list showing the date upon which each Employee's service commenced. The County shall forthwith provide the Union, upon request, with an up-to-date list and shall, in any event, post an up-to-date list on all bulletin boards where the Employees work on July 1<sup>st</sup> and January 1<sup>st</sup> in each year.

### 6.07 Union Officers-Stewards

The County recognizes the right of the Union to appoint "Employees" as Union Stewards who are to be recognized as official representatives of the Union.

### 6.08 Union Membership

Each person who becomes employed in positions covered by this Agreement shall automatically become a member of the Union as a condition of employment.

# 6.09 <u>Monthly Staffing Reports</u>

- (a) The Employer agrees to remit to the central office of the Union, the amounts equal to the dues that have been deducted from the pay of all Employees by the first working day after the fifteenth (15th) calendar day in the following month. Where an accounting adjustment is necessary to correct an over or under payment of dues, it shall be effected in the succeeding month.
- (b) The County recognizes the need of the Union to have required information concerning Employees eligible for membership in the Union and hereby agrees to submit on a monthly basis between the sixteenth (16th) day and the last day of each month, the following information: the name, address, phone number, status (regular or temporary), Identification number, classification, starting date and position filled by any new Employees In the bargaining unit. assigned to Public Works Operations, Vehicle and Equipment Services, Materials Management, Water and Wastewater Operations and eligible for membership in the Union.

The Employer shall provide the Union with a monthly list identifying each Employee, including new or terminated Employees. The list will include Employee name, address, identification number, starting date, classification, and dues deducted for that calendar year.

(c) The Employer recognizes the importance of new Union Employees receiving an orientation to the Union. To ensure this occurs, the Chapter Executive will be given a list of all newly hired Employees and a representative of the Union shall, within thirty (30) calendar days of the Employees' date of hire, make a presentation to the Employee(s) with respect to the structure of the Union as well as the rights, responsibilities and benefits of membership in the Union.

### ARTICLE 8 - JOB DESCRIPTIONS CLASSIFICATIONS & RECLASSIFICATIONS

8.01 The Employer shall make the Employee's job description available.

#### 8.02 New or Amended Classifications

The County Administration shall give written notice to the Union of the establishment of any new or amended job descriptions and/or salary classification levels (Schedule A). If it is established that a higher salary level is appropriate, the new salary shall be retroactive to the date of the classification was created. Where Union and management representatives through the Liaison Committee are unable to agree within thirty (30) days on either the correct classification of a revised job

description or on the compensation level of a new job description, the matter may be advanced to arbitration for the establishment of the compensation for the individuals for which the job description has changed. If a wage change results, the effective date of change shall be determined by the Arbitrator. Both the Employer and the Union will share the cost of the Arbitrator equally. The Arbitrator will be guided by:

- (a) Evidence and submissions of the parties;
- (b) The salary levels in the Collective Agreement;
- (c) The complexity, working conditions and knowledge requirements of the other positions in the Collective Agreement at the time that the new position is established.

However, the Arbitrator shall not be restricted to the existing salary levels.

#### 8.02 Classification Reviews

An Employee may at any time bring to management for review, changes that may have occurred to his existing job duties. The Employee/Union, may bring such issues forward to Liaison Committee.

### 8.03 Reclassification

When it is necessary to re-classify Regular Employees, the County may re-classify the affected Employees on the basis of Branch seniority, qualifications and ability.

- (a) An Employee to be re-classified may displace any Employee in a lower classification within that Branch that has less Branch seniority providing that the Employee being reclassified is qualified to perform the duties of that position;
- (b) Employees with special qualifications may be exempt from re-classification.
- 8.04-12.02 In those cases where re-classification according to Clause 12.01 8.03 have occurred and vacancies become available in the affected job classification, those Employees who were previously re-classified downward from that classification, shall be returned to that classification on the basis of their seniority so long as that Employee has maintained the ability and qualifications to be appointed to the position.
- 8.05 12.03 An apprentice who begins and completes their apprenticeship with the Employer shall be placed on level 2 of the appropriate trades rate upon confirmation of certification.

# ARTICLE 9 - POSTING OF POSITIONS, PROMOTIONS AND TRANSFERS

9.01 When a position of a Regular or Temporary nature is to be filled, notice of the

vacancy shall be posted as soon as reasonably possible for a minimum of one (1) week. The position can be filled by appointment if it is to be filled for a period of less than ninety (90) calendar days.

- 9.02 If a position with the same job code becomes vacant within six (6) months of the original closing date, the Employer may consider candidates from the original recruitment process, without re-posting. This is not applicable if the original posting was for a temporary position and the vacancy is for a permanent position.
- 9.02 3 Such notice will contain the following information: date of the posting, nature of position, qualifications, required knowledge and education, skills, shift, hours of work, experience, wage rate or range, consistent with the job description.
- 9.03 4 Where there are qualified Union candidates for a posted position, their applications shall be considered first.
- 9.04-5 Where qualified applicants have applied and a suitable candidate has been selected, the Employer shall appoint a candidate to the position within thirty (30) days of the competition closure.
- 9.05 6 Preference in the filling of positions shall be given to Employees of the County over all other persons, where the Employee has the required skills and ability to perform the tasks of the position. When filling positions or making promotions, qualifications, experience, skills, ability and seniority shall be the governing factors. When filling positions, employees of the County shall be given first preference.

#### 9.06 7 Trial Period

An Employee who is transferred or promoted to another position within this Agreement shall be confirmed in that other position within three (3) months unless the Employee wishes otherwise or proves unsatisfactory in the new position, in which case the Employee will be returned to his former position. Any other Employee promoted or transferred because of the displacement of positions shall also be returned to his former position. In any event, Employees promoted or transferred shall be paid at the rate applicable to the new position from commencement of employment in the position until returned to the previous position.

- (b) In making promotions and filling vacancies, appointments will be made on the basis of qualifications, experience, ability and seniority.
- 9.08 The posting requirements in this Article may be waived by mutual agreement in cases of medical accommodation.

#### ARTICLE 10 - SENIORITY

10.01 For the purpose of Clauses 10.01 to 10.03 10.10, 12.01 to 12.03 12.02, and 15.01 to

15.06 15.05, seniority shall be defined from the date of continuous service employment in any Branch covered by this Agreement.

In accordance with 1.01(k) seniority commences at date of hire, and shall include all continuous and accumulated service as a temporary Employee.

- 10.02(a) Employees who were employed by the County prior to January 1, 1979 and entered in any classification covered by this Agreement on that date shall have their seniority established as the date of continuous service employment with the County.
- 10.023(b) Seniority shall be retained and accumulated during absences resulting from illness or an accident where the Employee is entitled to compensation, suspension with or without pay less than sixty-one (61) days, approved leave without pay less than sixty-one (61) days.
- 10.034 (c)——Seniority shall be retained during absences resulting from; layoff (up to eighteen (18) months), suspension with or without pay greater than sixty (60) days, approved leave without pay greater than sixty (60) days, strike or lockout.
- 10.045 Where Employees have been laid off and recalled, they shall retain the seniority held at the date of layoff. The Union shall be notified of all recalls.
- 10.056(d) ——An Employee who accepts an internal permanent position outside this Agreement shall retain the seniority accrued to that time for a period of six (6) months.
- 10.0210.06 Accumulation of seniority shall recommence from the date the Employee returns to a classification covered by this Agreement and their seniority date shall be adjusted by the period of their absence. This retained seniority shall not be effective in exercising any seniority rights until the Employee has been back in the scope of this Agreement for six (6) months.
- An Employee shall lose their seniority and their name shall be removed from the seniority list for any of the following reasons: voluntary resignation, discharge for cause, layoff for more than eighteen (18) months, retirement, or when an Employee has been employed in a position outside this Agreement for a period of six (6) consecutive months or more.

#### **ARTICLE 11 - PERIOD OF PROBATION**

11.01 The probationary period for new Employees covered by this agreement in both permanent and temporary positions shall be five hundred (500) hours one thousand (1000) hours worked. , with the exception of Employees appointed to the positions listed below, whose probationary period will be one thousand (1000) hours worked:

3008	Public Works Technician	3015	<del>Operator I</del>
3009	Public Works Technician II	<del>3016</del>	<del>Operator II</del>
3012	Truck Driver I	<del>3017</del>	<del>Operator III</del>

3013	Truck Driver II	<del>3019</del>	Operations Foreman
3014	Truck Driver III		

3025	<del>Partsman I</del>	3030	<del>V&amp;E Chargehand</del>
<del>3026</del>	<del>Partsman II</del>	3040	<del>V&amp;E Leadhand</del>
3027	Welder	3041	Transit-Chargehand
3062	Fleet Services Technician	<del>3042</del>	Transit Leadhand

3021	<del>W&amp;WW Foreman</del>	<del>3052</del>	Utilities Electrician
3038	W&WW Leadhand	<del>3053</del>	Utilities Instrument Technician
3047	Infrastructure Foreman	<del>3051</del>	Ut.Inst Tech/Electrician Dual
3039	<del>Project Foreman</del>	<del>3063</del>	Control Systems Technician
3043	Utility Operator I	<del>3020</del>	W&WW Operator (Collection and Distribution)
3044	Utility Operator II	<del>3049</del>	W&WW Operator (Transmission)
3045	Utility Operator III	3037	Wastewater Infrastructure Operator
3046	Utility Leadhand		

If the Employer identifies performance concerns, written feedback shall be provided to the Employee. Where circumstances warrant an extension, the probationary period may be extended up to an additional two hundred and fifty hours (250) hours upon mutual agreement of the Union and the Employer. may be extended up to an additional 500 hours worked, at the Employer's discretion. The Employee and the Union shall be notified of the extension and the cause for the extension.

- An Employee who has previously been employed in a position covered by this agreement will have such previous employment considered part of the probationary period provided that a break in service is one (1) year or less and the Employee is returning to the same job.
- 11.03 At any time, an Employee, categorized as probationary, may be released by the Department Head or the County Administration. Such notice of release will be formalized in writing and will give the reason for the Employee being released as soon as practicable following at the time of the notification of the release.
- 11.04 Employees shall provide at least ten (10) working days written notice of their intention to resign.

- In the event that an Employee terminates his employment with the County without submitting the written notice in compliance with Clause 11.04, the Employee shall not be entitled to receive his regular cheque and/or his final cheque until the next regular distribution of regular wage and salary cheques.
- 11.06 4 A Probationary Employee in accordance with Articles 11.01 shall be granted all of the terms and conditions of this Agreement. During the probationary period the County may terminate the employee. The employee or the Union may grieve the termination, but the decision at Level 3 shall be final and binding and the matter shall not be referred to arbitration under the provisions of this Agreement.

however the termination of a Probationary Employee shall not be have access to the application of any grievance procedure for termination of employment as a result of:

- (a) Unsatisfactory probationary performance; or
- (b) The abolishment of the position occupied by the Probationary Employee

#### ARTICLE 12 - RECLASSIFICATION

- 12.01 When it is necessary to re-classify Regular Employees, the County may re-classify the affected Employees on the basis of Branch seniority, qualifications and ability.
  - (a) An Employee to be re-classified may displace any Employee in a lower classification within that Branch that has less Branch seniority providing that the Employee being reclassified is qualified to perform the duties of that position;
  - (b) Employees with special qualifications may be exempt from re-classification.
- 12.02 In those cases where re-classification according to Clause 12.01 have occurred and vacancies become available in the affected job classification, those Employees who were previously re-classified downward from that classification, shall be returned to that classification on the basis of their seniority so long as that Employee has maintained the ability and qualifications to be appointed to the position.
- 12.03 An apprentice who begins and completes their apprenticeship with the Employer shall be placed on level 2 of the appropriate trades rate upon confirmation of certification.

### **ARTICLE 14 - GRIEVANCES**

# 14.01 Statement of Intent

The Employer and the Union recognize the desirability of resolving differences through joint consultation and respectful and open discussion. Both will attempt

to resolve differences through exploring interests and options together informally means where possible, prior to proceeding to formal process.

# 14.02 <u>Definitions and Scope</u>

- (a) A Personal grievance is a dispute regarding:
  - (i) Alleged unjust treatment or discrimination
  - (ii) Alleged unfair working conditions
  - (iii) Any disciplinary action
  - (iv) Any concern regarding the interpretation, application, operation or alleged violation of this Agreement, including any questions as to the arbitral nature of the matter.
- (b) Grievances which involve a dismissal or a suspension will commence at level II and shall be initiated in writing within fourteen (14) days of the date when the Employee first became aware of the disciplinary action.
- (c) A Policy Grievance is a dispute between the parties which due to its nature is not properly subject of a Personal Grievance and concerns the interpretation, application, operation or alleged violation of this Agreement, including any questions as to the arbitral nature of the matter.
- (d) Formal personal grievances shall be initiated within fourteen (14) calendar days from the date of the incident prompting the grievance. Policy grievances shall be initiated either by the Union or the County in writing, within thirty (30) calendar days from the day that the matter giving rise to the grievance came to the attention of the Designated Officer of the Union or Official of the Employer.
- (d) Formal grievances shall be in writing (email will suffice) and must state the general nature of the grievance, identify the specific clauses in the collective agreement that are being violated, and provide details with respect to the redress or remedy being sought.
- (f) "Days" means calendar days.
- (g) A Union member submitting a grievance has the right to be assisted by an attending Union Representative or Union Steward.

# 14.03 <u>Grievance Process</u>

A Personal Grievance shall be initiated by the Employee as follows:

### Level I

The grievance shall be taken up with the Employee's manager or designate. an Employee wishes to pursue a Personal Grievance, the Employee and/or Union Representative shall advise the Employee's manager or designate that they wish to initiate a Level I grievance. The Employee and/or the

Union representative will discuss the subject of the proposed grievance with the Employee's manager or designate within fourteen (14) calendar days of the receipt of the grievance. date upon which the subject of the grievance, in an attempt to resolve the matter. The manager or designate first level classified supervisor will provide a response to the Employee and/or Union Representative within seven (7) calendar days from when the proposed grievance was discussed.

### Level II

If the Employee continues to be dissatisfied, he should submit the grievance in writing within fourteen (14) days of receiving the response from Level I, to the Department Director. The grievance shall state the general nature of the grievance, the Articles in the Collective Agreement alleged to have been breached and the redress or the remedy sought. The Director, in consultation with Human Resources, will hear the matter and will respond in writing to the Grievor within seven (7) calendar days.

#### Level III

If the Union does not accept the decision of the Department Director, they may request in writing within seven (7) days of receiving the Department Directors decision, that the Chief Commissioner review the grievance. The Chief Commissioner shall hear the matter within ten (10) days and will advise the Union of his decision in writing within fourteen (14) days of the hearing;

The Chief Commissioner may designate an Associate Commissioner to hear and decide on Level III grievances;

If the grievance is not settled as a result of the Chief Commissioner's decision, the Union **or the Employer** may refer the matter to arbitration pursuant to the Labour Relations Code.

### 14.04 <u>Arbitration Board</u>

In the event the matter being grieved is presented to an Arbitration Board, each Party shall bear the expense of its respective nominee and shall bear one-half (1/2) of the expenses of the Chairman of the Arbitration Board.

- 14.05 The Parties may mutually agree to have the matter heard by a single Arbitrator.
- 14.06 The Arbitration Board by its decision shall not alter, amend or otherwise change the terms of this Collective Agreement.
- 14.07 All Union members submitting a grievance have the right to be assisted by an attending Union Representative.

### 14.08 Procedure and Time Limits

Extension of Time Limits: Time limitations set out in the above grievance procedure

may be extended by mutual agreement between the parties and must be in writing.

Time Limits: Time limits and procedures contained in this grievance procedure are mandatory. Failure to pursue a grievance within the prescribed time limits and in accordance with the prescribed procedures shall result in abandonment of the grievance and all rights and recourse to the grievor.

Failure to reply to a grievance in a timely fashion shall advance the grievance to the next level. Grievances so advanced shall be subject to time limits as if a reply had been made on the last allowable day of the preceding level in the procedure.

Policy Grievances may be initiated either by the Union or the County in the following manner:

- (a) Policy Grievances shall be initiated either by the Union or the County in writing, within thirty (30) calendar days from the day that the matter giving rise to the grievance came to the attention of the Designated Officer of the Union or Official of the Employer. The written notice of the grievance shall be submitted to the President of the Union or the Director, Human Resources of Strathcona County and shall stipulate the general particulars of the grievance together with reference to the allegedly violated clause of the Agreement;
- (b) Grievances submitted in accordance with Clause 14.09 shall be referred to a Grievance Committee which shall endeavour to resolve the difference. This Committee will be composed of two representatives of the County and two members of the Union appointed by the Executive. This Committee shall meet within fifteen (15) calendar days of the submission of the written notice by either party;
- (c) After the parties have met in accordance with the preceding clause, each party will give any decision made in writing to the other party within fifteen (15) calendar days. In the event that the two parties reach a mutually acceptable agreement concerning the matter in question, a written and signed notice of mutual agreement will supersede the aforementioned notice of decision;
- (d) In the event that the parties do not reach a mutually acceptable agreement, or that either party does not accept the written notice of decision of the opposite party within thirty (30) calendar days of the Chief Commissioner's decision as per Clause 14.03 either party may notify the other party in writing of its intention to submit the grievance to arbitration pursuant to the provision of the Labour Relations Code.

#### ARTICLE 16 - HOURS OF WORK AND OVERTIME

14.09

16.01 The prescribed regular hours of work for an Employee, other than those referred to in Clause 16.04-5 following, shall be eight (8) hours in any one (1) day or forty

(40) hours in any one (1) calendar week or five (5) working day period. Hours worked in excess of the above shall be calculated as overtime hours and shall be paid accordingly. Notwithstanding the above, when an established schedule provides a pattern that is irregular, the forty (40) hours in one (1) calendar week or five (5) working day period shall mean average over a two (2) week period.

# 16.02 **Overtime**

The overtime pay rates shall be calculated at the rate of two (2.0) times the regular pay rate for all hours worked beyond eight (8) hours in a day.

- An Employee may bank overtime hours at the rate it is earned; up to the maximum of forty (40) regular hours within a calendar year. An Employee may request to be paid out for banked hours up to the time equivalent of forty (40) regular hours per calendar year. Banked hours may be taken as time off by Employees after receiving written authorization from the County designated official who shall consider the wishes and application of the Employee. Up to forty (40) banked regular hours may be carried forward into the next calendar year. Banked hours cannot exceed the time equivalent of eighty (80) regular hours at any one time.
- 16.04 For those Employees who are engaged in road construction, road maintenance or the removal of snow from roads or any of those undertakings, for the purpose of calculating overtime, their regular day shall consist of a twenty-four (24) hour period beginning at midnight. The maximum number of hours of work at regular rates of pay shall be those prescribed or allowed under the Employment Standards Code and Regulations thereunder, but shall not exceed:
  - (a) Ninety (90) hours bi-weekly, except where permitted under the Employment Standard Code;
  - (b) Eighty (80) hours bi-weekly during the period November 01 to April 30, inclusive;
- 16.05 Only hours actually worked, paid vacation (including general holidays) and banked time used to supplement hours are used in accumulation of overtime. Sick leave or other paid or unpaid absences are not included.
- 16.06 For those Employees specified in Clause 16.04 from May 1 to October 31, overtime rates shall be calculated at the rate of two (2.0) times the regular pay rate for all hours worked beyond nine (9) hours in a day. This article only comes into effect when actually scheduled to work nine (9) hours a day.

# 16.06 <u>Posting of Shift Schedules</u>

Barring unforeseen circumstances and emergency situations, shift schedules shall be posted no less than thirty (30) days before the expiry of the current schedule.

# 16.07 Shift Changes

(a) When an Employee is required to change from one shift to another shift, the Employer will provide at least twenty-four (24) hours written notice in advance of

- the start time of the originally scheduled shift and ten (10) hours off between shifts.
- (b) If the twenty-four (24) hours written notice and ten (10) hours off between shifts is not provided to the Employee, the Employee will be paid at the rate of two (2.0) times the regular rate of pay for the hours of the first shift worked.
- (c) If the Employee is on regular scheduled days off when the twenty-four (24) hours written notice is posted, the supervisor may contact the Employee directly to provide verbal twenty-four (24) hours notice change in shift.
- An Employee shall be paid for or expected to work for a minimum of three (3) hours if the regular shift is cancelled or rescheduled and the Employee was not notified of such cancellation or rescheduling on or before the day twelve (12) hours prior to the cancelled or rescheduled shift.
- 16.09 When an Employee calls in ill for a work period affected by the preceding Clause 16.08, that Employee will be paid only for the minimum of three (3) hours that he would have been paid had he reported for work in the regular fashion. This clause shall not apply in cases in which the Employee provides proof of attendance at either medical, dental or optical appointments. In such cases, the Employee will be paid for either the half-day or full-day as appropriate to the circumstances and as provided in Clause 21.03 and 21.04.

## 16.10 Early Reporting

- (a) When an Employee is requested prior to the conclusion of a shift, to report or work more than two (2) or more hours earlier than his regularly scheduled shift, that Employee may (subject to operational requirements) be given a choice of leaving work prior to the end of their regularly scheduled shift or being paid the regular overtime rate for those hours worked prior to the time scheduled for the commencement of the regular shift. The total of the early reporting hours worked and regular shift hours worked will not be less than the number of regularly scheduled hours. Such early reporting arrangement may continue for two consecutive shifts.
  - If requested to report early on the second shift the early reporting hours will be at the overtime rate. If requested for a third consecutive shift, Clause 16.07 will apply.
- (b) When an Employee is requested prior to the conclusion of a shift to report up to two (2) hours earlier or up to two (2) hours later for the next regular shift, that Employee shall not receive the overtime rate for the earlier or later reporting period. The total of the earlier reporting hours worked and regular shift hours worked will not be less than the number of regularly scheduled hours. If requested to report earlier or later on the second shift the earlier or later reporting hours will be at the overtime rate. If requested for a third consecutive shift, Clause 16.07 will apply.

# 16.11 Attendance

When an Employee is unable to report to work, the Employee shall follow the established protocol for their department to report such absence, and at a minimum shall report the absence one (1) hour prior to the start of the shift. within one hour prior to the scheduled shift start time. Failure to comply with this requirement shall result in loss of pay for the shift, and may result in discipline unless the Employee provides a reasonable explanation.

### 16.124 Work Break

All Employees shall be entitled to one fifteen (15) minute work break in each half day worked, or in any overtime work period in excess of two (2) hours. Such work break shall be taken at the work site and the time of such break shall be approved by the immediate supervisor of each Employee. The work break shall be considered part of a work hour for the purposes of the preceding paragraphs.

### 16.13 Sand Shift Workers and Sidewalk Crew

Notwithstanding Clause 16.01, when the needs of the operations require it, Sand Shift Workers and Sidewalk Crew may be required to work a modified shift schedule of up to twelve (12) hours. Modified hours of work shall commence annually on the first full pay period commencing on or after October 15 and shall continue until the first full pay period before or after April 15 based on operational need.

- (a) Overtime will be paid to Sand Shift Workers and Sidewalk Crew who exceed 80 hours of work in a bi-weekly pay period.
- (b) Employees working modified hours of work will have all benefits and entitlements which are expressed in terms of daily or weekly entitlement converted to produce the equivalent hours of benefits and entitlements as they would have had if the hours of work had not been modified. This will result in no loss or gain in Employee benefits and entitlements.

#### ARTICLE 17 - SHIFT AND WEEKEND PREMIUMS

- 17.01 (a) A shift premium of one
  - (a) A shift premium of one dollar and forty cents (\$1.40) one dollar and ninety cents (\$1.90) per hour shall be paid to all Union Employees for those hours worked between seventeen hundred (1700) hours and zero seven hundred (0700) hours Monday to Friday and for all hours worked on Saturday and Sunday and holidays.
  - (b) At no time shall shift premium be included with the Employee's regular rate of pay for purposes of computing overtime payments, other premium payments, or any Employee benefits.

### ARTICLE 20 - RELIEF AND ACTING INCUMBENT PAY

### 20.01 Relief & Acting Pay

Any Employee who works in a classification (within the Bargaining Unit) with a higher rate of pay in any one (1) shift shall receive the higher rate of pay for each hour worked in the higher classification. If this time period exceeds ninety (90) days, effective the ninety-first (91st) day, the Employee will receive the higher rate of pay for all approved vacation, general holiday, illness, bereavement, and compassionate leave while working in the higher classification.

- 20.02 Notwithstanding the premium in Clause 20.09, An Employee appointed to Leadhand or Project Foreman shall not receive an hourly rate of pay that is less than the "County Department Union Employees" that he is responsible for.
- 20.03 Certified Technicians appointed to a seasonal Field Certified Technician assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- 20.04 Vehicle Serviceman II appointed to a seasonal Field Vehicle Serviceman II assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- The qualification or eligibility of an Employee to relieve or be assigned to a higher classification will be the responsibility of management.
- 20.03 An Employee who has become incapacitated by injury or illness may be employed in other work which he can do. Such Employee may not displace an Employee with more seniority.
- 20.04 An Employee who refuses without good and satisfactory reason, to accept the alternate classification or position, may be suspended immediately. The supervisor issuing the suspension will report the suspension to the Department Head (Human Resources) as soon as possible who will notify the Union Executive.
- 20.05 Any Employee affected by the preceding Clauses 20.03 and 20.04 and not satisfied with the action recommended in accordance with Clause 20.04, may appeal the reclassification or suspension in keeping with Clause 14.03 of this Agreement.
- 20.06 Should an Employee who has completed the probationary period be unable to perform his usual duties by reason of the loss or suspension of his motor vehicle operator's license or a similar happening, that Employee may be placed in some other suitable position until the disability ends or is removed. The wages paid during the period of such disability shall be wages appropriate to the position in which the Employee is placed.
- 20.07 Notwithstanding the above clause, within fourteen (14) days of the suspension, an Employee may be allowed to book unused holidays at the wage prevailing prior to suspension of his operator's license.

# **Acting Incumbent**

- 20.08 Any member of the Union may be appointed for one (1) or more days as a Classified supervisor and such appointment will not cause the member to be disqualified from the Union.
- 20.09 If any Employee is appointed temporary Classified Supervisor, they shall receive a premium of two dollars and twenty-five cents (\$2.25) per hour in addition to their regular wage rate for one (1) complete shift or longer.
- 20.10 Any member of the Union may be appointed for a period of time to the position of temporary Classified Supervisor and such appointment will not cause the member to be disqualified from the Union.
- 20.11 Notwithstanding the premium in Clause 20.09, an Employee appointed to Leadhand or Project Foreman shall not receive an hourly rate of pay that is less than the "County Department Union Employees" that he is responsible for.
- 20.12 Certified Technicians appointed to a seasonal Field Certified Technician assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- 20.13 Vehicle Serviceman II appointed to a seasonal Field Vehicle Serviceman II assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.

## ARTICLE 21 - PROVISION FOR ILLNESS AND NORMAL ABSENCE

- 21.01 Sick leave benefits as herein stated are sponsored by the County to protect the Employee in the event of an unavoidable illness or injury not covered by the Alberta Workers' Compensation Act.
- 21.02 When an Employee is unable to report to work, the Employee shall follow the established protocol for their department to report such absence within one hour prior to the scheduled shift start time. Failure to comply with this requirement shall result in loss of pay for the shift, and may result in discipline unless the Employee provides a reasonable explanation. Moved to Article 16
- A Regular Employee, upon hire, shall be entitled to a maximum of eighty (80) hours ten (10) working days per calendar year of leave with pay for any casual illness, which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less. Up to sixteen (16) hours two (2) days of unused casual illness leave per calendar year shall be carried over and may be used in the next calendar year. The maximum paid casual illness leave in a calendar year is ninety-six (96) hours twelve (12) days including the carry over.

- A Temporary Employee after the completion of the standard probationary period, shall be entitled to a maximum of thirty-two (32) hours four (4) working days per calendar year of leave with pay for any casual illness, which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less. Temporary Employees with more than six (6) months of service in the current calendar year shall be entitled to an additional sixteen (16) hours two (2) working days in the current calendar year, of leave with pay for any casual illness. Up to two (2) days of allotment can be used to attend to illness of immediate family members.
- 21.05 Each day or portion of a day of absence due to casual illness used within a calendar year, shall be deducted from the remaining casual illness leave entitlement for that calendar year.
- NEW Where an employee terminates and is rehired in the same calendar year, the casual illness allotment will be the amount that they had in their available bank at the time of departure.
- In any case, an Employee may be required to provide satisfactory proof of illness upon return to work, where it can be shown that reasonable doubt exists in respect to the purpose of an absence claimed to be due to illness. The employee will be provided the reasons for the request. Such proof may take the form of a medical certificate signed by a qualified physician, dental practitioner or other professional as deemed appropriate by the Disability Management Supervisor/Human Resources.
- 21.07 Upon approval (which will not be unreasonably denied), time off of two (2) hours or less for doctor appointments will result in no loss of casual illness or regular time. The Employer may require the Employee to submit proof of attendance at a medical, dental or optical appointment when time off work is granted to attend such appointments.
- 21.08 Each Regular Employee will be entitled to leave of absence with pay for General Illness in accord with the following scale. General Illness is defined as "any illness which causes an Employee to be absent from work for a period of more than three (3) consecutive work days".

To obtain General Illness benefits resulting from the inability of the Employee to perform the duties of his position, a proper medical certificate(s) signed by a qualified physician, dental practitioner or other professional as deemed appropriate by the Disability Management Supervisor/Human Resources is required.

# 21.09 <u>Family Illness Leave</u>

A Regular Employee shall be entitled to a maximum of twenty-four (24) hours three (3) working days per calendar year with pay to attend to illness of immediate family members as per defined in Clause 22.06.

- 21.10 A Temporary Employee may use up to sixteen (16) hours two (2) days of their casual illness allotment, as per Clause 21.04 can be used to attend to illness of immediate family members.
- 21.0911 Effective the calendar year following ratification, Sick leave entitlement for illness longer than three (3) working days shall be as follows:
  - (a) After ninety (90) days three (3) months five hundred (500) hours of continuous employment the Employee shall receive one hundred sixty (160) hours twenty (20) working days at the base rate of pay;
  - (b) On the first day worked, in the next calendar year sick leave entitlement shall be increased to a maximum **two hundred eighty** (280) hours thirty-five (35) working days at the base rate of pay
  - (c) On the first day worked, in subsequent consecutive calendar years sick leave entitlement shall be increased to a maximum of **five hundred twenty (520) hours** sixty-five (65) working days at the base rate of pay, plus any unused casual illness up to a maximum of **six hundred (600) hours** seventy-five (75) working days.
- 21.<del>1012</del> When a day designated as a Paid Holiday under Clause 23.01 falls within a period of General Illness, it will be considered as a paid holiday.
- 21.<del>12</del>13 Transfers, promotions, demotions and/or reclassification shall not affect sick leave entitlement of any Employee.
- 21.<del>11</del>14 Workers Compensation
  - a) An Employee is not eligible to receive sick leave benefits if the absence is due to an injury while in the employ of any other Employer that is covered by Workers' Compensation Benefits.
  - b) Employee's will be eligible to receive benefits in accordance with the County's policies.
- 21.12 Transfers, promotions, demotions and/or reclassification shall not affect sick leave entitlement of any Employee.
- 21.4315 Sick leave entitlement will not be affected while the Employee is in receipt of Alberta Workers' Compensation benefits.
- 21.4416 Regular Employees who are eligible to receive Workers' Compensation benefits shall receive full pay from the County for a period not to exceed six

- (6) months provided that all Workers' Compensation payments are assigned to the County for that period.
- 21.4517 Expenses for medical certificates will be the responsibility of the employee, including medical certificates required to support eligibility for casual or general illness. Where the Employer has requested the employee to attend a medical examination and/or obtain additional medical documentation, the Employer shall reimburse the employee for the cost of the documentation (if any).
- 21.1618 An Employee who has become incapacitated by injury or illness may be employed in other work which he can do. Such Employee may not displace an Employee with more seniority.
- An Employee who refuses without good and satisfactory reason, to accept the alternate classification or position, may be suspended immediately. The supervisor issuing the suspension will report the suspension to the Department Head (Human Resources) as soon as possible who will notify the Union Executive.
- Any Employee affected by the preceding Clauses 21.176 and 21.187 and not satisfied with the action recommended in accordance with Clause 20.04, may appeal the re-classification or suspension in keeping with Clause 14.03 of this Agreement.
- 21.1921 Should an Employee who has completed the probationary period be unable to perform his usual duties by reason of the loss or suspension of his motor vehicle operator's license or a similar happening, that Employee may be placed in some other suitable position until the disability ends or is removed. The wages paid during the period of such disability shall be wages appropriate to the position in which the Employee is placed.
- 21.<del>2022</del> Notwithstanding the above clause, within fourteen (14) days of the suspension *of* their operator's license, an Employee may be allowed to book unused holidays at the wage prevailing prior to suspension of his operator's license.

### **ARTICLE 22 - LEAVES OF ABSENCE**

Other Leaves of Absence, without pay, may be granted at the sole discretion of the County, and in accordance with the Alberta Employment Standards Code, upon written application by the affected Employee at least seven (7) days prior to the beginning date of the applied for Leave of Absence. To be eligible for such leave, an Employee shall have been employed with the County for ninety (90) days of continuous employment, with the exception of reservist leave which is twenty-six (26) consecutive weeks, and must have his application endorsed or approved by the Department Head. The application shall include the reason for the request as well as the dates of the requested Leave of Absence. Such approval should not be

unreasonably withheld.

- In unusual circumstances or by mutual agreement, the requirement stated in Clause 22.01 for application to be made seven (7) days prior to the commencement of the requested leave of absence may be waived. Application for such waiver is to be made directly to the Department Head.
- In the case of an Employee who takes a leave of absence under the provisions of Clause 22.01, the benefits and conditions to which that Employee is entitled as a result of length of service in the County shall not be adversely affected as a result of having been granted such leave of absence.
- 22.04 An Employee on such approved leave may not engage in other employment for gain while on such leave without the express written consent of the County.

# 22.05 Court and Jury Duty

Leave of Absence shall be provided to an Employee who is required to serve as a juror, or is subpoenaed as a witness in a private capacity in any court proceeding. The Employee shall receive their regular rate of pay and shall remit to the Employer any allowance they may receive for such duty or service.

If the Employee is subpoenaed as a witness to appear in court in their official capacity to give evidence or to produce Strathcona County records, the Employee shall receive their regular rate of pay during regular working hours and is eligible to receive overtime pay to appear during non-work hours. The Employee shall remit to the Employer any allowance they may receive for such duty or service.

The Employee shall present proof of such duty or service and the amount of remuneration or allowance received in all cases.

Employees required to appear in court in a capacity other than subpoenaed as a witness or jurist will be granted leave without pay.

### 22.06 Bereavement Leave

An Employee shall be granted Bereavement Leave of Absence with pay for not more than five (5) consecutive days because of the death of spouse (includes common-law), child (includes step, foster child or ward), parent, (includes step or foster parent), brother, sister, parent of spouse, guardian, son-in-law, daughter-in-law, grandchild or any other relative who is a member of the Employee's household, or the spouse of any of the foregoing. In addition to the above, the Employer may grant up to one (1) day travel to attend the funeral if the travel is five hundred (500) kilometres or more away.

- Leave of absence for one (1) full day, with pay, shall be granted at the time of to attend the funeral or death of a grand-parent, grandparent-in-law, brother-in-law or sister-in-law, aunt, uncle, niece or nephew, if the funeral is on a work day.
- 22.08 Upon written application, additional Compassionate or Bereavement Leave of Absence with pay, due to exceptional circumstances, may be granted an Employee

at the discretion of the Chief Commissioner.

One (1) day of leave may be granted without loss of pay or benefits to attend as a pallbearer or mourner for a deceased Employee. Where the family of a deceased Employee requests pallbearers from the Union, the County may grant the necessary leave with pay for up to six (6) pallbearers.

# 22.1122.10 Family Illness Leave

A Regular Employee shall be entitled to a maximum of twenty-four (24) hours three (3) working days per calendar year with pay to attend to illness of immediate family members as per defined in Clause 22.06.

22.11 A Temporary Employee may use up to sixteen (16) hours two (2) days of their casual illness allotment, as per Clause 21.04 can be used to attend to illness of immediate family members.

# 22.10 <u>Compassionate Leave</u>

An Employee shall be granted Compassionate Leave of Absence with pay for not more than five (5) consecutive days because of the critical illness of a member of his immediate family as per defined in Clause 22.06 or any other relative who is a member of the Employee's household. The Employee shall provide the County with a letter from the attending physician concerning the fact of the critical illness.

Any additional Compassionate Care Leave shall be provided in accordance with *Alberta Employment Standards Code*.

- 22.11 A Regular Employee shall be entitled to a maximum of three (3) working days per calendar year with pay to attend to illness of immediate family members as per Clause 22.06.
- 22.12 One (1) day leave with pay will be allowed for an Employee to be present at the birth or adoption proceeding of an Employee's child, providing such occurrence is on a scheduled work day.

# 22.<del>13</del>11 <u>Maternity Leave</u>

Maternity Leave will be granted to Employees in accordance with the Alberta Employment Standards Code.

- (a) Employees must have been employed with Strathcona County for ninety (90) days of continuous employment.
- (b) The maternity leave is not more than sixteen (16) weeks starting at any time during the thirteen (13) weeks immediately before the estimated date of delivery.
- (c) The Employee must give six (6) weeks' written notice of the date she intends to begin leave and four (4) weeks' notice of the date she intends to return to work.

(d) If an Employee's pregnancy results in a miscarriage or stillbirth within sixteen (16) weeks of their due date, they are entitled to sixteen (16) weeks of maternity leave.

# 22.124 <u>Health Related Portion of the Maternity Leave</u>

During the one-week Employment Insurance waiting period, Strathcona County will top-up salary to ninety-five percent (95%) of the Employee's regular earnings. Verification that Employees are in receipt of employment insurance benefits is necessary to provide the top-up benefit. A medical certificate must be provided to Human Resources, confirming the date the child was born and the duration of the leave that the physician indicates is health related. During the health-related portion of the maternity leave Strathcona County will top-up salary to ninety-five percent (95%) of the Employees regular earnings. The benefit coverage that the Employee has at the time of her leave will continue during the health related portion of the absence. At the end of this period, parental leave will commence, and benefit coverage will continue if the Employee pays both Employer and Employee portions of the cost of the premiums.

When a pregnancy interferes with the performance of duties or the Employer has a concern for the Employee's health, the Employer may reassign the Employee's duties at no loss of pay, or by notice in writing to the Employee, require them to commence sick leave under Clause 21.08.

### 22.1416 Parental Leave

Employees who have been employed with Strathcona County for ninety (90) days of continuous employment are entitled to parental leave without pay for a maximum of sixty-two (62) weeks. The Employee must give at least six (6) weeks written notice of the date they intend to begin parental leave and four (4) weeks' notice of the date they intend to return to work. During that time they may receive benefit coverage by paying the Employee and Employer portions of the cost of the premiums.

22-1517 Employees wishing to return to work prior to the agreed return date may do so with two (2) weeks' notice.

# 22.1618 Adoption Leave

One (1) day leave with pay will be allowed for an Employee to be present at the birth or adoption proceeding of an Employee's child, providing such occurrence is on a scheduled workday.

The Employer will grant adoption leave to Employees who are adopting a child. The Employee must have been employed with Strathcona County for ninety (90) days of continuous employment. Written notice of leave must be provided to the Employer at least two (2) weeks before Employees can reasonably expect to obtain

custody of the child.

- 22.1719 If Employees cannot provide at least two (2) weeks written notice, they must notify the Employer immediately after receiving notice of the adoption.
- Adoption leave consists of not more than sixty-two (62) consecutive weeks of leave without pay within seventy-eight (78) weeks after the child is placed with the adoptive parents. During that time they may receive benefit coverage by paying the Employee and Employer portions of the cost of the premiums.
- 22.1921 Only one (1) parent of a child referred to in Clause 22.20 will be granted adoption leave under this clause.

# 22.2022 Shared Leave

If both parents are Employees of Strathcona County parental or adoption leave may be taken wholly by one of the Employees or be shared by both. Only one (1) parent will be granted parental or adoption leave at a time.

# 22.2123 Anniversary Date

Employees who have been on maternity, parental or adoption leave will have their anniversary date adjusted for evaluation and increment purposes only. They will not earn vacation leave during this period, the exception being for maternity health related reasons.

### 22.2224 Additional Leave

Leave of absence without pay beyond the approved maternity, parental or adoption leave may be granted upon written application up to a combined maximum of twelve (12) months at the discretion of the Employer.

### 22.23<del>25</del> Domestic Violence Leave

Domestic violence leave shall be granted in accordance with the *Alberta Employment Standards Code*.

# 22.2426 Military Leave

Military leave shall be granted in accordance with the *Alberta Employment Standards Code*.

# 22.25<del>27</del> Citizenship Ceremony Leave

Citizenship Ceremony Leave shall be granted in accordance with the *Alberta Employment Standards Code*.

#### **ARTICLE 23 - HOLIDAYS**

23.01 Employees shall be entitled to one (1) day's paid leave for each of the following

specified or general holidays:

New Year's Day Labour Day

Family Day Thanksgiving Day
Good Friday Remembrance Day
Easter Monday Christmas Day
Victoria Day Christmas Floater

Canada Day Boxing Day

1st Monday in August Truth and Reconciliation Day

And any other day designated as a General Holiday by the Lieutenant Governor in Council under the Employment Standards Code.

- Employees shall receive the recognized specified or general holidays for which they are eligible, with pay; or other days with pay in lieu of such holidays, or pay in lieu as approved by the County, providing they are available for work immediately preceding and following the days or an otherwise approved leave of absence, or they provide a medical certificate attesting to their illness on the day prior or following the general holiday.
- 23.03 The provisions for pay in lieu of such specified or general holidays referred to in Clauses 23.01 and 23.02 shall be based on the number of regular hours which would have been worked had that day been a regular working day.
- In the event a specified or general holiday falls on an Employee's regular day off, an additional day with pay shall be granted in lieu of that day.
- 23.05 If any of the above specified or general holidays should occur during an Employee's vacation, such vacation shall be extended by one day or an additional day with pay shall be granted in lieu of that day.
- 23.06 Temporary Employees, upon hire, shall be eligible for specified or general holidays as per Clause 23.02.
- 23.07 The Christmas Floater holiday shall be observed to give five (5) consecutive days off including the weekend as follows:
  - (a) On December 24 when Christmas Day falls on a Tuesday, a Thursday, a Friday or Saturday;
  - (b) On December 27 when Christmas Day falls on a Monday or a Wednesday;
  - (c) On December 28 when Christmas Day falls on a Sunday.

### **ARTICLE 24 - VACATION LEAVE**

- 24.01 Annual vacation entitlement with pay shall be as follows:
  - (a) An Employee who has completed less than six (6) years of service shall earn

- vacation leave at the rate of four point sixty-two (4.62) hours per bi-weekly period of continuous service [fifteen (15) days per year] or five point two (5.20) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
- (b) An Employee who has completed six (6) years but less than fourteen (14) years of service shall earn vacation leave at the rate of six point fifteen (6.15) hours per bi-weekly period of continuous service [twenty (20) days per year] or six point ninety-two (6.92) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
- (c) An Employee who has completed fourteen (14) years of service but less than twenty two (22) years shall earn vacation leave at the rate of seven point seven (7.70) hours per bi-weekly period of continuous service [twenty-five (25) days per year] or eight point sixty-seven (8.67) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
- (d) An Employee who has completed twenty two (22) or more years of service shall earn vacation leave at the rate of nine point twenty-three (9.23) hours per bi-weekly period of continuous service [thirty (30) days per year] or ten point thirty-nine (10.39) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week.
- 24.02 If Employees become ill on an annual vacation leave for more than three (3) consecutive days-paid vacation days, the applicable days will be deemed general illness, and they will have the applicable number of vacation days reinstated upon application-and provision of a medical certificate.
- 24.03 The number of days of vacation leave which may be accumulated by an Employee is limited to the number he can earn in eighteen (18) months of continuous service.

The County will automatically pay out any excess vacation leave over the maximum accumulation at the beginning of each December.

When vacation pay is paid out rather than being taken as vacation, the payout shall be based on the Employee's rate of pay at the time of payout.

- 24.04 Vacation leave may be taken in the calendar year it is earned. However, vacation leave may not be taken until it has been earned. The vacation leave may be granted as follows:
  - (a) In one continuous period;
  - (b) As single days off or some other combination;
  - (c) If an Employee earns vacation at a rate of four point sixty-two (4.62) hours per bi-weekly period of continuous service, one (1) week vacation leave may be taken as single days off or some other combination provided the total does not exceed five work days in any calendar year;

- (d) If an Employee earns vacation at a rate of six point fifteen (6.15) hours per bi-weekly period of continuous service or seven point seven (7.70) hours per bi-weekly period of continuous service, two (2) weeks' vacation leave may be taken as single days off or some other combination provided the total does not exceed ten (10) work days in any calendar year;
- (e) If an Employee earns vacation at a rate of nine point twenty-three (9.23) hours per bi-weekly period of continuous service, three (3) weeks' vacation leave may be taken as single days off or some other combination provided the total does not exceed fifteen (15) work days in any calendar year.
- In addition to the automatic payout of excess vacation referred to in 24.03, an Employee may be permitted to receive a cash payment in lieu of vacation leave, provided any and all obligations have been met in accordance with Employment Standards legislation and the Employee has received written approval from the Department Head.

Temporary Employees shall have vacation pay added to their bi-weekly pay cheques and shall be entitled to unpaid vacation leave in accordance with the Employment Standards Code.

- 24.06 An Employee who does not complete one (1) full year of service shall be entitled to vacation pay in accordance with the Employment Standards Code and regulations thereunder.
  - (a) Temporary Employees will receive vacation pay at four percent (4%) of their gross regular earnings during their first four thousand (4000) regular standard hours.
  - (b) Temporary Employees will receive vacation pay at six percent (6%) of their gross regular earnings after completion of four thousand (4000) regular standard hours.
  - (c) If a Temporary Employee has a break in service greater than eighteen (18) months then vacation pay shall be paid at four percent (4%) of their gross regular earnings as per Clause 24.076 a) above.
- 24.07 Regular vacation leave shall be taken by Employees after receiving written authorization from the County or its designated official who shall consider the wishes and application of the Employee. Length of regular service Seniority with the County shall be the determining factor in resolving any dispute or conflict regarding preference for the time at which vacations are to be taken. Changes in vacation scheduling shall be authorized in writing.
- 24.08 Vacation hours will be paid out at the Employee's base rate in place at the time the vacation is taken.

#### **Summer Vacation Schedules**

24.09 Summer Vacation Schedules

Employees must submit summer vacation applications to the Director or his designate by March 15th of each year. The Employer will post schedules, as approved by the Director, by May 1st of each year.

### **ARTICLE 25 - BENEFIT AND PENSION PLANS**

### 25.01 Benefit Plans

The County shall contribute on behalf of all participating Employees in group insurance and benefit plans, which provide minimum benefit levels agreed upon by the County and the Union at rates outlined in the following clauses:

- (a) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Strathcona County Benefit Plan Life, Accidental Death and Dismemberment and Long-Term Disability Insurance. All members of the Union hired after July 1, 1978, shall be members of the said plan as a condition of employment;
- (b) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Extended Health Care option of the Strathcona County Employee Benefit Plan;
- (c) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Dental option of the Strathcona County Employee Benefit Plan;
- (d) The County shall pay seventy five percent (75%) of the total premium for each eligible Union member enrolled in the Alberta Health Care Insurance Plan through the Strathcona County.

The County's liability shall be limited to the payment of premiums, where required, for the various plans and the County shall be relieved of any further liability with respect to the plans.

25.02 In the event that the County should undertake to transfer group insurance and benefit plans from one carrier to another, it is agreed that the Union will be consulted prior to the negotiation of such coverage.

# **Benefits for Temporary Employees**

### 25.03 <u>Benefits for Temporary Employees</u>

To be eligible to participate in the Strathcona County Employee Benefit plans, Temporary Employees must meet the following conditions:

(a) All Temporary Employees who have been employed for twelve (12) consecutive cumulative months or more, provided there has been no break in service for longer than six (6) months, shall receive the same benefits as Regular Employees, with the exception of pension and the

LTD benefit entitlement will be limited to a maximum of twenty-four months.

For the purposes of calculating the twelve (12) cumulative months, leaves of absence of more than thirty (30) calendar days shall be excluded.

(b) An exception to the above shall be made for Temporary Employees replacing Regular Employees on long term disability. A Temporary Employee in this situation will be eligible for benefits after six (6) consecutive months of employment.

### Pension Plan

### 25.04 Pension Plan

The County participates in the Local Authorities Pension Plan. Participation in this plan is compulsory for "Regular Employees" after they have completed one (1) year of service or immediately if transferring from another Local Authority Board Employer.

25.05 The County shall make payroll deductions as follows:

- (a) The County shall deduct legislated Revenue Canada deductions and Local Authorities Pension Plan contributions as required from each bi-weekly pay cheque;
- (b) The current authorized deductions for Canada Savings Bonds will be taken each pay cheque;
- (c) The Employee contribution to the Strathcona County Employee Benefit Plan will be deducted from the first bi-weekly pay cheque. each month.

# **ARTICLE 29 - TEMPORARY EMPLOYEES**

- 29.01 Temporary Employees shall be entitled to the terms and conditions of this Agreement, except where Regular Employees are referenced. Seniority for employment periods of less than six (6) months shall not be accumulated.
- 29.02 A Temporary Employee shall be considered probationary as per Article 11.01 and shall not serve probation for subsequent years of employment unless a break of service of greater than one (1) year has occurred or the Employee is returning to a different position. Temporary Employees shall have access to the application of the grievance procedure except in connection with termination of employment as a result of either:
  - (a) Unsatisfactory performance, or;
  - (b) The abolishment of the position occupied by the Temporary

Employee.

29.03 For the purpose of determination of Pay Grade and Step a Temporary Employee returning for subsequent employment within one (1) year may be required to undertake the appropriate probationary period unless he is returning to a position in which he has previously fulfilled the probationary requirements.

29.04 When Temporary Employees are recalled, it shall be based on:

- (a) A satisfactory performance evaluation from the previous year of employment;
- (b) The Employee having qualifications to do the job for which he is recalled;
- (c) The Employee's accumulated service.

The opportunity for recall shall exist for six (6) months within the Branch.

This clause will not apply to Employees participating in the NAIT or other post-secondary Water and Wastewater Training Program.

### ARTICLE 30 - GENERAL WAGE ADMINISTRATION

30.01 The rates of pay for all Employee members of the Union excluding Apprentices, covered by the Agreement shall be as specified in the attached Schedule A.

Temporary Employees returning for subsequent employment within one (1) year shall have their prior service recognized for the purposes of determining Pay Grade and Step.

- 30.02 An Employee is entitled to the rates indicated in Schedule A in the following way:
  - (a) To Step One (1) upon hiring;
  - (b) To Step Two (2) upon completion of the five hundred (500) hours of work in the classification and recommendation of the appropriate supervisor;
  - (c) To Step Three (3) upon completion of an additional fifteen hundred (1,500) hours of work in the classification and the recommendation of the appropriate supervisor
  - (d) Water & Wastewater Operators step progression is as stated in the Water and Wastewater Job Summary attached to the Collective Agreement. Water and Wastewater Operators step progression shall be:
    - (i) To Step One (1) upon hiring;
    - (ii) To Step Two (2) upon completion of five hundred (500) hours of work

in the classification, additional qualifications (in accordance with the job description), and recommendation of the appropriate supervisor;

- (iii) To Step Three (3) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (iv) To Step Four (4) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (v) To Step Five (5) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (vi) To Step Six (6) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;

# (e) Construction Operator step progression shall be:

- (i) To Step One (1) upon hiring;
- (ii) To Step Two (2) upon completion of two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (iii) To Step Three (3) upon completion of an additional two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (iv) To Step Four (4) upon completion of an additional two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the description), and the recommendation of the appropriate supervisor.

# (f) Water and Wastewater Leadhand step progression:

Water & Wastewater Operator,	Water & Wastewater Leadhand,
Steps 1 to 4	Step 1
Water & Wastewater Operator, Step 5	Water & Wastewater Leadhand, Step 2

Water & Wastewater Operator, Step	Water & Wastewater Leadhand,
6	Step 3

(g) Water and Wastewater Infrastructure Repair Technician step progression:

Water & Wastewater Operator, Steps 1 to 4	W&WW Infrastructure Repair Technician, Step 1
Water & Wastewater Operator, Step 5	W&WW Infrastructure Repair Technician, Step 2
Water & Wastewater Operator, Step 6	W&WW Infrastructure Repair Technician, Step 3

- (h) An appropriate form will be completed and placed In the Employee's personnel file when the wage rate is amended.
- 30.03 Should an Employee not be recommended by the appropriate supervisor for advancement as per Clause 30.02 b), c), d) or e), the Administration shall advise that Employee of the reasons in writing, and in addition:
  - (a) What action might be taken in order for the wage to be advanced.
  - (b) In the event that the wage will not be advanced, what alternate action the administration intended to take.
- 30.04 An Employee affected by Article 30.03 may file a grievance for a failure to be recommended for an increase.
- The rate of pay established in Schedule A shall apply during the term of this Agreement. Employees shall be paid bi-weekly. On each pay day, each Employee shall be provided with an itemized statement of his wages, overtime and other earnings and deductions.

# 30.06 Flag Person

The Employer may hire Temporary Employees as Flagperson at a rate in accordance with Schedule A. Should such employees be assigned to any other work they shall be paid in accordance with Clause 20.01. No one shall be regularly assigned duties as Flagperson until they have successfully completed an approved Flagperson's training course. Employees taking the Flagperson training shall be paid the appropriate rate of pay while training.

### ARTICLE X - DIRECT RESPONSILBE PAY (DRP)

- X.01 When a Water and Wastewater Operator or Water and Wastewater Leadhand is assigned to the Utilities On-Call Schedule as On-Call #1, On-Call #2 or Transmission Operator (for emergent calls) and deemed to be in direct responsible charge (DRC), the Employee shall be paid an additional \$2.25 per hour on the base rate of pay, for all hours worked between sixteen hundred and thirty (1630) hours and zero eight hundred (0800) hours Monday to Friday and for all hours worked on Saturday and Sunday and Holidays.
  - (a) At no time shall Operators called out to assist the On-Call #1, On-Call #2 or Transmission Operator with emergency service or other similar callouts receive DRP.
- X.02 Where applicable, an Employee shall be eligible to receive DRP in addition to overtime, shift and weekend premiums.
- X.03 At no time shall DRP be included with the Employee's regular rate of pay for the purpose of calculating overtime payments, other premium payments, or any Employee benefits.

### **Wage Increases**

```
January 1, 2020 – 1%
January 1, 2021 – 1%
January 1, 2022 – 1%
January 1, 2023 – 1.5%
```

The following is included for clarity, but will not form part of the collective agreement:

- Retroactive increases will be payable on all hours worked within 3 months of ratification:
- The new compensation structure and pay grid and increased premiums will be effective on the first pay period in the month following ratification
- Premiums referenced in Article 17 will be included in pensionable earnings.
- Temporary Employees will be eligible for benefits within two months of ratification
- All other agreed to amendments will be effective upon ratification.

The following articles have no amendments and will remain as current, except for any consequential amendments that may arise as a result of other agreed to amendments:

Article 5 – Management Rights Article 7 – Time off for Union Business Article 13 – Discipline Article 18 – Call out & Standby

#### **SCHEDULE A**

- Schedule A below is effective in the first pay period of the month following ratification, including the following changes:
  - Merged Labourer, Truck Driver (now Truck Driver Class 1) and Construction
     Operator (former Utility Operator) classifications
    - Note rates at first two steps for merged Labourer are lower than in previous tentative agreement
  - Retitled positions with same step levels: Equipment Operator 1, 2, and 3 (vs. Operator 1, 2, and 3); W&WW Infrastructure Repair Technician (vs. W&WW Project Foreman)
    - Note rates for Equipment Operator steps 2 and 3 are lower than in previous tentative agreement
  - o Deleted: Flag Person
  - Added: North Yardperson, Shipper-Receiver, Materials Management
  - Gender neutral title changes: PW Project Foreman Foreperson, Operations
     Foreman Foreperson, W&WW Foreman Foreperson, Vehicle Serviceman I
     Serviceperson I, Vehicle Serviceman II Serviceperson II, Partsman I Parts Person I,
     and Partsman II Parts Person II, and Infrastructure Foreman Foreperson

Job Code	Position		Hourly Rate	)
3001	Flag Person	<del>16.81</del> <del>17.42</del>		
3002	Labourer I	<del>20.78</del>	21.14	
3003	<del>Labourer II</del>	23.41	24.18	
3004	<del>Labourer III</del>	<del>25.77</del>	<del>26.45</del>	<del>26.90</del>
	PW Labourer	<del>20.78</del>	<del>23.41</del>	26.90
	Plus 1% 2020	<del>20.99</del>	<del>23.6</del> 4	<del>27.17</del>
	Plus 1% 2021	21.20	<del>23.88</del>	27.44
	Plus 1% 2022	21.41	24.12	27.71
TBD	Plus 1.5% 2023	21.73	24.48	28.13
	Pickup/Courier Driver	<del>24.85</del>	<del>25.40</del>	
	Plus 1% 2020	<del>25.10</del>	<del>25.65</del>	
	Plus 1% 2021	<del>25.35</del>	<del>25.91</del>	
	Plus 1% 2022	25.60	26.17	
3005	Plus 1.5% 2023	25.98	26.56	
,				
	Survey Aide I	<del>24.52</del>	<del>25.23</del>	
	Plus 1% 2020	<del>24.77</del>	<del>25.48</del>	
	Plus 1% 2021	<del>25.02</del>	<del>25.73</del>	
	Plus 1% 2022	25.27	25.99	
3006	Plus 1.5% 2023	25.65	26.38	
	Survey Aide II	<del>26.43</del>	<del>27.05</del>	<del>27.74</del>
	Plus 1% 2020	<del>26.69</del>	<del>27.32</del>	<del>28.02</del>
	Plus 1% 2021	<del>26.96</del>	<del>27.59</del>	<del>28.30</del>
	Plus 1% 2022	27.23	27.87	28.58
3007	Plus 1.5% 2023	27.64	28.29	29.01
	PW Tech I	<del>31.71</del>	<del>32.46</del>	<del>33.24</del>
	Plus 1% 2020	<del>32.03</del>	<del>32.78</del>	<del>33.57</del>
	Plus 1% 2021	<del>32.35</del>	<del>33.11</del>	33.91
	Plus 1% 2022	32.67	33.44	34.25
3008	Plus 1.5% 2023	33.16	33.94	34.76
	PW Tech II	<del>36.60</del>	<del>37.08</del>	<del>37.46</del>
	Plus 1% 2020	<del>36.97</del>	<del>37.45</del>	<del>37.83</del>
	Plus 1% 2021	<del>37.34</del>	<del>37.82</del>	38.21
2555	Plus 1% 2022	37.71	38.20	38.59
3009	Plus 1.5% 2023	38.28	38.81	39.17
Γ		20.40	20.75	1 20.22
	Gravel Checker	<del>29.10</del>	<del>29.76</del>	30.23
	Plus 1% 2020	<del>29.39</del>	<del>30.06</del>	<del>30.53</del>
	Plus 1% 2021	<del>29.68</del>	<del>30.36</del>	<del>30.84</del>
2225	Plus 1% 2022	29.98	30.66	31.15
3032	Plus 1.5% 2023	30.43	31.12	31.62

	Concrete Worker I	30.54	31.20	31.92
	Plus 1% 2020	30.85	31.51	32.24
	Plus 1% 2021	<del>31.16</del>	31.83	<del>32.56</del>
	Plus 1% 2022	31.47	32.15	32.89
3010	Plus 1.5% 2023	31.94	32.63	33.38
	Concrete Worker II	32.05	32.95	33.85
	Plus 1% 2020	<del>32.37</del>	33.28	34.19
	Plus 1% 2021	<del>32.69</del>	<del>33.61</del>	34.53
	Plus 1% 2022	33.02	33.95	34.88
3011	Plus 1.5% 2023	33.50	34.46	35.40
<del>3012</del>	Truck Driver I	<del>27.95</del>	<del>28.55</del>	29.14
3013	Truck Driver II	<del>29.60</del>	30.33	31.03
3014	Truck Driver III	32.32	32.97	33.76
	Truck Driver - Class 1	32.32	32.97	33.76
	Plus 1% 2020	<del>32.6</del> 4	33.30	34.10
	Plus 1% 2021	<del>32.97</del>	33.63	34.44
	Plus 1% 2022	33.30	33.97	34.78
TBD	Plus 1.5% 2023	33.80	34.48	35.30
3015	<del>Operator I</del>	<del>27.95</del>	28.67	29.08
<del>3016</del>	<del>Operator II</del>	31.47	32.20	32.84
3017	Operator III	33.52	34.37	35.25
	Equipment Operator I	27.95	28.67	29.08
	Plus 1% 2020	28.23	28.96	29.37
	Plus 1% 2021	<del>28.51</del>	29.25	<del>29.66</del>
	Plus 1% 2022	28.80	29.54	29.96
TBD	Plus 1.5% 2023	29.23	29.98	30.41
	Equipment Operator II	31.47	32.20	32.84
	Plus 1% 2020	<del>31.78</del>	32.52	33.17
	Plus 1% 2021	<del>32.10</del>	32.85	33.50
	Plus 1% 2022	32.42	33.18	33.84
TBD	Plus 1.5% 2023	32.91	33.68	34.36
	Equipment Operator III	33.52	34.37	35.25
	Plus 1% 2020	33.86	34.71	35.60
	Plus 1% 2021	<del>34.20</del>	35.06	<del>35.96</del>
	Plus 1% 2022	34.54	35.41	36.32
TBD	Plus 1.5% 2023	35.06	35.94	36.86
		** *** *** *** *** *** *** *** *** ***		
	Sign Person	<del>33.26</del>	34.12	35.00
	Plus 1% 2020	<del>33.59</del>	34.46	35.35
	Plus 1% 2021	<del>33.93</del>	34.80	35.70

	Plus 1.5% 2023	34.78	35.68	36.60
	Yardperson	<del>33.26</del>	<del>34.12</del>	<del>35.00</del>
	Plus 1% 2020	<del>33.59</del>	<del>34.46</del>	35,35
	Plus 1% 2021	<del>33.93</del>	<del>34.80</del>	<del>35.70</del>
	Plus 1% 2022	34.27	35.15	36.06
3018	Plus 1.5% 2023	34.78	35.68	36.60
	North Yardperson	<del>36.76</del>	<del>37.6</del> 4	
	Plus 1% 2020	<del>37.13</del>	<del>38.02</del>	
	Plus 1% 2021	<del>37.50</del>	<del>38.40</del>	
	Plus 1% 2022	37.88	38.78	
TBD	Plus 1.5% 2023	38.45	39.36	

	PW Leadhand	<del>36.76</del>	<del>37.64</del>	
	Plus 1% 2020	37.13	38.02	
	Plus 1% 2021	<del>37.50</del>	38.40	
	Plus 1% 2022	37.88	38.78	
3036	Plus 1.5% 2023	38.45	39.36	
	PW Project <del>Foreman</del> <b>Foreperson</b>	38.32	39.36	
	Plus 1% 2020	38.70	<del>39.75</del>	
	Plus 1% 2021	39.09	40.15	
	Plus 1% 2022	39.48	40.55	
3035	Plus 1.5% 2023	40.07	41.16	
***************************************	Operations Foreman Foreperson	41.48	42.36	43.30
	Plus 1% 2020	41.89	42.78	43.73
	Plus 1% 2021	42.31	43.21	44.17
	Plus 1% 2022	42.73	43.64	44.61
3019	Plus 1.5% 2023	43.37	44.29	45.28
	W&WW Labourer	20.78	24.18	<del>26.90</del>
	Plus 1% 2020	<del>20.99</del>	24.42	<del>27.17</del>
	Plus 1% 2021	<del>21.20</del>	<del>24.66</del>	27.44
	Plus 1% 2022	21.41	24.91	27.71
3034	Plus 1.5% 2023	21.73	25.28	28.12
	Wastewater Infrastructure Operator	30.27	<del>32.84</del>	35.25
	Plus 1% 2020	<del>30.57</del>	<del>33.17</del>	<del>35.60</del>
	Plus 1% 2021	30.88	<del>33.50</del>	<del>35.96</del>
	Plus 1% 2022	31.19	33.84	36.32
3037	Plus 1.5% 2023	31.66	34.35	36.86
	W&WW Operator (Collection & Distribution)	<del>30.27</del>	<del>32.77</del>	<del>34.26</del>
	Plus 1% 2020	<del>30.57</del>	<del>33.10</del>	<del>34.60</del>
	Plus 1% 2021	30.88	<del>33.43</del>	<del>34.95</del>
	Plus 1% 2022	31.19	33.76	35.30
3020	Plus 1.5% 2023	31.66	34.27	35.83
		<del>36.72</del>	<del>38.45</del>	40.91

	Plus 1% 2020	37.09	38.83	41.32
	Plus 1% 2021	<del>37.46</del>	39.22	41.73
	Plus 1% 2022	37.83	39.61	42.15
	Plus 1.5% 2023	33.40	40.20	42.78
	W&WW Operator (Transmission)	30.27	32.77	34.26
	Plus 1% 2020	30.57	33.10	34.60
	Plus 1% 2021	30.88	33.43	34.95
	Plus 1% 2022	31.19	33.76	35.30
3049	Plus 1.5% 2023	31.66	34.27	35.83
	Plus 1% 2020	36.72	38.45	40.91
	Plus 1% 2021	37.09	38.83	41.32
	Plus 1% 2022	<del>37.46</del>	39.22	41.73
	Plus 1.5% 2023	37.83	39.61	42.15
	W&WW Leadhand	38.27	40.00	42.46
	Plus 1% 2020	<del>38.65</del>	40.04	42.88
	Plus 1% 2021	39.04	40.80	43.31
	Plus 1% 2022	39.39	41.21	43.74
3038	Plus 1.5% 2023	40.29	41.83	44.18
3039	W&WW Project Foreman	39.66	41.36	43.87
	W&WW Infrastructure Repair Technician	39.66	41.36	43.87
	Plus 1% 2020	4 <del>0.06</del>	41.77	44.31
	Plus 1% 2021	40.46	4 <del>2.19</del>	44.75
	Plus 1% 2022	40.86	42.61	45.20
TBD	Plus 1.5% 2023	41.47	43.25	45.88
	W&WW <del>Foreman</del> Foreperson	<del>42.42</del>	44.08	4 <del>5.61</del>
	Plus 1% 2020	42.84	44.52	4 <del>6.07</del>
	Plus 1% 2021	43.27	44 <b>.97</b>	4 <del>6.53</del>
	Plus 1% 2022	43.70	45.42	47.00
3021	Plus 1.5% 2023	44.36	46.10	47.71
	W&WW Electrician	44.11	45.78	<del>47.28</del>
	Plus 1% 2020	44.55	4 <del>6.2</del> 4	4 <del>7.75</del>
	Plus 1% 2021	44.00	4 <del>6.70</del>	48.23
	Plus 1% 2022	45.45	47.17	48.71
3022	Plus 1.5% 2023	46.13	47.88	49.44
	Utilities Facilities Maintenance Technician	<del>52.47</del>	<del>54.23</del>	<del>56.09</del>
	Plus 1% 2020	<del>52.99</del>	<del>54.77</del>	<del>56.65</del>
	Plus 1% 2021	<del>53.52</del>	<del>55.32</del>	<del>57.22</del>
	Plus 1% 2022	54.06	55.87	57.79
3033	Plus 1.5% 2023	57.87	56.71	58.66
			Т	ı
	Vehicle <del>Serviceman I</del> <b>Serviceperson I</b>	<del>33.06</del>	<del>33.52</del>	33.97
	Plus 1% 2020	<del>33.39</del>	<del>33.86</del>	<del>34.31</del>
2022	ml   40/ mom4			

<del>33.72</del>

<del>34.20</del>

34.65

3023

Plus 1% 2021

	Plus 1% 2022	34.06	34.54	35.00
	Plus 1.5% 2023	34.57	35.06	35.53
	Vehicle Serviceman II Serviceperson II	36.18	<del>36.92</del>	<del>37.61</del>
	Plus 1% 2020	<del>36.5</del> 4	<del>37.29</del>	<del>37.99</del>
	Plus 1% 2021	<del>36.91</del>	<del>37.66</del>	<del>38.37</del>
	Plus 1% 2022	37.28	38.04	38.75
3024	Plus 1.5% 2023	37.84	38.61	39.33

	Shipper-Receiver, Materials Management	<del>20.78</del>	<del>24.18</del>	<del>26.90</del>
	Plus 1% 2020	20.99	24.42	<del>27.17</del>
	Plus 1% 2021	21.20	<del>24.66</del>	<del>27.44</del>
	Plus 1% 2022	21.41	24.91	27.71
TBD	Plus 1.5% 2023	21.73	25.28	28.13
	Partsman   Parts Person	31.78	<del>32.61</del>	<del>33.25</del>
	Plus 1% 2020	<del>32.10</del>	<del>32.9</del> 4	<del>33.58</del>
	Plus 1% 2021	<del>32.42</del>	<del>33.27</del>	<del>33.92</del>
	Plus 1% 2022	32.74	33.60	34.26
3025	Plus 1.5% 2023	33.23	34.10	34.77
	Partsman II Parts Person II	34.14	<del>35.08</del>	<del>35.95</del>
	Plus 1% 2020	34.48	<del>35.43</del>	<del>36.31</del>
	Plus 1% 2021	<del>34.82</del>	<del>35.78</del>	<del>36.67</del>
	Plus 1% 2022	35.17	36.14	37.04
3026	Plus 1.5% 2023	35.70	36.68	37.60
	Welder	42.04	43.58	4 <del>5.22</del>
	Plus 1% 2020	42.46	44. <del>02</del>	4 <del>5.67</del>
	Plus 1% 2021	42.88	<del>44.46</del>	4 <del>6.13</del>
	Plus 1% 2022	43.31	44.90	46.59
3027	Plus 1.5% 2023	43.96	45.57	47.29
	Small Engine Mechanic	<del>30.75</del>	<del>31.59</del>	<del>32.44</del>
	Plus 1% 2020	<del>31.06</del>	<del>31.91</del>	<del>32.76</del>
	Plus 1% 2021	31.37	<del>32.23</del>	<del>33.09</del>
	Plus 1% 2022	31.68	32.55	33.42
3028	Plus 1.5% 2023	32.16	33.04	33.92
	V&E Leadhand	<del>47.02</del>	<del>48.42</del>	
	Plus 1% 2020	<del>47.49</del>	4 <del>8.90</del>	
	Plus 1% 2021	<del>47.96</del>	4 <del>9.39</del>	
	Plus 1% 2022	48.44	49.88	
3040	Plus 1.5% 2023	49.17	50.63	
	V&E Chargehand	45.91	<del>46.34</del>	<del>48.42</del>
	Plus 1% 2020	46.37	4 <del>6.80</del>	4 <del>8.90</del>
	Plus 1% 2021	46.83	4 <del>7.27</del>	49.40
	Plus 1% 2022	47.29	47.74	49.89
3030	Plus 1.5% 2023	48.00	48.46	50.64

	Transit Leadhand	47.02	48.42	
		4 <del>7.02</del>		
	Plus 1% 2020	4 <del>7.49</del>	4 <del>8.90</del>	
	Plus 1% 2021	4 <del>7.96</del>	49.39	
2042	Plus 1% 2022	48.44	49.88	
3042	Plus 1.5% 2023	49.17	50.63	4= 00
	Fleet Services Technician	42.04	43.58	45.22
	Plus 1% 2020	42.46	44.02	45.67
	Plus 1% 2021	42.88	44.46	46.13
	Plus 1% 2022	43.31	44.90	46.59
3062	Plus 1.5% 2023	43.96	45.57	47.29
2042	111177	27.05	20.67	20.00
3043	Utility Operator I	<del>27.95</del>	<del>28.67</del>	<del>29.08</del>
3044	Utility Operator II	31.47	32.20	32.84
3045	Utility Operator III	33.62	<del>35.72</del>	<del>37.81</del>
	Construction Operator	<del>29.14</del>	<del>32.20</del>	<del>35.72</del>
	Plus 1% 2020	<del>29.43</del>	<del>32.52</del>	<del>36.07</del>
	Plus 1% 2021	<del>29.72</del>	<del>32.85</del>	<del>36.43</del>
	Plus 1% 2022	30.02	33.18	36.79
TBD	Plus 1.5% 2023	30.47	33.68	37.34
		<del>37.81</del>		
	Plus 1% 2020	<del>38.19</del>		
	Plus 1% 2021	<del>38.57</del>		
	Plus 1% 2022	38.96		
	Plus 1.5% 2023	39.54		
	Utility Leadhand	<del>37.13</del>	<del>39.25</del>	<del>41.42</del>
	Plus 1% 2020	<del>37.50</del>	<del>39.64</del>	41.83
	Plus 1% 2021	<del>37.88</del>	40.04	4 <del>2.25</del>
	Plus 1% 2022	38.26	40.44	42.67
3046	Plus 1.5% 2023	38.83	41.05	43.31
	Infrastructure Foreman Foreperson	41.70	43.31	44.90
	Plus 1% 2020	42.12	43.74	4 <del>5.35</del>
	Plus 1% 2021	42.54	44.18	4 <del>5.80</del>
	Plus 1% 2022	42.97	44.62	46.26
3047	Plus 1.5% 2023	43.61	45.29	46.95
	Utility Labourer	20.78	24.18	<del>26.90</del>
	Plus 1% 2020	20.99	24.42	<del>27.17</del>
	Plus 1% 2021	<del>21.20</del>	24.66	<del>27.44</del>
	Plus 1% 2022	21.41	24.91	27.71
3048	Plus 1.5% 2023	21.73	25.28	28.13
	Traffic Support Aid			<del>32.84</del>
	Plus 1% 2020			<del>33.17</del>
3050	Plus 1% 2021			<del>33.50</del>

***************************************	Plus 1% 2022			33.84
	Plus 1.5% 2023			34.35
	Utilities Instrument Technician/Electrician			
	Dual	46.30	<del>48.05</del>	<del>49.66</del>
	Plus 1% 2020	4 <del>6.73</del>	4 <del>8.53</del>	<del>50.16</del>
	Plus 1% 2021	4 <del>7.20</del>	4 <del>9.02</del>	<del>50.66</del>
	Plus 1% 2022	47.67	49.51	51.17
3051	Plus 1.5% 2023	48.39	50.25	51.94
	Utilities Electrician	44.11	45.78	47.28
	Plus 1% 2020	44.55	46.24	4 <del>7.75</del>
	Plus 1% 2021	44.00	4 <del>6.70</del>	48.23
	Plus 1% 2022	45.45	47.17	48.71
3052	Plus 1.5% 2023	46.13	47.88	49.44
	Utilities Instrument Technician	44.11	4 <del>5.78</del>	47.28
	Plus 1% 2020	44.55	46.24	47.75
	Plus 1% 2021	44.00	4 <del>6.70</del>	48.23
	Plus 1% 2022	45.45	47.17	48.71
3053	Plus 1.5% 2023	46.13	47.88	49.44
	Control System Technician	44.11	45.78	47.28
	Plus 1% 2020	44.55	4 <del>6.2</del> 4	4 <del>7.75</del>
	Plus 1% 2021	44.00	4 <del>6.70</del>	48.23
	Plus 1% 2022	45.45	47.17	48.71
3063	Plus 1.5% 2023	46.13	47.88	49.44

	Apprentice Auto Mechanic, 1st year (55%)	<del>24.87</del>		
	Plus 1% 2020	<del>25.12</del>		
	Plus 1% 2021	<del>25.37</del>		
	Plus 1% 2022	25.62		
3101	Plus 1.5% 2023	26.00		
	Apprentice Auto Mechanic, 2nd year (70%)	31.66		
	Plus 1% 2020	<del>31.98</del>		
	Plus 1% 2021	<del>32.30</del>		
	Plus 1% 2022	32.62		
3102	Plus 1.5% 2023	33.11		
	Apprentice Auto Mechanic, 3rd year (80%)	<del>36.17</del>	1000	
	Plus 1% 2020	<del>36.53</del>		
	Plus 1% 2021	<del>36.90</del>		
	Plus 1% 2022	37.27		
3103	Plus 1.5% 2023	37.83		
	Apprentice Auto Mechanic, 4th year (90%)	40.69		
	Plus 1% 2020	41.10		
	Plus 1% 2021	41.51		
3104	Plus 1% 2022	41.93		

	Plus 1.5% 2023	42.56		
	A service Heavy Date Marketing 1 days			
	Apprentice Heavy Duty Mechanic, 1st year	27.42		
	(60%)	<del>27.13</del>		
	Plus 1% 2020	<del>27.40</del>		
	Plus 1% 2021 Plus 1% 2022	<del>27.67</del> 27.95		
3111	Plus 1.5% 2023	27.95		
2111		20.37		
	Apprentice Heavy Duty Mechanic, 2nd year (70%)	<del>31.66</del>		
	Plus 1% 2020	31.98		
	Plus 1% 2021	32.30		
	Plus 1% 2021	32.62		
3112	Plus 1.5% 2023	33.11		
3112	Apprentice Heavy Duty Mechanic, 3rd year	33.11		
	(80%)	<del>36.17</del>		
	Plus 1% 2020	<del>36.53</del>		
	Plus 1% 2021	<del>36.90</del>		
	Plus 1% 2022	37.27		
3113	Plus 1.5% 2023	37.83		100
3113	Apprentice Heavy Duty Mechanic, 4th year	57.05		
	(90%)	<del>40.69</del>		
	Plus 1% 2020	41.10		
	Plus 1% 2021	41.51		
	Plus 1% 2022	41.93		
3114	Plus 1.5% 2023	42.56		
	Certified Technician (HD)	<del>42.04</del>	43.58	45.22
	Plus 1% 2020	4 <del>2.46</del>	44.02	4 <del>5.67</del>
	Plus 1% 2021	4 <del>3.07</del>	44.46	46.13
	Plus 1% 2022	43.50	44.90	46.59
3029	Plus 1.5% 2023	44.15	45.57	47.29
	A	27.42		
	Apprentice Transit Mechanic, 1st year (60%)	<del>27.13</del>		
	Plus 1% 2020	<del>27.40</del>		
	Plus 1% 2021	<del>27.67</del>		
2424	Plus 1% 2022	27.95		
3131	Plus 1.5% 2023	28.37		
	Apprentice Transit Mechanic, 2nd year (70%)	<del>31.66</del>		
	Plus 1% 2020	<del>31.98</del>		
	Plus 1% 2021	<del>32.30</del>		
2422	Plus 1% 2022	32.62		
3132	Plus 1.5% 2023	33.11		

	Apprentice Transit Mechanic, 3rd year (80%)	<del>36.17</del>	
	Plus 1% 2020	<del>36.53</del>	
	Plus 1% 2021	<del>36.90</del>	
	Plus 1% 2022	37.27	
3133	Plus 1.5% 2023	37.83	
	Apprentice Transit Mechanic, 4th year (90%)	4 <del>0.69</del>	
	Plus 1% 2020	41.10	
	Plus 1% 2021	41.51	
	Plus 1% 2022	41.93	
3134	Plus 1.5% 2023	42.56	

Hourly rates for apprentice technicians are a percentage of step 3 of Fleet Services Technician.

# **SCHEDULE B**

EQUIPMENT	MINIMUM OF CLASSIFICATION RATE PAID
Non-Finishing Packers (Rubber Roller, Pad Foot Operator, and combination for non- finishing work)	Equipment Operator I – Step 1
Finishing Packer (Smooth Drum Rollers and combination of finishing work)	Equipment Operator I
Skid Steer / Sidewalk Machine	Equipment Operator II
Single Axle Truck	Truck Driver I Equipment Operator I
Sign Truck (Single Axle with Manlift)	Truck Driver II Equipment Operator I
<del>Tracked Dozer</del>	Equipment Operator II
Wheel Loader	Equipment Operator II
Tandem Axle Truck	Truck Driver II Equipment Operator II
Street Sweepers	Equipment Operator II
Truck Plow	Equipment Operator II
Sander	Truck Driver II Equipment Operator II
RPM Snow Blower	Equipment Operator II
Road Reclaimer	Equipment Operator II

PB Patcher Patch Truck	Equipment Operator II
Rubber Tire Backhoe	Equipment Operator II
Graders	Operator II Equipment Operator III
Track Hoe	Equipment Operator III + \$0.20 / hour
Paver	Equipment Operator III
Tractor-Trailer	Truck Driver ## Class 1
Directional Driller	Utility Operator III

# **Letter of Understanding**

#### **Between**

# **Strathcona County**

and

The Alberta Union of Provincial Employees, Local 118, Chapter 009

# Subject: New Grid Placement and Recognition of Operator II hours

- 1. Employees with hours worked gained at the Operator II classification prior to ratification, including such hours worked gained on the Skid Steer/Sidewalk Machine, shall continue to have such hours recognized for the purposes of:
  - a. Determining step levels on the new wage grid for Operator I and II classifications;
  - b. For those employed at date of ratification, the skid steer/sidewalk machines will qualify as one of the pieces of equipment when applying for an Equipment Operator II position, for the term of this agreement.

This letter of understanding expires on December 31, 2023.

From: Chantel Kassongo

**Sent:** Tuesday, August 09, 2022 10:12 AM

To: Merryn Edwards
Cc: Sheron M. Ward

Subject: RE: Strathcona County & AUPE

Hi Merryn:

Thanks for your patience. See attached proposal for settlement.

I have highlighted in yellow what I have changed for your ease of reference.

In summary, our counter includes the following:

- Cola:
  - $\circ$  2020 0.5% on grid Note this is the same as the classified employees.
  - o 2021 1% lump sum
  - o 2022 1% on grid
  - o 2023 1.5% on grid
- Increase DRP to a Flat rate of \$2.25 and will apply to Holidays
- Introduction of an LoU dealing with the Operator II hours on the Skid Steer

You will see that we have proposed a new LoU that in effect "grandparents" current employees with a class 3 license and give time for those to obtain their class 3 if they do not have it, so that they will continue to be Pid at the OP II rates.

We did not accept your proposal to change the pay rates for the Equipment Operator as we feel that the proposed structure is fundamentally the right structure and we because we introduced a 2-year time frame on the "grandparenting".

In short, we feel that the with the increase to the Grid, DRP and grandparenting current employees with a class 3 licences meets you more than halfway. I will also point out that with respect to the grandparenting there are only 5 employees that this would impact, and we are firm in our view that we do not want the inequity to continue indefinitely between those employees and anyone else doing the same work on the same piece of equipment.

You will also see that the retroactive pay date is somewhat far out. As we discussed at the table, HR is implementing a whole new HR/pay system this fall and early winter and will not be in a position to process all of these payments until the Spring. Had it ratified earlier, we could have processed the payments before the implementation began.

Р	lease	le:	t me	know	ıt١	you	have	any	questions	

Thanks.

Chantel



## Chantel Kassongo, Q.C. Barrister & Solicitor

#301, 550 – 91 Street SW, Edmonton, AB T6X 0V1 **D:** 780-733-6563 | C: 780-938-8040 | **F:** 780-488-0026

Email | Website | LinkedIn | Employment Law Alliance



This email and any attachments may contain privileged and confidential information. Review, use or retention of this information by someone other than the recipient is prohibited. If you have received this message in error, please contact the sender and delete all references to the email from your system.

From: Merryn Edwards < m.edwards@aupe.org>

**Sent:** Wednesday, May 18, 2022 9:23 AM **To:** Chantel Kassongo <ckassongo@ntlaw.ca>

Cc: Sharla Mishna <Sharla.Mishna@strathcona.ca>; Sharry Sowiak <Sharry.Sowiak@strathcona.ca>; Sheron M. Ward

<sward@ntlaw.ca>

Subject: Re: Strathcona County & AUPE

Hi Chantel,

Thanks for your patience while we have consulted with our membership.

The main concerns we heard were:

- The sense that the inclusion of seasonals in benefits at the same time as reducing pay due to the downgrading of the sidewalk equipment (and reduction in callout pay for utilities) was an unacceptable "give with one hand/take with the other" approach
- The insistence that wage increases be on the grid, not as lump sums
- Ongoing frustrations with scheduling for TAS staff over Christmas

The results of that discussion have culminated in the attached proposal, which includes the following changes from the previous TA:

- 1% on the grid instead of lump sums for 2020 and 2021 (maintained 1% for 2022 and 1.5% for 2023)
- Sidewalk machine/skid steer maintained at Equipment Operator II
  - o LOU re Recognition of Operator II hours on sidewalk/skid steer not necessary
- Rates for Equipment Operator I and new merged Labourer classification reduced from TA
  - Labourer steps correspond to existing LI step 1, LII step 2, and LIII step 3 (vs. LI step 2, LII step 2, LIII step 3)
  - o Equipment Operator I steps same as current Operator I steps (vs. increased steps 2 and 3)
- Removed deletion of "five (5) consecutive days" from Christmas Day/Christmas floater/Boxing Day and weekend (23.07)

We are available June 27 and/or 28 to reconvene in bargaining.

Thank you, Merryn

Merryn Edwards

Negotiations

Alberta Union of Provincial Employees

phone: 780-930-5287 (temporarily unavailable)

cell: 780-952-1951 fax: 780-930-3392

toll-free: 1-800-232-7284 email: m.edwards@aupe.org

El respeto al derecho ajeno es la paz. Respect for the rights of others is peace. --Benito Juarez

From: Chantel Kassongo < ckassongo@ntlaw.ca>

Sent: May 16, 2022 12:22 PM

To: Merryn Edwards < m.edwards@aupe.org >

Cc: Sharla Mishna <Sharla.Mishna@strathcona.ca>; Sharry Sowiak <Sharry.Sowiak@strathcona.ca>; Sheron M. Ward

<sward@ntlaw.ca>

Subject: Strathcona County & AUPE

CAUTION: This email has been sent from an external source. Treat links and attachments in this email with care.

Hi Merryn:

Four months have passed since you advised that the TA did not ratify.

The County would like to schedule dates to bargain and/or at the very least discuss next steps.

Please let us know of your availability over the next couple of months.

Thanks.

Chantel



## Chantel Kassongo Barrister & Solicitor

#301, 550 – 91 Street SW, Edmonton, AB T6X 0V1 **D:** 780-733-6563 | C: 780-938-8040 | **F:** 780-488-

Email | Website | LinkedIn | Employment Law Alliance



This email and any attachments may contain privileged and confidential information. Review, use or retention of this information by someone other than the recipient is prohibited. If you have received this message in error, please contact the sender and delete all references to the email from your system.

As the CMO has implemented mandatory restrictions, we ask that all members and staff comply with those restrictions. It is expected that members continue to hold all meetings virtually. Meeting space at AUPE offices will not be available as the offices will remain closed until further notice.

#### Disclaimer

The information contained in this communication from the sender is confidential. It is intended solely for use by the recipient and others authorized to receive it. If you are not the recipient, you are hereby notified that any disclosure, copying, distribution or taking action in relation of the contents of this information is strictly prohibited and may be unlawful.

# PROPOSED MEMORANDUM OF SETTLEMENT BETWEEN

## ALBERTA UNION OF PROVINCIAL EMPLOYEES

AND

# STRATHCONA COUNTY August 9, 2022

The attached documents constitute a full settlement of all issues and proposals remaining outstanding in the current round of negotiations. This proposed Memorandum of Settlement combined with the previously signed off Articles, and Letters of Understanding will form a Tentative Agreement between the parties, which will be subject to the normal ratification process of each party.

The Parties agree to recommend to their respective principals the terms and conditions contained in the Tentative Agreement.

Signed at Sherwood Park, this	day of August 9, 2022.
FOR STRATHCONA COUNTY	FOR AUPE LOCAL 118
Slisha	
Sharla Mishna	

THIS AGREEMENT is made this 18th day of April, 2018, A.D. pursuant to the Labour Relations Code of Alberta (2000) as amended.

This Agreement is entered into by and between the Strathcona County, municipal corporation covered by the Collective Agreement (hereinafter referred to as "the County" or "the Employer") and the Alberta Union of Provincial Employees Local 118/009 (hereinafter referred to as "the Union").

Whereas, the above parties are mutually desirous of entering into a Collective Agreement with the intent and purpose of:

- A. Promoting a harmonious relationship;
- B. Providing for conditions of employment, hours of work and rates of pay;
- C. Establishing the means for prompt and equitable settlement of disputes or grievances between the County and Employees represented by the Union, and;
- D. Encouraging efficiency in operation.

Now therefore, the parties hereto mutually agree as follows:

#### **ARTICLE 1 – DEFINITIONS**

- 1.01 In this Agreement:
  - (a) A word used in the singular applies also in the plural. A word used in the masculine applies also in the feminine.
  - (b) Employee: means a person employed by the Strathcona County in the Public Works Operations, Vehicle and Equipment Services, Transit Services, Materials Management, Water and Wastewater Operations, and as listed in Schedule A.
    - (i) Regular Employees: are Employees who occupy established positions and who have successfully completed the probationary period and work the fully scheduled hours.
    - (ii) Part-time Employees: are Employees who occupy established positions and have successfully completed the probationary period and work less than the fully scheduled hours of Regular Employees.
    - (iii) Temporary Employees: are Employees who are indefinite term Employees or Employees for specific projects, seasonal employment, replacements for Regular Employees who are on illness leave, General Illness, Long Term Disability, Workers' Compensation, Maternity Leave, Parental Leave and Adoption Leave, or other long term leaves of absence. Indefinite term Employees upon completion of more than fifteen (15) months of continuous service will be deemed to be Regular Employees,

except if they are covering for an Employee on Long Term Disability, are moving from one Department to another Department for a temporary or seasonal position, or are covering off consecutive leaves, as referenced above. "Continuous service", for the purpose of this clause, means any period of employment that does not include a break in service of more than thirty (30) calendar days.

- (iv) Apprenticed Employees: are Employees who are registered as Apprentices under the Apprenticeship and Industry Training Act.
- (v) Probationary Employee: means an Employee who during his initial period of employment occupies an established position during a probationary period.
- (c) Grade: means the step assigned to a class within the pay grid.
- (d) Union Steward: means a person who is selected by the members of the Union to act on behalf of those Employees as empowered by the Union.
- (e) Union Representative: means a person authorized by the Union to act on behalf of Employees and includes a Union Steward, elected or appointed officer of the Union, or staff representative.
- (f) Liaison Committee: is the joint Liaison Committee established by the two parties as stipulated within this Agreement.
- (f) Casual Illness: is any illness which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less.
- (g) General Illness: is any illness which causes an Employee to be absent from work for a period of more than three (3) consecutive work days.
- (h) Layoff: includes a reduction in the work force as a result of shortage of work.
- (i) Branch: means
  - (i) Public Works Operations
  - (ii) Vehicle and Equipment Services
  - (iii) Transit Services
  - (iv) Water and Wastewater Services
  - (v) Materials Management
- (j) Classification: means each line in Schedule A.
- (k) Seniority: commences at date of hire, and shall include all continuous **and accumulated service** as a temporary Employee. <del>provided there was no break in service greater than thirty (30) days.</del>

3

(l) Day: shall mean calendar day.

- (m) Continuous Service: any period of employment that does not include a break in service of more than thirty (30) calendar days.
- (n) Accumulated Service: Total number of regular hours worked, with no break in service of more than six (6) months.

#### ARTICLE 2 – JURISDICTION

- 2.01 During its term, this Agreement shall be applicable to any person employed in the Public Works Operations, Vehicle and Equipment Services, Transit Services, Materials Management and Water and Wastewater Services of Strathcona County, who are described in Schedule A attached hereto, or to positions that fall within the jurisdiction of the Union.
- 2.02 When additional work is required by the Employer, individuals who have paid Union dues to the Alberta Union of Provincial Employees, Local 118, Chapter 009 in the past twelve (12) months, will be called first and, if available, offered the additional work.

#### **ARTICLE 4 - APPLICATION**

- 4.01 Regular Employees shall be granted all the terms and conditions of this Agreement.
- 4.02 Part-time Employees shall be granted all the terms and conditions of this Agreement except those conditions which require a particular length of service, such benefits shall be granted on a pro-rata basis in relation to hours of work of full-time employment.
- 4.03 Apprenticed Employees as defined within Apprenticeship and Industry Training Act shall be granted all the terms and conditions of this Agreement.
- 4.04 Temporary Employees shall be entitled to the terms and conditions of this Agreement, except where Regular Employees are referenced. Seniority for employment periods of less than six (6) months shall not be accumulated.

#### **ARTICLE 6 - UNION RECOGNITION**

- 6.01 The County recognizes the Union as the sole bargaining agent of the Employees covered by this Agreement.
- No Union Employee shall be required or permitted to make any written or verbal agreement which may be in conflict with the terms of this Collective Agreement.

## 6.03 New Employees

The Employer shall supply all new Employees with a copy of the Collective Agreement.

## 6.04 <u>Monthly Staffing Reports</u>

The County recognizes the need of the Union to have required information concerning Employees eligible for membership in the Union and hereby agrees to submit between the sixteenth (16th) day and the last day of each month, the name, address, and position filled by any new Employees assigned to Public Works Operations, Vehicle and Equipment Services, Materials Management, Water and Wastewater Operations and eligible for membership in the Union.

The Employer recognizes the importance of new Union Employees receiving an orientation to the Union. To ensure this occurs, the Chapter Executive will be given a list of all newly hired Employees and a representative of the Union shall, within thirty (30) calendar days of the Employees' date of hire, make a presentation to the Employee(s) with respect to the structure of the Union as well as the rights, responsibilities and benefits of membership in the Union.

The Employer will generally, provide the Union with one (1) weeks' advance notice of the Employee's scheduled orientation.

## 6.04 No Discrimination

The parties agree that there shall be no discrimination or coercion exercised or practiced with respect to any Employee for reason of function, membership or legitimate activity in the Union.

The Employer shall exercise its rights in a fair and reasonable manner, nor shall these rights be used in a manner which would deprive any Employee of his employment, except for just cause.

#### 6.05 **Bulletin Boards**

The County recognizes the right and need of the Union to be able to post notices of meetings and other legitimate Union activities at various sites wherein members of the Union are assigned or employed.

## 6.06 <u>Seniority Lists</u>

County shall maintain a list showing the date upon which each Employee's service commenced. The County shall forthwith provide the Union, upon request, with an up-to-date list and shall, in any event, post an up-to-date list on all bulletin boards where the Employees work on July 1<sup>st</sup> and January 1<sup>st</sup> in each year.

#### 6.07 Union Officers Stewards

The County recognizes the right of the Union to appoint "Employees" as Union Stewards who are to be recognized as official representatives of the Union.

#### 6.08 <u>Union Membership</u>

Each person who becomes employed in positions covered by this Agreement shall automatically become a member of the Union as a condition of employment.

### 6.09 Monthly Staffing Reports

- (a) The Employer agrees to remit to the central office of the Union, the amounts equal to the dues that have been deducted from the pay of all Employees by the first working day after the fifteenth (15th) calendar day in the following month. Where an accounting adjustment is necessary to correct an over or under payment of dues, it shall be effected in the succeeding month.
- (b) The County recognizes the need of the Union to have required information concerning Employees eligible for membership in the Union and hereby agrees to submit on a monthly basis between the sixteenth (16th) day and the last day of each month, the following information: the name, address, phone number, status (regular or temporary), Identification number, classification, starting date and position filled by any new Employees In the bargaining unit. assigned to Public Works Operations, Vehicle and Equipment Services, Materials Management, Water and Wastewater Operations and eligible for membership in the Union.

The Employer shall provide the Union with a monthly list identifying each Employee, including new or terminated Employees. The list will include Employee name, address, identification number, starting date, classification, and dues deducted for that calendar year.

(c) The Employer recognizes the importance of new Union Employees receiving an orientation to the Union. To ensure this occurs, the Chapter Executive will be given a list of all newly hired Employees and a representative of the Union shall, within thirty (30) calendar days of the Employees' date of hire, make a presentation to the Employee(s) with respect to the structure of the Union as well as the rights, responsibilities and benefits of membership in the Union.

### ARTICLE 8 - JOB DESCRIPTIONS CLASSIFICATIONS & RECLASSIFICATIONS

The Employer shall make the Employee's job description available.

#### 8.02 New or Amended Classifications

The County Administration shall give written notice to the Union of the establishment of any new or amended job descriptions and/or salary classification levels (Schedule A). If it is established that a higher salary level is appropriate, the new salary shall be retroactive to the date of the classification was created. Where Union and management representatives through the Liaison Committee are unable to agree within thirty (30) days on either the correct classification of a revised job description or on the compensation level of a new job description, the matter may be advanced to arbitration for the establishment of the compensation for the individuals for which the job description has changed. If a wage change results, the effective date of change shall be determined by the Arbitrator. Both the Employer and the Union will share the cost of the Arbitrator equally. The Arbitrator

will be guided by:

- (a) Evidence and submissions of the parties;
- (b) The salary levels in the Collective Agreement;
- (c) The complexity, working conditions and knowledge requirements of the other positions in the Collective Agreement at the time that the new position is established.

However, the Arbitrator shall not be restricted to the existing salary levels.

#### 8.02 Classification Reviews

An Employee may at any time bring to management for review, changes that may have occurred to his existing job duties. The Employee/Union, may bring such issues forward to Liaison Committee.

## 8.03 Reclassification

When it is necessary to re-classify Regular Employees, the County may re-classify the affected Employees on the basis of Branch seniority, qualifications and ability.

- (a) An Employee to be re-classified may displace any Employee in a lower classification within that Branch that has less Branch seniority providing that the Employee being reclassified is qualified to perform the duties of that position;
- (b) Employees with special qualifications may be exempt from re-classification.
- 8.04-12.02 In those cases where re-classification according to Clause 12.01 8.03 have occurred and vacancies become available in the affected job classification, those Employees who were previously re-classified downward from that classification, shall be returned to that classification on the basis of their seniority so long as that Employee has maintained the ability and qualifications to be appointed to the position.
- 8.05 12.03 An apprentice who begins and completes their apprenticeship with the Employer shall be placed on level 2 of the appropriate trades rate upon confirmation of certification.

### ARTICLE 9 - POSTING OF POSITIONS, PROMOTIONS AND TRANSFERS

- 9.01 When a position of a Regular or Temporary nature is to be filled, notice of the vacancy shall be posted as soon as reasonably possible for a minimum of one (1) week. The position can be filled by appointment if it is to be filled for a period of less than ninety (90) calendar days.
- 9.02 If a position with the same job code becomes vacant within six (6) months of the original closing date, the Employer may consider candidates from the original

recruitment process, without re-posting. This is not applicable if the original posting was for a temporary position and the vacancy is for a permanent position.

- 9.02 3 Such notice will contain the following information: date of the posting, nature of position, qualifications, required knowledge and education, skills, shift, hours of work, experience, wage rate or range, consistent with the job description.
- 9.03 4 Where there are qualified Union candidates for a posted position, their applications shall be considered first.
- 9.04-5 Where qualified applicants have applied and a suitable candidate has been selected, the Employer shall appoint a candidate to the position within thirty (30) days of the competition closure.
- 9.05 6 Preference in the filling of positions shall be given to Employees of the County over all other persons, where the Employee has the required skills and ability to perform the tasks of the position. When filling positions or making promotions, qualifications, experience, skills, ability and seniority shall be the governing factors. When filling positions, employees of the County shall be given first preference.

#### 9.06 7 Trial Period

An Employee who is transferred or promoted to another position within this Agreement shall be confirmed in that other position within three (3) months unless the Employee wishes otherwise or proves unsatisfactory in the new position, in which case the Employee will be returned to his former position. Any other Employee promoted or transferred because of the displacement of positions shall also be returned to his former position. In any event, Employees promoted or transferred shall be paid at the rate applicable to the new position from commencement of employment in the position until returned to the previous position.

- (b) In making promotions and filling vacancies, appointments will be made on the basis of qualifications, experience, ability and seniority.
- 9.08 The posting requirements in this Article may be waived by mutual agreement in cases of medical accommodation.

#### ARTICLE 10 - SENIORITY

10.01 For the purpose of Clauses 10.01 to 10.03 10.10, 12.01 to 12.03 12.02, and 15.01 to 15.06 15.05, seniority shall be defined from the date of continuous service employment in any Branch covered by this Agreement.

In accordance with 1.01(k) seniority commences at date of hire, and shall include all continuous and accumulated service as a temporary Employee.

10.02(a) Employees who were employed by the County prior to January 1, 1979 and entered in any classification covered by this Agreement on that date shall have their

seniority established as the date of continuous **service** <del>employment</del> with the County.

- 10.023(b)—Seniority shall be retained and accumulated during absences resulting from illness or an accident where the Employee is entitled to compensation, suspension with or without pay less than sixty-one (61) days, approved leave without pay less than sixty-one (61) days.
- 10.034 (c)—Seniority shall be retained during absences resulting from; layoff (up to eighteen (18) months), suspension with or without pay greater than sixty (60) days, approved leave without pay greater than sixty (60) days, strike or lockout.
- 10.045 Where Employees have been laid off and recalled, they shall retain the seniority held at the date of layoff. The Union shall be notified of all recalls.
- 10.056(d) An Employee who accepts an internal permanent position outside this Agreement shall retain the seniority accrued to that time for a period of six (6) months.
- 40.0210.06 Accumulation of seniority shall recommence from the date the Employee returns to a classification covered by this Agreement and their seniority date shall be adjusted by the period of their absence. This retained seniority shall not be effective in exercising any seniority rights until the Employee has been back in the scope of this Agreement for six (6) months.
- An Employee shall lose their seniority and their name shall be removed from the seniority list for any of the following reasons: voluntary resignation, discharge for cause, layoff for more than eighteen (18) months, retirement, or when an Employee has been employed in a position outside this Agreement for a period of six (6) consecutive months or more.

#### **ARTICLE 11 - PERIOD OF PROBATION**

11.01 The probationary period for new Employees covered by this agreement in both permanent and temporary positions shall be five hundred (500) hours one thousand (1000) hours worked. , with the exception of Employees appointed to the positions listed below, whose probationary period will be one thousand (1000) hours worked:

3008	Public Works Technician	<del>3015</del>	<del>Operator I</del>
3009	Public Works Technician II	<del>3016</del>	<del>Operator II</del>
3012	Truck Driver I	<del>3017</del>	Operator III
3013	Truck Driver II	<del>3019</del>	Operations Foreman
3014	Truck Driver III		

3025	<del>Partsman I</del>	3030	<del>V&amp;E Chargehand</del>
------	-----------------------	------	-------------------------------

<del>3026</del>	<del>Partsman II</del>	3040	<del>V&amp;E Leadhand</del>
3027	Welder	<del>3041</del>	Transit Chargehand
3062	Fleet Services Technician	<del>3042</del>	Transit Leadhand

3021	<del>W&amp;WW Foreman</del>	3052	Utilities Electrician	
3038	W&WW Leadhand	3053	Utilities Instrument Technician	
3047	7 Infrastructure Foreman		Ut.Inst Tech/Electrician Dual	
<del>3039</del>	<del>Project Foreman</del>	3063	Control Systems Technician	
3043	Utility Operator I	3020	W&WW Operator (Collection and Distribution)	
3044	Utility Operator II	<del>3049</del>	W&WW Operator (Transmission)	
3045	Utility Operator III	3037	Wastewater Infrastructure Operator	
3046	Utility Leadhand			

If the Employer identifies performance concerns, written feedback shall be provided to the Employee. Where circumstances warrant an extension, the probationary period may be extended up to an additional two hundred and fifty hours (250) hours upon mutual agreement of the Union and the Employer. may be extended up to an additional 500 hours worked, at the Employer's discretion. The Employee and the Union shall be notified of the extension and the cause for the extension.

- An Employee who has previously been employed in a position covered by this agreement will have such previous employment considered part of the probationary period provided that a break in service is one (1) year or less and the Employee is returning to the same job.
- 11.03 At any time, an Employee, categorized as probationary, may be released by the Department Head or the County Administration. Such notice of release will be formalized in writing and will give the reason for the Employee being released as soon as practicable following at the time of the notification of the release.
- 11.04 Employees shall provide at least ten (10) working days written notice of their intention to resign.
- In the event that an Employee terminates his employment with the County without submitting the written notice in compliance with Clause 11.04, the Employee shall not be entitled to receive his regular cheque and/or his final cheque until the next regular distribution of regular wage and salary cheques.

11.06 4 A Probationary Employee in accordance with Articles 11.01 shall be granted all of the terms and conditions of this Agreement. During the probationary period the County may terminate the employee. The employee or the Union may grieve the termination, but the decision at Level 3 shall be final and binding and the matter shall not be referred to arbitration under the provisions of this Agreement.

however the termination of a Probationary Employee shall not be have access to the application of any grievance procedure for termination of employment as a result of:

- (a) Unsatisfactory probationary performance; or
- (b) The abolishment of the position occupied by the Probationary Employee

#### ARTICLE 12 - RECLASSIFICATION

- 12.01 When it is necessary to re-classify Regular Employees, the County may re-classify the affected Employees on the basis of Branch seniority, qualifications and ability.
  - (a) An Employee to be re-classified may displace any Employee in a lower classification within that Branch that has less Branch seniority providing that the Employee being reclassified is qualified to perform the duties of that position;
  - (b) Employees with special qualifications may be exempt from re-classification.
- In those cases where re-classification according to Clause 12.01 have occurred and vacancies become available in the affected job classification, those Employees who were previously re-classified downward from that classification, shall be returned to that classification on the basis of their seniority so long as that Employee has maintained the ability and qualifications to be appointed to the position.
- 12.03 An apprentice who begins and completes their apprenticeship with the Employer shall be placed on level 2 of the appropriate trades rate upon confirmation of certification.

## **ARTICLE 14 - GRIEVANCES**

#### 14.01 Statement of Intent

The Employer and the Union recognize the desirability of resolving differences through joint consultation and **respectful and open** discussion. Both will attempt to resolve differences through **exploring interests and options together informally** means where possible, prior to proceeding to formal **process**.

#### 14.02 Definitions and Scope

- (a) A Personal grievance is a dispute regarding:
  - (i) Alleged unjust treatment or discrimination
  - (ii) Alleged unfair working conditions
  - (iii) Any disciplinary action
  - (iv) Any concern regarding the interpretation, application, operation or alleged violation of this Agreement, including any questions as to the arbitral nature of the matter.
- (b) Grievances which involve a dismissal or a suspension will commence at level II and shall be initiated in writing within fourteen (14) days of the date when the Employee first became aware of the disciplinary action.
- (c) A Policy Grievance is a dispute between the parties which due to its nature is not properly subject of a Personal Grievance and concerns the interpretation, application, operation or alleged violation of this Agreement, including any questions as to the arbitral nature of the matter.
- (d) Formal personal grievances shall be initiated within fourteen (14) calendar days from the date of the incident prompting the grievance. Policy grievances shall be initiated either by the Union or the County in writing, within thirty (30) calendar days from the day that the matter giving rise to the grievance came to the attention of the Designated Officer of the Union or Official of the Employer.
- (d) Formal grievances shall be in writing (email will suffice) and must state the general nature of the grievance, identify the specific clauses in the collective agreement that are being violated, and provide details with respect to the redress or remedy being sought.
- (f) "Days" means calendar days.
- (g) A Union member submitting a grievance has the right to be assisted by an attending Union Representative or Union Steward.

## 14.03 <u>Grievance Process</u>

A Personal Grievance shall be initiated by the Employee as follows:

#### Level I

The grievance shall be taken up with the Employee's manager or designate. an Employee wishes to pursue a Personal Grievance, the Employee and/or Union Representative shall advise the Employee's manager or designate that they wish to initiate a Level I grievance. The Employee and/or the Union representative will discuss the subject of the proposed grievance with the Employee's manager or designate within fourteen (14) calendar days of the receipt of the grievance. date upon which the subject of the grievance, in an attempt to resolve the matter. The manager or designate

first level classified supervisor will provide a response to the Employee and/or Union Representative within seven (7) calendar days from when the proposed grievance was discussed.

## Level II

If the Employee continues to be dissatisfied, he should submit the grievance in writing within fourteen (14) days of receiving the response from Level I, to the Department Director. The grievance shall state the general nature of the grievance, the Articles in the Collective Agreement alleged to have been breached and the redress or the remedy sought. The Director, in consultation with Human Resources, will hear the matter and will respond in writing to the Grievor within seven (7) calendar days.

#### Level III

If the Union does not accept the decision of the Department Director, they may request in writing within seven (7) days of receiving the Department Directors decision, that the Chief Commissioner review the grievance. The Chief Commissioner shall hear the matter within ten (10) days and will advise the Union of his decision in writing within fourteen (14) days of the hearing;

The Chief Commissioner may designate an Associate Commissioner to hear and decide on Level III grievances;

If the grievance is not settled as a result of the Chief Commissioner's decision, the Union or the Employer may refer the matter to arbitration pursuant to the Labour Relations Code.

#### 14.04 Arbitration Board

In the event the matter being grieved is presented to an Arbitration Board, each Party shall bear the expense of its respective nominee and shall bear one-half (1/2) of the expenses of the Chairman of the Arbitration Board.

- 14.05 The Parties may mutually agree to have the matter heard by a single Arbitrator.
- 14.06 The Arbitration Board by its decision shall not alter, amend or otherwise change the terms of this Collective Agreement.
- 14.07 All Union members submitting a grievance have the right to be assisted by an attending Union Representative.

#### 14.08 Procedure and Time Limits

Extension of Time Limits: Time limitations set out in the above grievance procedure may be extended by mutual agreement between the parties and must be in writing.

Time Limits: Time limits and procedures contained in this grievance procedure are mandatory. Failure to pursue a grievance within the prescribed time limits and in accordance with the prescribed procedures shall result in abandonment of the

grievance and all rights and recourse to the grievor.

Failure to reply to a grievance in a timely fashion shall advance the grievance to the next level. Grievances so advanced shall be subject to time limits as if a reply had been made on the last allowable day of the preceding level in the procedure.

14.09 Policy Grievances may be initiated either by the Union or the County in the following manner:

- (a) Policy Grievances shall be initiated either by the Union or the County in writing, within thirty (30) calendar days from the day that the matter giving rise to the grievance came to the attention of the Designated Officer of the Union or Official of the Employer. The written notice of the grievance shall be submitted to the President of the Union or the Director, Human Resources of Strathcona County and shall stipulate the general particulars of the grievance together with reference to the allegedly violated clause of the Agreement;
- (b) Grievances submitted in accordance with Clause 14.09 shall be referred to a Grievance Committee which shall endeavour to resolve the difference. This Committee will be composed of two representatives of the County and two members of the Union appointed by the Executive. This Committee shall meet within fifteen (15) calendar days of the submission of the written notice by either party;
- (c) After the parties have met in accordance with the preceding clause, each party will give any decision made in writing to the other party within fifteen (15) calendar days. In the event that the two parties reach a mutually acceptable agreement concerning the matter in question, a written and signed notice of mutual agreement will supersede the aforementioned notice of decision:
- (d) In the event that the parties do not reach a mutually acceptable agreement, or that either party does not accept the written notice of decision of the opposite party within thirty (30) calendar days of the Chief Commissioner's decision as per Clause 14.03 either party may notify the other party in writing of its intention to submit the grievance to arbitration pursuant to the provision of the Labour Relations Code.

#### ARTICLE 16 - HOURS OF WORK AND OVERTIME

The prescribed regular hours of work for an Employee, other than those referred to in Clause 16.04-5 following, shall be eight (8) hours in any one (1) day or forty (40) hours in any one (1) calendar week or five (5) working day period. Hours worked in excess of the above shall be calculated as overtime hours and shall be paid accordingly. Notwithstanding the above, when an established schedule provides a pattern that is irregular, the forty (40) hours in one (1) calendar week or

five (5) working day period shall mean average over a two (2) week period.

### 16.02 Overtime

The overtime pay rates shall be calculated at the rate of two (2.0) times the regular pay rate for all hours worked beyond eight (8) hours in a day.

- An Employee may bank overtime hours at the rate it is earned; up to the maximum of forty (40) regular hours within a calendar year. An Employee may request to be paid out for banked hours up to the time equivalent of forty (40) regular hours per calendar year. Banked hours may be taken as time off by Employees after receiving written authorization from the County designated official who shall consider the wishes and application of the Employee. Up to forty (40) banked regular hours may be carried forward into the next calendar year. Banked hours cannot exceed the time equivalent of eighty (80) regular hours at any one time.
- 16.04 For those Employees who are engaged in road construction, road maintenance or the removal of snow from roads or any of those undertakings, for the purpose of calculating overtime, their regular day shall consist of a twenty-four (24) hour period beginning at midnight. The maximum number of hours of work at regular rates of pay shall be those prescribed or allowed under the Employment Standards Code and Regulations thereunder, but shall not exceed:
  - (a) Ninety (90) hours bi-weekly, except where permitted under the Employment Standard Code;
  - (b) Eighty (80) hours bi-weekly during the period November 01 to April 30, inclusive;
- 16.05 Only hours actually worked, paid vacation (including general holidays) and banked time used to supplement hours are used in accumulation of overtime. Sick leave or other paid or unpaid absences are not included.
- 16.06 For those Employees specified in Clause 16.04 from May 1 to October 31, overtime rates shall be calculated at the rate of two (2.0) times the regular pay rate for all hours worked beyond nine (9) hours in a day. This article only comes into effect when actually scheduled to work nine (9) hours a day.

## 16.06 <u>Posting of Shift Schedules</u>

Barring unforeseen circumstances and emergency situations, shift schedules shall be posted no less than thirty (30) days before the expiry of the current schedule.

#### 16.07 Shift Changes

- (a) When an Employee is required to change from one shift to another shift, the Employer will provide at least twenty-four (24) hours written notice in advance of the start time of the originally scheduled shift and ten (10) hours off between shifts.
- (b) If the twenty-four (24) hours written notice and ten (10) hours off between shifts

- is not provided to the Employee, the Employee will be paid at the rate of two (2.0) times the regular rate of pay for the hours of the first shift worked.
- (c) If the Employee is on regular scheduled days off when the twenty-four (24) hours written notice is posted, the supervisor may contact the Employee directly to provide verbal twenty-four (24) hours notice change in shift.
- An Employee shall be paid for or expected to work for a minimum of three (3) hours if the regular shift is cancelled or rescheduled and the Employee was not notified of such cancellation or rescheduling on or before the day twelve (12) hours prior to the cancelled or rescheduled shift.
- 16.09 When an Employee calls in ill for a work period affected by the preceding Clause 16.08, that Employee will be paid only for the minimum of three (3) hours that he would have been paid had he reported for work in the regular fashion. This clause shall not apply in cases in which the Employee provides proof of attendance at either medical, dental or optical appointments. In such cases, the Employee will be paid for either the half-day or full-day as appropriate to the circumstances and as provided in Clause 21.03 and 21.04.

## 16.10 Early Reporting

- (a) When an Employee is requested prior to the conclusion of a shift, to report or work more than two (2) or more hours earlier than his regularly scheduled shift, that Employee may (subject to operational requirements) be given a choice of leaving work prior to the end of their regularly scheduled shift or being paid the regular overtime rate for those hours worked prior to the time scheduled for the commencement of the regular shift. The total of the early reporting hours worked and regular shift hours worked will not be less than the number of regularly scheduled hours. Such early reporting arrangement may continue for two consecutive shifts.
  - If requested to report early on the second shift the early reporting hours will be at the overtime rate. If requested for a third consecutive shift, Clause 16.07 will apply.
- (b) When an Employee is requested prior to the conclusion of a shift to report up to two (2) hours earlier or up to two (2) hours later for the next regular shift, that Employee shall not receive the overtime rate for the earlier or later reporting period. The total of the earlier reporting hours worked and regular shift hours worked will not be less than the number of regularly scheduled hours. If requested to report earlier or later on the second shift the earlier or later reporting hours will be at the overtime rate. If requested for a third consecutive shift, Clause 16.07 will apply.

#### 16.11 Attendance

When an Employee is unable to report to work, the Employee shall follow the established protocol for their department to report such absence, and at a

minimum shall report the absence one (1) hour prior to the start of the shift. within one hour prior to the scheduled shift start time. Failure to comply with this requirement shall result in loss of pay for the shift, and may result in discipline unless the Employee provides a reasonable explanation.

#### 16.124 Work Break

All Employees shall be entitled to one fifteen (15) minute work break in each half day worked, or in any overtime work period in excess of two (2) hours. Such work break shall be taken at the work site and the time of such break shall be approved by the immediate supervisor of each Employee. The work break shall be considered part of a work hour for the purposes of the preceding paragraphs.

#### 16.13 Sand Shift Workers and Sidewalk Crew

Notwithstanding Clause 16.01, when the needs of the operations require it, Sand Shift Workers and Sidewalk Crew may be required to work a modified shift schedule of up to twelve (12) hours. Modified hours of work shall commence annually on the first full pay period commencing on or after October 15 and shall continue until the first full pay period before or after April 15 based on operational need.

- (a) Overtime will be paid to Sand Shift Workers and Sidewalk Crew who exceed 80 hours of work in a bi-weekly pay period.
- (b) Employees working modified hours of work will have all benefits and entitlements which are expressed in terms of daily or weekly entitlement converted to produce the equivalent hours of benefits and entitlements as they would have had if the hours of work had not been modified. This will result in no loss or gain in Employee benefits and entitlements.

### ARTICLE 17 - SHIFT AND WEEKEND PREMIUMS

17.01 (a) A shift premium of one do

- (a) A shift premium of one dollar and forty cents (\$1.40) one dollar and ninety cents (\$1.90) per hour shall be paid to all Union Employees for those hours worked between seventeen hundred (1700) hours and zero seven hundred (0700) hours Monday to Friday and for all hours worked on Saturday and Sunday and holidays.
- (b) At no time shall shift premium be included with the Employee's regular rate of pay for purposes of computing overtime payments, other premium payments, or any Employee benefits.

#### ARTICLE 20 - RELIEF AND ACTING INCUMBENT PAY

#### 20.01 Relief & Acting Pay

Any Employee who works in a classification (within the Bargaining Unit) with a higher rate of pay in any one (1) shift shall receive the higher rate of pay for each hour worked in the higher classification. If this time period exceeds ninety (90) days, effective the ninety-first (91st) day, the Employee will receive the higher rate of pay for all approved vacation, general holiday, illness, bereavement, and compassionate leave while working in the higher classification.

- 20.02 Notwithstanding the premium in Clause 20.09, An Employee appointed to Leadhand or Project Foreman shall not receive an hourly rate of pay that is less than the "County Department Union Employees" that he is responsible for.
- 20.03 Certified Technicians appointed to a seasonal Field Certified Technician assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- 20.04 Vehicle Serviceman II appointed to a seasonal Field Vehicle Serviceman II assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- The qualification or eligibility of an Employee to relieve or be assigned to a higher classification will be the responsibility of management.
- 20.03 An Employee who has become incapacitated by injury or illness may be employed in other work which he can do. Such Employee may not displace an Employee with more seniority.
- 20.04 An Employee who refuses without good and satisfactory reason, to accept the alternate classification or position, may be suspended immediately. The supervisor issuing the suspension will report the suspension to the Department Head (Human Resources) as soon as possible who will notify the Union Executive.
- 20.05 Any Employee affected by the preceding Clauses 20.03 and 20.04 and not satisfied with the action recommended in accordance with Clause 20.04, may appeal the reclassification or suspension in keeping with Clause 14.03 of this Agreement.
- 20.06 Should an Employee who has completed the probationary period be unable to perform his usual duties by reason of the loss or suspension of his motor vehicle operator's license or a similar happening, that Employee may be placed in some other suitable position until the disability ends or is removed. The wages paid during the period of such disability shall be wages appropriate to the position in which the Employee is placed.
- 20.07 Notwithstanding the above clause, within fourteen (14) days of the suspension, an Employee may be allowed to book unused holidays at the wage prevailing prior to suspension of his operator's license.

#### Acting Incumbent

- 20.08 Any member of the Union may be appointed for one (1) or more days as a **Classified** supervisor and such appointment will not cause the member to be disqualified from the Union.
- 20.09 If any Employee is appointed temporary Classified Supervisor, they shall receive a premium of two dollars and twenty-five cents (\$2.25) per hour in addition to their regular wage rate for one (1) complete shift or longer.
- 20.10 Any member of the Union may be appointed for a period of time to the position of temporary Classified Supervisor and such appointment will not cause the member to be disqualified from the Union.
- 20.11 Notwithstanding the premium in Clause 20.09, an Employee appointed to Leadhand or Project Foreman shall not receive an hourly rate of pay that is less than the "County Department Union Employees" that he is responsible for.
- 20.12 Certified Technicians appointed to a seasonal Field Certified Technician assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- 20.13 Vehicle Serviceman II appointed to a seasonal Field Vehicle Serviceman II assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.

#### ARTICLE 21 - PROVISION FOR ILLNESS AND NORMAL ABSENCE

- 21.01 Sick leave benefits as herein stated are sponsored by the County to protect the Employee in the event of an unavoidable illness or injury not covered by the Alberta Workers' Compensation Act.
- 21.02 When an Employee is unable to report to work, the Employee shall follow the established protocol for their department to report such absence within one hour prior to the scheduled shift start time. Failure to comply with this requirement shall result in loss of pay for the shift, and may result in discipline unless the Employee provides a reasonable explanation. Moved to Article 16
- A Regular Employee, upon hire, shall be entitled to a maximum of eighty (80) hours ten (10) working days per calendar year of leave with pay for any casual illness, which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less. Up to sixteen (16) hours two (2) days of unused casual illness leave per calendar year shall be carried over and may be used in the next calendar year. The maximum paid casual illness leave in a calendar year is ninety-six (96) hours twelve (12) days including the carry over.
- 21.04 A Temporary Employee after the completion of the standard probationary period,

shall be entitled to a maximum of thirty-two (32) hours four (4) working days per calendar year of leave with pay for any casual illness, which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less. Temporary Employees with more than six (6) months of service in the current calendar year shall be entitled to an additional sixteen (16) hours two (2) working days in the current calendar year, of leave with pay for any casual illness. Up to two (2) days of allotment can be used to attend to illness of immediate family members.

- 21.05 Each day or portion of a day of absence due to casual illness used within a calendar year, shall be deducted from the remaining casual illness leave entitlement for that calendar year.
- NEW Where an employee terminates and is rehired in the same calendar year, the casual illness allotment will be the amount that they had in their available bank at the time of departure.
- In any case, an Employee may be required to provide satisfactory proof of illness upon return to work, where it can be shown that reasonable doubt exists in respect to the purpose of an absence claimed to be due to illness. The employee will be provided the reasons for the request. Such proof may take the form of a medical certificate signed by a qualified physician, dental practitioner or other professional as deemed appropriate by the Disability Management Supervisor/Human Resources.
- 21.07 Upon approval (which will not be unreasonably denied), time off of two (2) hours or less for doctor appointments will result in no loss of casual illness or regular time. The Employer may require the Employee to submit proof of attendance at a medical, dental or optical appointment when time off work is granted to attend such appointments.
- 21.08 Each Regular Employee will be entitled to leave of absence with pay for General Illness in accord with the following scale. General Illness is defined as "any illness which causes an Employee to be absent from work for a period of more than three (3) consecutive work days".

To obtain General Illness benefits resulting from the inability of the Employee to perform the duties of his position, a proper medical certificate(s) signed by a qualified physician, dental practitioner or other professional as deemed appropriate by the Disability Management Supervisor/Human Resources is required.

## 21.09 <u>Family Illness Leave</u>

A Regular Employee shall be entitled to a maximum of twenty-four (24) hours three (3) working days per calendar year with pay to attend to illness of immediate family

members as per defined in Clause 22.06.

- 21.10 A Temporary Employee may use up to sixteen (16) hours two (2) days of their casual illness allotment, as per Clause 21.04 can be used to attend to illness of immediate family members.
- 21.0911 Effective the calendar year following ratification, Sick leave entitlement for illness longer than three (3) working days shall be as follows:
  - (a) After ninety (90) days three (3) months five hundred (500) hours of continuous employment the Employee shall receive one hundred sixty (160) hours twenty (20) working days at the base rate of pay;
  - (b) On the first day worked, in the next calendar year sick leave entitlement shall be increased to a maximum **two hundred eighty (280) hours thirty-**five (35) working days at the base rate of pay
  - (c) On the first day worked, in subsequent consecutive calendar years sick leave entitlement shall be increased to a maximum of **five hundred twenty** (520) hours sixty five (65) working days at the base rate of pay, plus any unused casual illness up to a maximum of **six hundred** (600) hours seventy five (75) working days.
- 21.<del>1012</del> When a day designated as a Paid Holiday under Clause 23.01 falls within a period of General Illness, it will be considered as a paid holiday.
- 21.<del>12</del>13 Transfers, promotions, demotions and/or reclassification shall not affect sick leave entitlement of any Employee.
- 21.<del>11</del>14 Workers Compensation
  - a) An Employee is not eligible to receive sick leave benefits if the absence is due to an injury while in the employ of any other Employer that is covered by Workers' Compensation Benefits.
  - b) Employee's will be eligible to receive benefits in accordance with the County's policies.
- 21.12 Transfers, promotions, demotions and/or reclassification shall not affect sick leave entitlement of any Employee.
- 21.4315 Sick leave entitlement will not be affected while the Employee is in receipt of Alberta Workers' Compensation benefits.
- 21.1416 Regular Employees who are eligible to receive Workers' Compensation benefits shall receive full pay from the County for a period not to exceed six (6) months provided that all Workers' Compensation payments are assigned to the County for that period.

- 21.4517 Expenses for medical certificates will be the responsibility of the employee, including medical certificates required to support eligibility for casual or general illness. Where the Employer has requested the employee to attend a medical examination and/or obtain additional medical documentation, the Employer shall reimburse the employee for the cost of the documentation (if any).
- 21.<del>1618</del> An Employee who has become incapacitated by injury or illness may be employed in other work which he can do. Such Employee may not displace an Employee with more seniority.
- An Employee who refuses without good and satisfactory reason, to accept the alternate classification or position, may be suspended immediately. The supervisor issuing the suspension will report the suspension to the Department Head (Human Resources) as soon as possible who will notify the Union Executive.
- 21.1820 Any Employee affected by the preceding Clauses 21.176 and 21.187 and not satisfied with the action recommended in accordance with Clause 20.04, may appeal the re-classification or suspension in keeping with Clause 14.03 of this Agreement.
- 21.1921 Should an Employee who has completed the probationary period be unable to perform his usual duties by reason of the loss or suspension of his motor vehicle operator's license or a similar happening, that Employee may be placed in some other suitable position until the disability ends or is removed. The wages paid during the period of such disability shall be wages appropriate to the position in which the Employee is placed.
- 21.<del>2022</del> Notwithstanding the above clause, within fourteen (14) days of the suspension *of* their operator's license, an Employee may be allowed to book unused holidays at the wage prevailing prior to suspension of his operator's license.

#### ARTICLE 22 -LEAVES OF ABSENCE

- Other Leaves of Absence, without pay, may be granted at the sole discretion of the County, and in accordance with the Alberta Employment Standards Code, upon written application by the affected Employee at least seven (7) days prior to the beginning date of the applied for Leave of Absence. To be eligible for such leave, an Employee shall have been employed with the County for ninety (90) days of continuous employment, with the exception of reservist leave which is twenty-six (26) consecutive weeks, and must have his application endorsed or approved by the Department Head. The application shall include the reason for the request as well as the dates of the requested Leave of Absence. Such approval should not be unreasonably withheld.
- 22.02 In unusual circumstances or by mutual agreement, the requirement stated in Clause 22.01 for application to be made seven (7) days prior to the commencement of the requested leave of absence may be waived. Application for such waiver is

to be made directly to the Department Head.

22.03 In the case of an Employee who takes a leave of absence under the provisions of Clause 22.01, the benefits and conditions to which that Employee is entitled as a result of length of service in the County shall not be adversely affected as a result of having been granted such leave of absence.

An Employee on such approved leave may not engage in other employment for gain while on such leave without the express written consent of the County.

## 22.05 Court and Jury Duty

Leave of Absence shall be provided to an Employee who is required to serve as a juror, or is subpoenaed as a witness in a private capacity in any court proceeding. The Employee shall receive their regular rate of pay and shall remit to the Employer any allowance they may receive for such duty or service.

If the Employee is subpoenaed as a witness to appear in court in their official capacity to give evidence or to produce Strathcona County records, the Employee shall receive their regular rate of pay during regular working hours and is eligible to receive overtime pay to appear during non-work hours. The Employee shall remit to the Employer any allowance they may receive for such duty or service.

The Employee shall present proof of such duty or service and the amount of remuneration or allowance received in all cases.

Employees required to appear in court in a capacity other than subpoenaed as a witness or jurist will be granted leave without pay.

#### 22.06 Bereavement Leave

An Employee shall be granted Bereavement Leave of Absence with pay for not more than five (5) consecutive days because of the death of spouse (includes common-law), child (includes step, foster child or ward), parent, (includes step or foster parent), brother, sister, parent of spouse, guardian, son-in-law, daughter-in-law, grandchild or any other relative who is a member of the Employee's household, or the spouse of any of the foregoing. In addition to the above, the Employer may grant up to one (1) day travel to attend the funeral if the travel is five hundred (500) kilometres or more away.

- 22.07 Leave of absence for one (1) full day, with pay, shall be granted at the time of to attend the funeral or death of a grand-parent, grandparent-in-law, brother-in-law or sister-in-law, aunt, uncle, niece or nephew, if the funeral is on a work day.
- 22.08 Upon written application, additional Compassionate or Bereavement Leave of Absence with pay, due to exceptional circumstances, may be granted an Employee at the discretion of the Chief Commissioner.
- 22.09 One (1) day of leave may be granted without loss of pay or benefits to attend as a pallbearer or mourner for a deceased Employee. Where the family of a deceased Employee requests pallbearers from the Union, the County may grant the

necessary leave with pay for up to six (6) pallbearers.

## 22.1122.10 Family Illness Leave

A Regular Employee shall be entitled to a maximum of twenty-four (24) hours three (3) working days per calendar year with pay to attend to illness of immediate family members as per defined in Clause 22.06.

22.11 A Temporary Employee may use up to sixteen (16) hours two (2) days of their casual illness allotment, as per Clause 21.04 can be used to attend to illness of immediate family members.

## 22.10 <u>Compassionate Leave</u>

An Employee shall be granted Compassionate Leave of Absence with pay for not more than five (5) consecutive days because of the critical illness of a member of his immediate family as per defined in Clause 22.06 or any other relative who is a member of the Employee's household. The Employee shall provide the County with a letter from the attending physician concerning the fact of the critical illness.

Any additional Compassionate Care Leave shall be provided in accordance with *Alberta Employment Standards Code*.

- 22.11 A Regular Employee shall be entitled to a maximum of three (3) working days per calendar year with pay to attend to illness of immediate family members as per Clause 22.06.
- 22.12 One (1) day leave with pay will be allowed for an Employee to be present at the birth or adoption proceeding of an Employee's child, providing such occurrence is on a scheduled work day.

#### 22.<del>13</del>11 Maternity Leave

Maternity Leave will be granted to Employees in accordance with the Alberta Employment Standards Code.

- (a) Employees must have been employed with Strathcona County for ninety (90) days of continuous employment.
- (b) The maternity leave is not more than sixteen (16) weeks starting at any time during the thirteen (13) weeks immediately before the estimated date of delivery.
- (c) The Employee must give six (6) weeks' written notice of the date she intends to begin leave and four (4) weeks' notice of the date she intends to return to work.
- (d) If an Employee's pregnancy results in a miscarriage or stillbirth within sixteen (16) weeks of their due date, they are entitled to sixteen (16) weeks of maternity leave.

## 22.124 <u>Health Related Portion of the Maternity Leave</u>

During the one-week Employment Insurance waiting period, Strathcona County will top-up salary to ninety-five percent (95%) of the Employee's regular earnings. Verification that Employees are in receipt of employment insurance benefits is necessary to provide the top-up benefit. A medical certificate must be provided to Human Resources, confirming the date the child was born and the duration of the leave that the physician indicates is health related. During the health-related portion of the maternity leave Strathcona County will top-up salary to ninety-five percent (95%) of the Employees regular earnings. The benefit coverage that the Employee has at the time of her leave will continue during the health related portion of the absence. At the end of this period, parental leave will commence, and benefit coverage will continue if the Employee pays both Employer and Employee portions of the cost of the premiums.

When a pregnancy interferes with the performance of duties or the Employer has a concern for the Employee's health, the Employer may reassign the Employee's duties at no loss of pay, or by notice in writing to the Employee, require them to commence sick leave under Clause 21.08.

#### 22.1416 Parental Leave

Employees who have been employed with Strathcona County for ninety (90) days of continuous employment are entitled to parental leave without pay for a maximum of sixty-two (62) weeks. The Employee must give at least six (6) weeks written notice of the date they intend to begin parental leave and four (4) weeks' notice of the date they intend to return to work. During that time they may receive benefit coverage by paying the Employee and Employer portions of the cost of the premiums.

22-1517 Employees wishing to return to work prior to the agreed return date may do so with two (2) weeks' notice.

## 22.1618 Adoption Leave

One (1) day leave with pay will be allowed for an Employee to be present at the birth or adoption proceeding of an Employee's child, providing such occurrence is on a scheduled workday.

The Employer will grant adoption leave to Employees who are adopting a child. The Employee must have been employed with Strathcona County for ninety (90) days of continuous employment. Written notice of leave must be provided to the Employer at least two (2) weeks before Employees can reasonably expect to obtain custody of the child.

- 22.1719 If Employees cannot provide at least two (2) weeks written notice, they must notify the Employer immediately after receiving notice of the adoption.
- 22.1820 Adoption leave consists of not more than sixty-two (62) consecutive weeks of leave without pay within seventy-eight (78) weeks after the child is placed with the

adoptive parents. During that time they may receive benefit coverage by paying the Employee and Employer portions of the cost of the premiums.

22.1921 Only one (1) parent of a child referred to in Clause 22.20 will be granted adoption leave under this clause.

#### 22.2022 Shared Leave

If both parents are Employees of Strathcona County parental or adoption leave may be taken wholly by one of the Employees or be shared by both. Only one (1) parent will be granted parental or adoption leave at a time.

## 22.2123 Anniversary Date

Employees who have been on maternity, parental or adoption leave will have their anniversary date adjusted for evaluation and increment purposes only. They will not earn vacation leave during this period, the exception being for maternity health related reasons.

#### 22.**22**<del>24</del> Additional Leave

Leave of absence without pay beyond the approved maternity, parental or adoption leave may be granted upon written application up to a combined maximum of twelve (12) months at the discretion of the Employer.

#### 22.23<del>25</del> Domestic Violence Leave

Domestic violence leave shall be granted in accordance with the *Alberta Employment Standards Code*.

## 22.**2426** Military Leave

Military leave shall be granted in accordance with the *Alberta Employment Standards Code*.

## 22.25<del>27</del> <u>Citizenship Ceremony Leave</u>

Citizenship Ceremony Leave shall be granted in accordance with the *Alberta Employment Standards Code*.

#### **ARTICLE 23 - HOLIDAYS**

Employees shall be entitled to one (1) day's paid leave for each of the following specified or general holidays:

New Year's Day

Family Day

Good Friday

Easter Monday

Victoria Day

Labour Day

Thanksgiving Day

Remembrance Day

Christmas Day

Christmas Floater

Canada Day Boxing Day

1st Monday in August Truth and Reconciliation Day

And any other day designated as a General Holiday by the Lieutenant Governor in Council under the Employment Standards Code.

- Employees shall receive the recognized specified or general holidays for which they are eligible, with pay; or other days with pay in lieu of such holidays, or pay in lieu as approved by the County, providing they are available for work immediately preceding and following the days or an otherwise approved leave of absence, or they provide a medical certificate attesting to their illness on the day prior or following the general holiday.
- 23.03 The provisions for pay in lieu of such specified or general holidays referred to in Clauses 23.01 and 23.02 shall be based on the number of regular hours which would have been worked had that day been a regular working day.
- In the event a specified or general holiday falls on an Employee's regular day off, an additional day with pay shall be granted in lieu of that day.
- 23.05 If any of the above specified or general holidays should occur during an Employee's vacation, such vacation shall be extended by one day or an additional day with pay shall be granted in lieu of that day.
- 23.06 Temporary Employees, upon hire, shall be eligible for specified or general holidays as per Clause 23.02.
- 23.07 The Christmas Floater holiday shall be observed to give five (5) consecutive days off including the weekend as follows:
  - (a) On December 24 when Christmas Day falls on a Tuesday, a Thursday, a Friday or Saturday;
  - (b) On December 27 when Christmas Day falls on a Monday or a Wednesday;
  - (c) On December 28 when Christmas Day falls on a Sunday.

#### **ARTICLE 24 – VACATION LEAVE**

- 24.01 Annual vacation entitlement with pay shall be as follows:
  - (a) An Employee who has completed less than six (6) years of service shall earn vacation leave at the rate of four point sixty-two (4.62) hours per bi-weekly period of continuous service [fifteen (15) days per year] or five point two (5.20) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
  - (b) An Employee who has completed six (6) years but less than fourteen (14) years of service shall earn vacation leave at the rate of six point fifteen (6.15) hours per bi-weekly period of continuous service [twenty (20) days per year]

- or six point ninety-two (6.92) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
- (c) An Employee who has completed fourteen (14) years of service but less than twenty two (22) years shall earn vacation leave at the rate of seven point seven (7.70) hours per bi-weekly period of continuous service [twenty-five (25) days per year] or eight point sixty-seven (8.67) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
- (d) An Employee who has completed twenty two (22) or more years of service shall earn vacation leave at the rate of nine point twenty-three (9.23) hours per bi-weekly period of continuous service [thirty (30) days per year] or ten point thirty-nine (10.39) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week.
- 24.02 If Employees become ill on an annual vacation leave for more than three (3) consecutive days-paid vacation days, the applicable days will be deemed general illness, and they will have the applicable number of vacation days reinstated upon application and provision of a medical certificate.
- The number of days of vacation leave which may be accumulated by an Employee is limited to the number he can earn in eighteen (18) months of continuous service.

The County will automatically pay out any excess vacation leave over the maximum accumulation at the beginning of each December.

When vacation pay is paid out rather than being taken as vacation, the payout shall be based on the Employee's rate of pay at the time of payout.

- 24.04 Vacation leave may be taken in the calendar year it is earned. However, vacation leave may not be taken until it has been earned. The vacation leave may be granted as follows:
  - (a) In one continuous period;
  - (b) As single days off or some other combination;
  - (c) If an Employee earns vacation at a rate of four point sixty-two (4.62) hours per bi-weekly period of continuous service, one (1) week vacation leave may be taken as single days off or some other combination provided the total does not exceed five work days in any calendar year;
  - (d) If an Employee earns vacation at a rate of six point fifteen (6.15) hours per bi-weekly period of continuous service or seven point seven (7.70) hours per bi-weekly period of continuous service, two (2) weeks' vacation leave may be taken as single days off or some other combination provided the total does not exceed ten (10) work days in any calendar year;
  - (e) If an Employee earns vacation at a rate of nine point twenty-three (9.23) hours per bi-weekly period of continuous service, three (3) weeks' vacation

leave may be taken as single days off or some other combination provided the total does not exceed fifteen (15) work days in any calendar year.

In addition to the automatic payout of excess vacation referred to in 24.03, an Employee may be permitted to receive a cash payment in lieu of vacation leave, provided any and all obligations have been met in accordance with Employment Standards legislation and the Employee has received written approval from the Department Head.

Temporary Employees shall have vacation pay added to their bi-weekly pay cheques and shall be entitled to unpaid vacation leave in accordance with the Employment Standards Code.

- 24.06 An Employee who does not complete one (1) full year of service shall be entitled to vacation pay in accordance with the Employment Standards Code and regulations thereunder.
  - (a) Temporary Employees will receive vacation pay at four percent (4%) of their gross regular earnings during their first four thousand (4000) regular standard hours.
  - (b) Temporary Employees will receive vacation pay at six percent (6%) of their gross regular earnings after completion of four thousand (4000) regular standard hours.
  - (c) If a Temporary Employee has a break in service greater than eighteen (18) months then vacation pay shall be paid at four percent (4%) of their gross regular earnings as per Clause 24.076 a) above.
- 24.07 Regular vacation leave shall be taken by Employees after receiving written authorization from the County or its designated official who shall consider the wishes and application of the Employee. Length of regular service Seniority with the County shall be the determining factor in resolving any dispute or conflict regarding preference for the time at which vacations are to be taken. Changes in vacation scheduling shall be authorized in writing.
- 24.08 Vacation hours will be paid out at the Employee's base rate in place at the time the vacation is taken.

#### **Summer Vacation Schedules**

#### 24.09 Summer Vacation Schedules

Employees must submit summer vacation applications to the Director or his designate by March 15th of each year. The Employer will post schedules, as approved by the Director, by May 1st of each year.

#### **ARTICLE 25 - BENEFIT AND PENSION PLANS**

## 25.01 Benefit Plans

The County shall contribute on behalf of all participating Employees in group insurance and benefit plans, which provide minimum benefit levels agreed upon by the County and the Union at rates outlined in the following clauses:

- (a) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Strathcona County Benefit Plan Life, Accidental Death and Dismemberment and Long-Term Disability Insurance. All members of the Union hired after July 1, 1978, shall be members of the said plan as a condition of employment;
- (b) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Extended Health Care option of the Strathcona County Employee Benefit Plan;
- (c) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Dental option of the Strathcona County Employee Benefit Plan;
- (d) The County shall pay seventy five percent (75%) of the total premium for each eligible Union member enrolled in the Alberta Health Care Insurance Plan through the Strathcona County.

The County's liability shall be limited to the payment of premiums, where required, for the various plans and the County shall be relieved of any further liability with respect to the plans.

25.02 In the event that the County should undertake to transfer group insurance and benefit plans from one carrier to another, it is agreed that the Union will be consulted prior to the negotiation of such coverage.

## **Benefits for Temporary Employees**

#### 25.03 Benefits for Temporary Employees

To be eligible to participate in the Strathcona County Employee Benefit plans, Temporary Employees must meet the following conditions:

(a) All Temporary Employees who have been employed for twelve (12) consecutive cumulative months or more, provided there has been no break in service for longer than six (6) months, shall receive the same benefits as Regular Employees, with the exception of pension and the LTD benefit entitlement will be limited to a maximum of twenty-four months.

For the purposes of calculating the twelve (12) cumulative months, leaves of absence of more than thirty (30) calendar days shall be excluded.

(b) An exception to the above shall be made for Temporary Employees replacing Regular Employees on long term disability. A Temporary Employee in this situation will be eligible for benefits after six (6) consecutive months of employment.

#### Pension Plan

#### 25.04 Pension Plan

The County participates in the Local Authorities Pension Plan. Participation in this plan is compulsory for "Regular Employees" after they have completed one (1) year of service or immediately if transferring from another Local Authority Board Employer.

25.05 The County shall make payroll deductions as follows:

- (a) The County shall deduct legislated Revenue Canada deductions and Local Authorities Pension Plan contributions as required from each bi-weekly pay cheque;
- (b) The current authorized deductions for Canada Savings Bonds will be taken each pay cheque;
- (c) The Employee contribution to the Strathcona County Employee Benefit Plan will be deducted from the first bi-weekly pay cheque. each month.

#### **ARTICLE 29 - TEMPORARY EMPLOYEES**

- 29.01 Temporary Employees shall be entitled to the terms and conditions of this Agreement, except where Regular Employees are referenced. Seniority for employment periods of less than six (6) months shall not be accumulated.
- A Temporary Employee shall be considered probationary as per Article 11.01 and shall not serve probation for subsequent years of employment unless a break of service of greater than one (1) year has occurred or the Employee is returning to a different position. Temporary Employees shall have access to the application of the grievance procedure except in connection with termination of employment as a result of either:
  - (a) Unsatisfactory performance, or;
  - (b) The abolishment of the position occupied by the Temporary Employee.
- 29.03 For the purpose of determination of Pay Grade and Step a Temporary Employee returning for subsequent employment within one (1) year may be required to undertake the appropriate probationary period unless he is returning to a position in which he has previously fulfilled the probationary requirements.
- 29.04 When Temporary Employees are recalled, it shall be based on:

- (a) A satisfactory performance evaluation from the previous year of employment;
- (b) The Employee having qualifications to do the job for which he is recalled;
- (c) The Employee's accumulated service.

The opportunity for recall shall exist for six (6) months within the Branch.

This clause will not apply to Employees participating in the NAIT or other post-secondary Water and Wastewater Training Program.

#### ARTICLE 30 - GENERAL WAGE ADMINISTRATION

The rates of pay for all Employee members of the Union excluding Apprentices, covered by the Agreement shall be as specified in the attached Schedule A.

Temporary Employees returning for subsequent employment within one (1) year shall have their prior service recognized for the purposes of determining Pay Grade and Step.

30.02 An Employee is entitled to the rates indicated in Schedule A in the following way:

- (a) To Step One (1) upon hiring;
- (b) To Step Two (2) upon completion of the five hundred (500) hours of work in the classification and recommendation of the appropriate supervisor;
- (c) To Step Three (3) upon completion of an additional fifteen hundred (1,500) hours of work in the classification and the recommendation of the appropriate supervisor
- (d) Water & Wastewater Operators step progression is as stated in the Water and Wastewater Job Summary attached to the Collective Agreement. Water and Wastewater Operators step progression shall be:
  - (i) To Step One (1) upon hiring;
  - (ii) To Step Two (2) upon completion of five hundred (500) hours of work in the classification, additional qualifications (in accordance with the job description), and recommendation of the appropriate supervisor;
  - (iii) To Step Three (3) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
  - (iv) To Step Four (4) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
  - (v) To Step Five (5) upon completion of an additional one thousand eight

hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;

(vi) To Step Six (6) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;

## (e) Construction Operator step progression shall be:

- (i) To Step One (1) upon hiring;
- (ii) To Step Two (2) upon completion of two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (iii) To Step Three (3) upon completion of an additional two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (iv) To Step Four (4) upon completion of an additional two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the description), and the recommendation of the appropriate supervisor.

## (f) Water and Wastewater Leadhand step progression:

Water & Wastewater Operator, Steps 1 to 4	Water & Wastewater Leadhand, Step 1
Water & Wastewater Operator, Step 5	Water & Wastewater Leadhand, Step 2
Water & Wastewater Operator, Step 6	Water & Wastewater Leadhand, Step 3

## (g) Water and Wastewater Infrastructure Repair Technician step progression:

Water & Wastewater Operator, Steps 1 to 4	W&WW Infrastructure Repair Technician, Step 1
Water & Wastewater Operator, Step 5	W&WW Infrastructure Repair Technician, Step 2
Water & Wastewater Operator, Step 6	W&WW Infrastructure Repair Technician, Step 3

- (h) An appropriate form will be completed and placed In the Employee's personnel file when the wage rate is amended.
- 30.03 Should an Employee not be recommended by the appropriate supervisor for advancement as per Clause 30.02 b), c), d) or e), the Administration shall advise that Employee of the reasons in writing, and in addition:
  - (a) What action might be taken in order for the wage to be advanced.
  - (b) In the event that the wage will not be advanced, what alternate action the administration intended to take.
- 30.04 An Employee affected by Article 30.03 may file a grievance for a failure to be recommended for an increase.
- 30.05 The rate of pay established in Schedule A shall apply during the term of this Agreement. Employees shall be paid bi-weekly. On each pay day, each Employee shall be provided with an itemized statement of his wages, overtime and other earnings and deductions.

## 30.06 Flag Person

The Employer may hire Temporary Employees as Flagperson at a rate in accordance with Schedule A. Should such employees be assigned to any other work they shall be paid in accordance with Clause 20.01. No one shall be regularly assigned duties as Flagperson until they have successfully completed an approved Flagperson's training course. Employees taking the Flagperson training shall be paid the appropriate rate of pay while training.

#### ARTICLE X - DIRECT RESPONSILBE PAY (DRP)

- When a Water and Wastewater Operator or Water and Wastewater Leadhand is assigned to the Utilities On-Call Schedule as On-Call #1, On-Call #2 or Transmission Operator (for emergent calls) and deemed to be in direct responsible charge (DRC), the Employee shall be paid an additional, two dollars and twenty-five cents (\$2.25) per hour on the base rate of pay, for all hours worked between sixteen hundred and thirty (1630) hours and zero eight hundred (0800) hours Monday to Friday and for all hours worked on Saturday and Sunday and Holidays.
  - (a) At no time shall Operators called out to assist the On-Call #1, On-Call #2 or Transmission Operator with emergency service or other similar callouts receive DRP.
- X.02 Where applicable, an Employee shall be eligible to receive DRP in addition to overtime, shift and weekend premiums.

X.03 At no time shall DRP be included with the Employee's regular rate of pay for the purpose of calculating overtime payments, other premium payments, or any Employee benefits.

#### WAGE INCREASES:

```
January 1, 2020 – 0.5% - Retro to be paid out by April 12, 2023

January 1, 2021 – 1% LUMP SUM – Retro to be paid out by April 12, 2023

January 1, 2022 – 1% - Retro to be paid out by April 12, 2023

January 1, 2023 New Compensation structure and pay grid

January 1, 2023 – 1.5% - January 1, 2023 implementation
```

The following is included for clarity, but will not form part of the collective agreement:

- January 1, 2020 Cola increase will be retroactive to January 1, 2020 and payable by April 12, 2023.
- January 1, 2021 lump sum will be payable by April 12, 2023.
- January 1, 2022 COLA increase will be retroactive to January 1, 2022 and payable by April 12, 2023.
- January 1, 2023 COLA increase will be implemented by January 1, 2023, as long as ratification occurs by November 1, 2022.
- The new compensation structure and pay grid will be effective January 1, 2023.
- Temporary Employees will be eligible for benefits December 12, 2022 as long as ratification occurs by November 1, 2022.
- Shift Premiums are not retroactive and will be effective December 12, 2022 as long as ratification occurs by November 1, 2022.
- Unless otherwise specified, all changes are effective on the date of ratification.
- Premiums referenced in Article 17 will be included in pensionable earnings.

The following articles have no amendments and will remain as current, except for any consequential amendments that may arise as a result of other agreed to amendments:

**Article 5 – Management Rights** 

Article 7 – Time off for Union Business

Article 13 – Discipline

Article 18 – Call out & Standby

# Schedule A January 1, 2023

Code	Position	Junuar y 1, 2023	Hourly	/ Rates	
		2020 - 0.5% increase	\$16.89	\$17.51	
3001	Flag Person	2022 - 1.0% increase	\$17.06	\$17.69	
	_	2023 - removed			
		2020 - 0.5% increase	\$20.88	\$21.25	
3002	Labourer I	2022 - 1.0% increase	\$21.09	\$21.46	
		2023 - removed			
		2020 - 0.5% increase	\$23.53	\$24.30	
3003	Labourer II	2022 - 1.0% increase	\$23.77	\$24.54	
		2023 - removed			
		2020 - 0.5% increase	\$25.90	\$26.58	\$27.03
3004	Labourer III	2022 - 1.0% increase	\$26.16	\$26.85	\$27.30
		2023 - removed			
TBD	PW Labourer	2023 - added	\$21.78	\$24.91	\$27.71
		2020 - 0.5% increase	\$24.97	\$25.53	
3005	Pickup/Courier Driver	2022 - 1.0% increase	\$25.22	\$25.79	
		2023 - 1.5% increase	\$25.60	\$26.18	
		2020 - 0.5% increase	\$24.64	\$25.36	
3006	Survey Aide I	2022 - 1.0% increase	\$24.89	\$25.61	
		2023 - 1.5% increase	\$25.26	\$25.99	
		2020 - 0.5% increase	\$26.56	\$27.19	\$27.88
3007	Survey Aide II	2022 - 1.0% increase	\$26.83	\$27.46	\$28.16
		2023 - 1.5% increase	\$27.23	\$27.87	\$28.58
		2020 - 0.5% increase	\$31.87	\$32.62	\$33.41
3008	PW Tech I	2022 - 1.0% increase	\$32.19	\$32.95	\$33.74
		2023 - 1.5% increase	\$32.67	\$33.44	\$34.25
		2020 - 0.5% increase	\$36.78	\$37.27	\$37.65
3009	PW Tech II	2022 - 1.0% increase	\$37.15	\$37.64	\$38.03
		2023 - 1.5% increase	\$37.71	\$38.20	\$38.60
		2020 - 0.5% increase	\$29.25	\$29.91	\$30.38
3032	Gravel Checker	2022 - 1.0% increase	\$29.54	\$30.21	\$30.68
		2023 - 1.5% increase	\$29.98	\$30.66	\$31.14
		2020 - 0.5% increase	\$30.69	\$31.36	\$32.08
3010	Concrete Worker I	2022 - 1.0% increase	\$31.00	\$31.67	\$32.40
		2023 - 1.5% increase	\$31.47	\$32.15	\$32.89
		2020 - 0.5% increase	\$32.21	\$33.11	\$34.02
3011	Concrete Worker II	2022 - 1.0% increase	\$32.53	\$33.44	\$34.36
		2023 - 1.5% increase	\$33.02	\$33.94	\$34.88

Code	Position	Hourly Rates					
	Market and Artificial	2020 - 0.5% increase	\$28.09	\$28.69	\$29.29		
3012	Truck Driver I	2022 - 1.0% increase	\$28.37	\$28.98	\$29.58		
		2023 - removed					
		2020 - 0.5% increase	\$29.75	\$30.48	\$31.19		
3013	Truck Driver II	2022 - 1.0% increase	\$30.05	\$30.78	\$31.50		
		2023 - removed					
		2020 - 0.5% increase	\$32.48	\$33.13	\$33.93		
3014	Truck Driver III	2022 - 1.0% increase	\$32.80	\$33.46	\$34.27		
		2023 - removed					
TBD	Truck Driver - Class 1	2023 - added	\$33.29	\$33.96	\$34.78		
·							
		2020 - 0.5% increase	\$28.09	\$28.81	\$29.23		
3015	Operator I	2022 - 1.0% increase	\$28.37	\$29.10	\$29.52		
		2023 - removed					
		2020 - 0.5% increase	\$31.63	\$32.36	\$33.00		
3016	Operator II	2022 - 1.0% increase	\$31.95	\$32.68	\$33.33		
		2023 - removed					
		2020 - 0.5% increase	\$33.69	\$34.54	\$35.43		
3017	Operator III	2022 - 1.0% increase	\$34.03	\$34.89	\$35.78		
		2023 - removed					
TBD	Equipment Operator I	2023 - added	\$28.80	\$30.43	\$30.92		
TBD	Equipment Operator II	2023 - added	\$32.43	\$33.17	\$33.83		
TBD	Equipment Operator III	2023 - added	\$34.54	\$35.41	\$36.32		
		2020 - 0.5% increase	\$33.43	\$34.29	\$35.18		
3031	Sign Person	2022 - 1.0% increase	\$33.76	\$34.63	\$35.53		
		2023 - 1.5% increase	\$34.27	\$35.15	\$36.06		
		2020 - 0.5% increase	\$33.43	\$34.29	\$35.18		
3018	Yardperson	2022 - 1.0% increase	\$33.76	\$34.63	\$35.53		
		2023 - 1.5% increase	\$34.27	\$35.15	\$36.06		
TBD	North Yardperson	2023 - added	\$37.87	\$38.78			

Code	Position		Hourt	y Rates				
		2020 - 0.5% increase	\$36.94	\$37.83				
3036	PW Leadhand	2022 - 1.0% increase	\$37.31	\$38.21				
		2023 - 1.5% increase	\$37.87	\$38.78				
		2020 - 0.5% increase	\$38.51	\$39.56				
3035	PW Project Foreperson	2022 - 1.0% increase	\$38.90	\$39.96				
		2023 - 1.5% increase	\$39.48	\$40.56				
		2020 - 0.5% increase	\$41.69	\$42.57	\$43.52			
3019	Operations Foreperson	2022 - 1.0% increase	\$42.11	\$43.00	\$43.96			
		2023 - 1.5% increase	\$42.74	\$43.65	\$44.62			
		2020 - 0.5% increase	\$20.88	\$24.30	\$27.03			
3034	W&WW Labourer	2022 - 1.0% increase	\$21.09	\$24.54	\$27.30			
		2023 - 1.5% increase	\$21.41	\$24.91	\$27.71			
	\\\-\-\-\-\-\-\-\-\-\-\-\-\-\-\-\-\-\-	2020 - 0.5% increase	\$30.42	\$33.00	\$35.43			
3037	Wastewater Infrastructure	2022 - 1.0% increase	\$30.72	\$33.33	\$35.78			
	Operator	2023 - 1.5% increase	\$31.18	\$33.83	\$36.32			
	VA (Q. VA (A. V. Q. V. V. V. V. V. C. V.	2020 - 0.5% increase	\$30.42	\$32.93	\$34.43	\$36.90	\$38.64	\$41.11
3020	W&WW Operator (Collection &	2022 - 1.0% increase	\$30.72	\$33.26	\$34.77	\$37.27	\$39.03	\$41.52
	Distribution)	2023 - 1.5% increase	\$31.18	\$33.76	\$35.29	\$37.83	\$39.62	\$42.14
		2020 - 0.5% increase	\$30.42	\$32.93	\$34.43	\$36.90	\$38.64	\$41.11
3049	W&WW Operator (Transmission)	2022 - 1.0% increase	\$30.72	\$33.26	\$34.77	\$37.27	\$39.03	\$41.52
	· · · · · ·	2023 - 1.5% increase	\$31.18	\$33.76	\$35.29	\$37.83	\$39.62	\$42.14
		2020 - 0.5% increase	\$38.46	\$40.20	\$42.67			
3038	W&WW Leadhand	2022 - 1.0% increase	\$38.84	\$40.60	\$43.10			
		2023 - 1.5% increase	\$39.42	\$41.21	\$43.75			
		2020 - 0.5% increase	\$39.86	\$41.57	\$44.09			
3039	W&WW Project Foreperson	2022 - 1.0% increase	\$40.26	\$41.99	\$44.53			
		2023 - removed						
TBD	W&WW Infrastructure Repair Technician	2023 - added	\$40.86	\$42.62	\$45.20			
		2020 - 0.5% increase	\$42.63	\$44.30	\$45.84			
3021	W&WW Foreperson	2022 - 1.0% increase	\$43.06	\$44.74	\$46.30			
		2023 - 1.5% increase	\$43.71	\$45.41	\$46.99			
		2020 - 0.5% increase	\$44.33	\$46.01	\$47.52			
3022	W&WW Electrician	2022 - 1.0% increase	\$44.77	\$46.47	\$48.00			
		2023 - 1.5% increase	\$45.44	\$47.17	\$48.72			
	LUMBER FOR THE BALL	2020 - 0.5% increase	\$52.73	\$54.50	\$56.37			
3033	Utilities Facilities Maintenance Technician	2022 - 1.0% increase	\$53.26	\$55.05	\$56.93			
	Technican	2023 - 1.5% increase	\$54.06	\$55.88	\$57.78			
		2020 - 0.5% increase	\$33.23	\$33.69	\$34.14			
3023	Vehicle Serviceperson I	2022 - 1.0% increase	\$33.56	\$34.03	\$34.48			
		2023 - 1.5% increase	\$34.06	\$34.54	\$35.00			

Code	Position			Hourly Rates		
		2020 - 0.5% increase	\$36.36	\$37.10	\$37.80	
3024	Vehicle Serviceperson II	2022 - 1.0% increase	\$36.72	\$37.47	\$38.18	
		2023 - 1.5% increase	\$37.27	\$38.03	\$38.75	
TBD	Shipper-Receiver, Materials Management	2023 - added	\$21.41	\$24.91	\$27.71	
		2020 - 0.5% increase	\$31.94	\$32.77	\$33.42	
3025	Parts Person I	2022 - 1.0% increase	\$32.26	\$33.10	\$33.75	
		2023 - 1.5% increase	\$32.74	\$33.60	\$34.26	
		2020 - 0.5% increase	\$34.31	\$35.26	\$36.13	
3026	Parts Person II	2022 - 1.0% increase	\$34.65	\$35.61	\$36.49	
		2023 - 1.5% increase	\$35.17	\$36.14	\$37.04	
		2020 - 0.5% increase	\$42.25	\$43.80	\$45.45	
3027	Welder	2022 - 1.0% increase	\$42.67	\$44.24	\$45.90	
		2023 - 1.5% increase	\$43.31	\$44.90	\$46.59	
		2020 - 0.5% increase	\$30.90	\$31.75	\$32.60	
3028	Small Engine Mechanic	2022 - 1.0% increase	\$31.21	\$32.07	\$32.93	
		2023 - 1.5% increase	\$31.68	\$32.55	\$33.42	
		2020 - 0.5% increase	\$47.26	\$48.66		
3040	V&E Leadhand	2022 - 1.0% increase	\$47.73	\$49.15		
		2023 - 1.5% increase	\$48.45	\$49.89		
		2020 - 0.5% increase	\$46.14	\$46.57	\$48.66	
3030	V&E Chargehand	2022 - 1.0% increase	\$46.60	\$47.04	\$49.15	
		2023 - 1.5% increase	\$47.30	\$47.75	\$49.89	
		2020 - 0.5% increase	\$47.26	\$48.66		
3042	Transit Leadhand	2022 - 1.0% increase	\$47.73	\$49.15		
		2023 - 1.5% increase	\$48.45	\$49.89		
		2020 - 0.5% increase	\$42.25	\$43.80	\$45.45	
3062	Fleet Services Technician	2022 - 1.0% increase	\$42.67	\$44.24	\$45.90	
		2023 - 1.5% increase	\$43.31	\$44.90	\$46.59	
		2020 - 0.5% increase	\$28.09	\$28.81	\$29.23	
3043	Utility Operator I	2022 - 1.0% increase	\$28.37	\$29.10	\$29.52	
		2023 - removed				
		2020 - 0.5% increase	\$31.63	\$32.36	\$33.00	
3044	Utility Operator II	2022 - 1.0% increase	\$31.95	\$32.68	\$33.33	

2023 - removed

2023 - removed

2020 - 0.5% increase

2022 - 1.0% increase

\$35.90

\$36.26

\$33.79

\$34.13

\$38.00

\$38.38

{00179431} 39

3045 Utility Operator III

Code	e Position		Hourly Rates			
TBD	Construction Operator	2023 - added	\$30.02	\$33.17	\$36.80	\$38.96
		2020 - 0.5% increase	\$37.32	\$39.45	\$41.63	
3046	Utility Leadhand	2022 - 1.0% increase	\$37.69	\$39.84	\$42.05	
		2023 - 1.5% increase	\$38.26	\$40.44	\$42.68	
		2020 - 0.5% increase	\$41.91	\$43.53	\$45.12	
3047	Infrastructure Foreperson	2022 - 1.0% increase	\$42.33	\$43.97	\$45.57	
		2023 - 1.5% increase	\$42.96	\$44.63	\$46.25	
		2020 - 0.5% increase	\$20.88	\$24.30	\$27.03	
3048	Utility Labourer	2022 - 1.0% increase	\$21.09	\$24.54	\$27.30	
		2023 - 1.5% increase	\$21.41	\$24.91	\$27.71	

		2020 - 0.5% increase			\$33.00
3050	Traffic Support Aid	2022 - 1.0% increase			\$33.33
		2023 - 1.5% increase			\$33.83
	I Itilities bestween sect	2020 - 0.5% increase	\$46.53	\$48.29	\$49.91
3051	Utilities Instrument	2022 - 1.0% increase	\$47.00	\$48.77	\$50.41
	Technician/Electrician Dual	2023 - 1.5% increase	\$47.71	\$49.50	\$51.17
		2020 - 0.5% increase	\$44.33	\$46.01	\$47.52
3052	Utilities Electrician	2022 - 1.0% increase	\$44.77	\$46.47	\$48.00
		2023 - 1.5% increase	\$45.44	\$47.17	\$48.72
		2020 - 0.5% increase	\$44.33	\$46.01	\$47.52
3053	Utilities Instrument Technician	2022 - 1.0% increase	\$44.77	\$46.47	\$48.00
		2023 - 1.5% increase	\$45.44	\$47.17	\$48.72
		2020 - 0.5% increase	\$44.33	\$46.01	\$47.52
3063	Control System Technician	2022 - 1.0% increase	\$44.77	\$46.47	\$48.00
		2023 - 1.5% increase	\$45.44	\$47.17	\$48.72

		2020 - 0.5% increase	\$24.99
3101	Apprentice Auto Mechanic,	2022 - 1.0% increase	\$25.24
	1st year (55%)	2023 - 1.5% increase	\$25.62
	A A. t. B.4 I	2020 - 0.5% increase	\$31.82
3102	Apprentice Auto Mechanic,	2022 - 1.0% increase	\$32.14
	2nd year (70%)	2023 - 1.5% increase	\$32.62
	Anaromtica Auto Machania	2020 - 0.5% increase	\$36.35
3103	Apprentice Auto Mechanic,	2022 - 1.0% increase	\$36.71
	3rd year (80%)	2023 - 1.5% increase	\$37.26
	Appropriate Auto Machania	2020 - 0.5% increase	\$40.89
3104	Apprentice Auto Mechanic,	2022 - 1.0% increase	\$41.30
	4th year (90%)	2023 - 1.5% increase	\$41.92

	Association Hamis Dut Basics	2020 - 0.5% increase	\$27.27			
3111	Apprentice Heavy Duty Mechanic,	2022 - 1.0% increase	\$27.54			
	1st year (60%)	2023 - 1.5% increase	\$27.95			
	A	2020 - 0.5% increase	\$31.82			
3112	Apprentice Heavy Duty Mechanic,	2022 - 1.0% increase	\$32.14			
	2nd year (70%)	2023 - 1.5% increase	\$32.62			
	A	2020 - 0.5% increase	\$36.35			
3113	Apprentice Heavy Duty Mechanic,	2022 - 1.0% increase	\$36.71			
	3rd year (80%)	2023 - 1.5% increase	\$37.26			
	A	2020 - 0.5% increase	\$40.89			
3114	Apprentice Heavy Duty Mechanic,	2022 - 1.0% increase	\$41.30			
	4th year (90%)	2023 - 1.5% increase	\$41.92			
		2020 - 0.5% increase	\$42.25	\$43.80	\$45.45	
3129	Certified Technician (HD)	2022 - 1.0% increase	\$42.67	\$44.24	\$45.90	
		2023 - 1.5% increase	\$43.31	\$44.90	\$46.59	
				,		
	Apprentice Transit Mechanic,	2020 - 0.5% increase	\$27.27			
3131	1st year (60%)	2022 - 1.0% increase	\$27.54			
		2023 - 1.5% increase	\$27.95			
	Apprentice Transit Mechanic,	2020 - 0.5% increase	\$31.82			
3132	2nd year (70%)	2022 - 1.0% increase	\$32.14			
·		2023 - 1.5% increase	\$32.62			
	Apprentice Transit Mechanic,	2020 - 0.5% increase	\$36.35			
3133	3rd year (80%)	2022 - 1.0% increase	\$36.71			
		2023 - 1.5% increase	\$37.26			
	Apprentice Transit Mechanic,	2020 - 0.5% increase	\$40.89			
3134	4th year (90%)	2022 - 1.0% increase	\$41.30			
	4tn year (90%)	2023 - 1.5% increase	\$41.92			

# **SCHEDULE B**

# **January 1, 2023**

EQUIPMENT	MINIMUM OF CLASSIFICATION RATE PAID
Non-Finishing Packers (Rubber Roller, Pad Foot Operator, and combination for non- finishing work)	Equipment Operator I – Step 1
Finishing Packer (Smooth Drum Rollers and combination of finishing work)	Equipment Operator I
Skid Steer / Sidewalk Machine	Operator II Equipment Operator I
Single Axle Truck	Truck Driver I Equipment Operator I
Sign Truck (Single Axle with Manlift)	Truck Driver II Equipment Operator I
<del>Tracked Dozer</del>	Equipment Operator II
Wheel Loader	Equipment Operator II
Tandem Axle Truck	Truck Driver II Equipment Operator II
Street Sweepers	Equipment Operator II
Truck Plow	Equipment Operator II
Sander	Truck Driver II Equipment Operator II
RPM Snow Blower	Equipment Operator II
Road Reclaimer	Equipment Operator II
PB Patcher Patch Truck	Equipment Operator II
Rubber Tire Backhoe	Equipment Operator II
Graders	Operator II Equipment Operator III
Track Hoe	Equipment Operator III + \$0.20 / hour
Paver	Equipment Operator III
Tractor-Trailer	Truck Driver ## Class 1
Directional Driller	Utility Operator III

## Letter of Understanding

#### Between

#### Strathcona County

and

The Alberta Union of Provincial Employees, Local 118, Chapter 009

Subject: Skid Steer/Sidewalk Machine Rate of Pay

- 1. Employees who, at the date of ratification, have a class 3 license, will be paid the Equipment Operator II rate of pay, for hours worked on the Skid Steer/Sidewalk Machine (the "Old Rate").
- 2. Eligible employees will only be paid the Old Rate for a period of two years following the date of ratification. At the expiry of the two (2) years, the employees will then be paid in accordance with Schedule B.
- 3. Employees employed at the time of ratification, who have not yet obtained their class 3 licence, have until April 1, 2023 to obtain their class 3 licence. Employees who obtain their class 3 license by April 1, 2023, will be paid in accordance with paragraphs 1 & 2 above.

From: Merryn Edwards <m.edwards@aupe.org>
Sent: Friday, September 09, 2022 11:27 AM

To: Chantel Kassongo; Sheron M. Ward

Subject: Re: SC / AUPE Tentative Agreement

Attachments: Aug. 24 AUPE Strathcona County settlement proposal .docx

Hi Chantel,

Apologies for not responding sooner.

Please find attached the union response to your August 9 proposal.

In this proposal we have agreed with the Employer on moving the Skid Steer/Sidewalk Machine to the Operator I/Equipment Operator I rate of pay in Schedule B, but we have a counter proposal for you on the Letter of Understanding.

We understood the rationale for encouraging employees to obtain class 3 licenses and recognizing those who already hold them. We expect that part of this rationale is to broaden the pool of employees who are able to operate any of the equipment associated with the Operator II/Equipment Operator II classification.

Our counter would further achieve the employer's stated goal of avoiding the current situation where employees in a particular classification are paid at variable rates of pay, depending on the equipment they are operating. We are proposing that eligible Employees (ie those with prior experience on the skid steer, but who are employed at a lower paid classification, and who hold or achieve a class 3 license by Dec. 31, 2023) would be employed in the Operator II/Equipment Operator II classification.

Our proposal also builds on current practice of reimbursing employees who upgrade their certification, which we understand to be budgeted for at approximately \$1500/employee, which happens to be the approximate cost of class 3 certification.

However, we also assert that simply holding a class 3 license is not sufficient to safely operate a piece of equipment for which you have little to no hands-on training, which is why we have also proposed training criteria. I understand that when the training program was running (within approximately the last 5 years), 80 hours was the benchmark for hands on training before operating a new piece of machinery solo, so we have used that as our threshold with the addition of "refresher training" if the Employee has not used that equipment within the last year and/or if a newer version of such equipment is purchased.

We believe there may be 11 current employees who operate the Skid Steer, but who do not hold class 3 licenses. We are proposing that they continue to be paid at Operator II/Equipment operator rates until Dec. 31, 2024 (if they did not achieve their class 3 by the year before). While you consider our proposal, it would be helpful to have a current listing of such employees, especially since your estimated number (5) doesn't match with our understanding.

We also have a counter for you regarding the wages for 2021 (employer proposed 1% lump sum, whereas we proposed 1% on the grid). Instead of wages for 2021, we are proposing a \$500 increase to the flexible health

spending account (not currently listed in the CBA, but provided to members at an amount of \$500/yr currently).

The other change on wages is to dovetail the rates for Gravel Checker, which are slightly higher, with those for Equipment Operator I. The skill level required for operating the Skid Steer/Sidewalk machine is much greater than that for the Gravel Checker classification, so it doesn't make sense that the rate should be lower, even marginally so.

We also have a counter for you regarding the 5 continuous days off at Christmas. This continues to be a highly contentious issue regarding which members report a great deal of confusion about the employer's intentions. Ideally, this confusion could be put to rest with clear language in the CBA.

We have tried to capture what we believe are the Employer's estoppel notice intentions, except for the addition of up to 2 additional lieu days. This proposal still represents a substantial savings for the employer compared to past practice, especially when it comes to the calculation of overtime in the week on for TAS employees, when the stats were counted as paid hours in the prior week off, leading to overtime in the week following Christmas.

I would suggest that, in the absence of language such as we are proposing here, the Employer would need to continue that calculation, especially with the addition of the agreed upon language proposed by the employer at 16.05:

16.05

Only hours actually worked, paid vacation (including general holidays) and banked time used to supplement hours are used in accumulation of overtime. Sick leave or other paid or unpaid absences are not included.

As always, I'm available to discuss our proposal with you in more detail. We look forward to your response.

Thank you,

Merryn

Merryn Edwards

phone: 780-930-5287 cell: 780-952-1951 fax: 780-930-3392

toll-free: 1-800-232-7284 email: m.edwards@aupe.org

El respeto al derecho ajeno es la paz. Respect for the rights of others is peace. --Benito Juarez

August 24, 2022 – Sent via email Sept. 9, 2022

Union Proposal for Settlement

THIS AGREEMENT is made this 18th day of April, 2018, A.D. pursuant to the Labour Relations Code of Alberta (2000) as amended.

This Agreement is entered into by and between the Strathcona County, municipal corporation covered by the Collective Agreement (hereinafter referred to as "the County" or "the Employer") and the Alberta Union of Provincial Employees Local 118/009 (hereinafter referred to as "the Union").

Whereas, the above parties are mutually desirous of entering into a Collective Agreement with the intent and purpose of:

- A. Promoting a harmonious relationship;
- B. Providing for conditions of employment, hours of work and rates of pay;
- C. Establishing the means for prompt and equitable settlement of disputes or grievances between the County and Employees represented by the Union, and;
- D. Encouraging efficiency in operation.

Now therefore, the parties hereto mutually agree as follows:

#### ARTICLE 1 – DEFINITIONS

- 1.01 In this Agreement:
  - (a) A word used in the singular applies also in the plural. A word used in the masculine applies also in the feminine.
  - (b) Employee: means a person employed by the Strathcona County in the Public Works Operations, Vehicle and Equipment Services, Transit Services, Materials Management, Water and Wastewater Operations, and as listed in Schedule A.
    - (i) Regular Employees: are Employees who occupy established positions and who have successfully completed the probationary period and work the fully scheduled hours.
    - (ii) Part-time Employees: are Employees who occupy established positions and have successfully completed the probationary period and work less than the fully scheduled hours of Regular Employees.
    - (iii) Temporary Employees: are Employees who are indefinite term Employees or Employees for specific projects, seasonal employment, replacements for Regular Employees who are

on illness leave, General Illness, Long Term Disability, Workers' Compensation, Maternity Leave, Parental Leave and Adoption Leave, or other long term leaves of absence. Indefinite term Employees upon completion of more than fifteen (15) months of continuous service will be deemed to be Regular Employees, except if they are covering for an Employee on Long Term Disability, are moving from one Department to another Department for a temporary or seasonal position, or are covering off consecutive leaves, as referenced above. "Continuous service", for the purpose of this clause, means any period of employment that does not include a break in service of more than thirty (30) calendar days.

- (iv) Apprenticed Employees: are Employees who are registered as Apprentices under the Apprenticeship and Industry Training Act.
- (v) Probationary Employee: means an Employee who during his initial period of employment occupies an established position during a probationary period.
- (c) Grade: means the step assigned to a class within the pay grid.
- (d) Union Steward: means a person who is selected by the members of the Union to act on behalf of those Employees as empowered by the Union.
- (e) Union Representative: means a person authorized by the Union to act on behalf of Employees and includes a Union Steward, elected or appointed officer of the Union, or staff representative.
- (f) Liaison Committee: is the joint Liaison Committee established by the two parties as stipulated within this Agreement.
- (f) Casual Illness: is any illness which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less.
- (g) General Illness: is any illness which causes an Employee to be absent from work for a period of more than three (3) consecutive work days.
- (h) Layoff: includes a reduction in the work force as a result of shortage of work.
- (i) Branch: means
  - (i) Public Works Operations
  - (ii) Vehicle and Equipment Services
  - (iii) Transit Services
  - (iv) Water and Wastewater Services

- (v) Materials Management
- (j) Classification: means each line in Schedule A.
- (k) Seniority: commences at date of hire, and shall include all continuous **and accumulated service** as a temporary Employee. <del>provided there was no break in service greater than thirty (30) days.</del>
- (l) Day: shall mean calendar day.
- (m) Continuous Service: any period of employment that does not include a break in service of more than thirty (30) calendar days.
- (n) Accumulated Service: Total number of regular hours worked, with no break in service of more than six (6) months.

#### ARTICLE 2 – JURISDICTION

- 2.01 During its term, this Agreement shall be applicable to any person employed in the Public Works Operations, Vehicle and Equipment Services, Transit Services, Materials Management and Water and Wastewater Services of Strathcona County, who are described in Schedule A attached hereto, or to positions that fall within the jurisdiction of the Union.
- 2.02 When additional work is required by the Employer, individuals who have paid Union dues to the Alberta Union of Provincial Employees, Local 118, Chapter 009 in the past twelve (12) months, will be called first and, if available, offered the additional work.

#### **ARTICLE 4 - APPLICATION**

- 4.01 Regular Employees shall be granted all the terms and conditions of this Agreement.
- 4.02 Part-time Employees shall be granted all the terms and conditions of this Agreement except those conditions which require a particular length of service, such benefits shall be granted on a pro-rata basis in relation to hours of work of full-time employment.
- 4.03 Apprenticed Employees as defined within Apprenticeship and Industry Training Act shall be granted all the terms and conditions of this Agreement.
- 4.04 Temporary Employees shall be entitled to the terms and conditions of this Agreement, except where Regular Employees are referenced. Seniority for employment periods of less than six (6) months shall not be accumulated.

#### **ARTICLE 6 - UNION RECOGNITION**

6.01 The County recognizes the Union as the sole bargaining agent of the Employees covered by this Agreement.

6.02 No Union Employee shall be required or permitted to make any written or verbal agreement which may be in conflict with the terms of this Collective Agreement.

## 6.03 New Employees

The Employer shall supply all new Employees with a copy of the Collective Agreement.

## 6.04 <u>Monthly Staffing Reports</u>

The County recognizes the need of the Union to have required information concerning Employees eligible for membership in the Union and hereby agrees to submit between the sixteenth (16th) day and the last day of each month, the name, address, and position filled by any new Employees assigned to Public Works Operations, Vehicle and Equipment Services, Materials Management, Water and Wastewater Operations and eligible for membership in the Union.

The Employer recognizes the importance of new Union Employees receiving an orientation to the Union. To ensure this occurs, the Chapter Executive will be given a list of all newly hired Employees and a representative of the Union shall, within thirty (30) calendar days of the Employees' date of hire, make a presentation to the Employee(s) with respect to the structure of the Union as well as the rights, responsibilities and benefits of membership in the Union.

The Employer will generally, provide the Union with one (1) weeks' advance notice of the Employee's scheduled orientation.

#### 6.04 No Discrimination

The parties agree that there shall be no discrimination or coercion exercised or practiced with respect to any Employee for reason of function, membership or legitimate activity in the Union.

The Employer shall exercise its rights in a fair and reasonable manner, nor shall these rights be used in a manner which would deprive any Employee of his employment, except for just cause.

## 6.05 <u>Bulletin Boards</u>

The County recognizes the right and need of the Union to be able to post notices of meetings and other legitimate Union activities at various sites wherein members of the Union are assigned or employed.

#### 6.06 <u>Seniority Lists</u>

County shall maintain a list showing the date upon which each Employee's service commenced. The County shall forthwith provide the Union, upon request, with an up-to-date list and shall, in any event, post an up-to-date list on all bulletin boards

where the Employees work on July 1<sup>st</sup> and January 1<sup>st</sup> in each year.

## 6.07 Union Officers Stewards

The County recognizes the right of the Union to appoint "Employees" as Union Stewards who are to be recognized as official representatives of the Union.

## 6.08 <u>Union Membership</u>

Each person who becomes employed in positions covered by this Agreement shall automatically become a member of the Union as a condition of employment.

## 6.09 <u>Monthly Staffing Reports</u>

- (a) The Employer agrees to remit to the central office of the Union, the amounts equal to the dues that have been deducted from the pay of all Employees by the first working day after the fifteenth (15th) calendar day in the following month. Where an accounting adjustment is necessary to correct an over or under payment of dues, it shall be effected in the succeeding month.
- (b) The County recognizes the need of the Union to have required information concerning Employees eligible for membership in the Union and hereby agrees to submit on a monthly basis between the sixteenth (16th) day and the last day of each month, the following information: the name, address, phone number, status (regular or temporary), Identification number, classification, starting date and position filled by any new Employees In the bargaining unit. assigned to Public Works Operations, Vehicle and Equipment Services, Materials Management, Water and Wastewater Operations and eligible for membership in the Union.

The Employer shall provide the Union with a monthly list identifying each Employee, including new or terminated Employees. The list will include Employee name, address, identification number, starting date, classification, and dues deducted for that calendar year.

(c) The Employer recognizes the importance of new Union Employees receiving an orientation to the Union. To ensure this occurs, the Chapter Executive will be given a list of all newly hired Employees and a representative of the Union shall, within thirty (30) calendar days of the Employees' date of hire, make a presentation to the Employee(s) with respect to the structure of the Union as well as the rights, responsibilities and benefits of membership in the Union.

#### ARTICLE 8 - JOB DESCRIPTIONS CLASSIFICATIONS & RECLASSIFICATIONS

8.01 The Employer shall make the Employee's job description available.

## 8.02 New or Amended Classifications

The County Administration shall give written notice to the Union of the establishment of any new or amended job descriptions and/or salary classification levels (Schedule A). If it is established that a higher salary level is appropriate, the new salary shall be retroactive to the date of the classification was created. Where Union and management representatives through the Liaison Committee are unable to agree within thirty (30) days on either the correct classification of a revised job description or on the compensation level of a new job description, the matter may be advanced to arbitration for the establishment of the compensation for the individuals for which the job description has changed. If a wage change results, the effective date of change shall be determined by the Arbitrator. Both the Employer and the Union will share the cost of the Arbitrator equally. The Arbitrator will be guided by:

- (a) Evidence and submissions of the parties;
- (b) The salary levels in the Collective Agreement;
- (c) The complexity, working conditions and knowledge requirements of the other positions in the Collective Agreement at the time that the new position is established.

However, the Arbitrator shall not be restricted to the existing salary levels.

## 8.02 <u>Classification Reviews</u>

An Employee may at any time bring to management for review, changes that may have occurred to his existing job duties. The Employee/Union, may bring such issues forward to Liaison Committee.

## 8.03 Reclassification

When it is necessary to re-classify Regular Employees, the County may re-classify the affected Employees on the basis of Branch seniority, qualifications and ability.

- (a) An Employee to be re-classified may displace any Employee in a lower classification within that Branch that has less Branch seniority providing that the Employee being reclassified is qualified to perform the duties of that position;
- (b) Employees with special qualifications may be exempt from re-classification.
- 8.04-12.02 In those cases where re-classification according to Clause 12.01 8.03 have occurred and vacancies become available in the affected job classification, those Employees who were previously re-classified downward from that classification, shall be returned to that classification on the basis of their seniority so long as that Employee has maintained the ability and qualifications to be appointed to the position.
- 8.05 12.03 An apprentice who begins and completes their apprenticeship with the Employer

shall be placed on level 2 of the appropriate trades rate upon confirmation of certification.

## ARTICLE 9 - POSTING OF POSITIONS, PROMOTIONS AND TRANSFERS

- 9.01 When a position of a Regular or Temporary nature is to be filled, notice of the vacancy shall be posted as soon as reasonably possible for a minimum of one (1) week. The position can be filled by appointment if it is to be filled for a period of less than ninety (90) calendar days.
- 9.02 If a position with the same job code becomes vacant within six (6) months of the original closing date, the Employer may consider candidates from the original recruitment process, without re-posting. This is not applicable if the original posting was for a temporary position and the vacancy is for a permanent position.
- 9.02 3 Such notice will contain the following information: date of the posting, nature of position, qualifications, required knowledge and education, skills, shift, hours of work, experience, wage rate or range, consistent with the job description.
- 9.03 4 Where there are qualified Union candidates for a posted position, their applications shall be considered first.
- 9.04-5 Where qualified applicants have applied and a suitable candidate has been selected, the Employer shall appoint a candidate to the position within thirty (30) days of the competition closure.
- 9.05 6 Preference in the filling of positions shall be given to Employees of the County over all other persons, where the Employee has the required skills and ability to perform the tasks of the position. When filling positions or making promotions, qualifications, experience, skills, ability and seniority shall be the governing factors. When filling positions, employees of the County shall be given first preference.

## 9.06 7 Trial Period

An Employee who is transferred or promoted to another position within this Agreement shall be confirmed in that other position within three (3) months unless the Employee wishes otherwise or proves unsatisfactory in the new position, in which case the Employee will be returned to his former position. Any other Employee promoted or transferred because of the displacement of positions shall also be returned to his former position. In any event, Employees promoted or transferred shall be paid at the rate applicable to the new position from commencement of employment in the position until returned to the previous position.

(b) In making promotions and filling vacancies, appointments will be made on the basis of qualifications, experience, ability and seniority.

The posting requirements in this Article may be waived by mutual agreement in cases of medical accommodation.

#### **ARTICLE 10 – SENIORITY**

9.08

10.01 For the purpose of Clauses 10.01 to 10.03 10.10, 12.01 to 12.03 12.02, and 15.01 to 15.06 15.05, seniority shall be defined from the date of continuous service employment in any Branch covered by this Agreement.

In accordance with 1.01(k) seniority commences at date of hire, and shall include all continuous and accumulated service as a temporary Employee.

- 10.02(a) Employees who were employed by the County prior to January 1, 1979 and entered in any classification covered by this Agreement on that date shall have their seniority established as the date of continuous service employment with the County.
- 10.023(b) ——Seniority shall be retained and accumulated during absences resulting from illness or an accident where the Employee is entitled to compensation, suspension with or without pay less than sixty-one (61) days, approved leave without pay less than sixty-one (61) days.
- 10.034 (c)—Seniority shall be retained during absences resulting from; layoff (up to eighteen (18) months), suspension with or without pay greater than sixty (60) days, approved leave without pay greater than sixty (60) days, strike or lockout.
- 10.045 Where Employees have been laid off and recalled, they shall retain the seniority held at the date of layoff. The Union shall be notified of all recalls.
- 10.056(d)——An Employee who accepts an internal permanent position outside this Agreement shall retain the seniority accrued to that time for a period of six (6) months.
- Accumulation of seniority shall recommence from the date the Employee returns to a classification covered by this Agreement and their seniority date shall be adjusted by the period of their absence. This retained seniority shall not be effective in exercising any seniority rights until the Employee has been back in the scope of this Agreement for six (6) months.
- An Employee shall lose their seniority and their name shall be removed from the seniority list for any of the following reasons: voluntary resignation, discharge for cause, layoff for more than eighteen (18) months, retirement, or when an Employee has been employed in a position outside this Agreement for a period of six (6) consecutive months or more.

## **ARTICLE 11 - PERIOD OF PROBATION**

11.01 The probationary period for new Employees covered by this agreement in both permanent and temporary positions shall be five hundred (500) hours one thousand (1000) hours worked. , with the exception of Employees appointed to

the positions listed below, whose probationary period will be one thousand (1000) hours worked:

3008	Public Works Technician	3015	<del>Operator I</del>
3009	Public Works Technician II	3016	<del>Operator II</del>
3012	Truck Driver I	3017	Operator III
3013	Truck Driver II	<del>3019</del>	Operations Foreman
3014	Truck Driver III		

3025	Partsman I	3030	<del>V&amp;E Chargehand</del>
3026	Partsman II	3040	<del>V&amp;E Leadhand</del>
3027	Welder	3041	Transit Chargehand
3062	Fleet Services Technician	3042	Transit Leadhand

3021	W&WW Foreman	3052	Utilities Electrician
3038	W&WW Leadhand	3053	Utilities Instrument Technician
3047	Infrastructure Foreman	3051	Ut.Inst Tech/Electrician Dual
<del>3039</del>	<del>Project Foreman</del>	3063	Control Systems Technician
3043	Utility Operator I	3020	W&WW Operator (Collection and Distribution)
3044	Utility Operator II	<del>3049</del>	W&WW Operator (Transmission)
3045	Utility Operator III	3037	Wastewater Infrastructure Operator
3046	Utility Leadhand		

If the Employer identifies performance concerns, written feedback shall be provided to the Employee. Where circumstances warrant an extension, the probationary period may be extended up to an additional two hundred and fifty hours (250) hours upon mutual agreement of the Union and the Employer. may be extended up to an additional 500 hours worked, at the Employer's discretion. The Employee and the Union shall be notified of the extension and the cause for the extension.

An Employee who has previously been employed in a position covered by this agreement will have such previous employment considered part of the probationary period provided that a break in service is one (1) year or less and the Employee is returning to the same job.

- 11.03 At any time, an Employee, categorized as probationary, may be released by the Department Head or the County Administration. Such notice of release will be formalized in writing and will give the reason for the Employee being released as soon as practicable following at the time of the notification of the release.
- 11.04 Employees shall provide at least ten (10) working days written notice of their intention to resign.
- In the event that an Employee terminates his employment with the County without submitting the written notice in compliance with Clause 11.04, the Employee shall not be entitled to receive his regular cheque and/or his final cheque until the next regular distribution of regular wage and salary cheques.
- A Probationary Employee in accordance with Articles 11.01 shall be granted all of the terms and conditions of this Agreement. During the probationary period the County may terminate the employee. The employee or the Union may grieve the termination, but the decision at Level 3 shall be final and binding and the matter shall not be referred to arbitration under the provisions of this Agreement.

however the termination of a Probationary Employee shall not be have access to the application of any grievance procedure for termination of employment as a result of:

- (a) Unsatisfactory probationary performance; or
- (b) The abolishment of the position occupied by the Probationary Employee

#### ARTICLE 12 - RECLASSIFICATION

- 12.01 When it is necessary to re-classify Regular Employees, the County may re-classify the affected Employees on the basis of Branch seniority, qualifications and ability.
  - (a) An Employee to be re-classified may displace any Employee in a lower classification within that Branch that has less Branch seniority providing that the Employee being reclassified is qualified to perform the duties of that position;
  - (b) Employees with special qualifications may be exempt from re-classification.
- 12.02 In those cases where re-classification according to Clause 12.01 have occurred and vacancies become available in the affected job classification, those Employees who were previously re-classified downward from that classification, shall be returned to that classification on the basis of their seniority so long as that Employee has maintained the ability and qualifications to be appointed to the position.
- 12.03 An apprentice who begins and completes their apprenticeship with the Employer

shall be placed on level 2 of the appropriate trades rate upon confirmation of certification.

## **ARTICLE 14 - GRIEVANCES**

## 14.01 <u>Statement of Intent</u>

The Employer and the Union recognize the desirability of resolving differences through joint consultation and **respectful and open** discussion. Both will attempt to resolve differences through **exploring interests and options together informally** means where possible, prior to proceeding to formal **process**.

## 14.02 <u>Definitions and Scope</u>

- (a) A Personal grievance is a dispute regarding:
  - (i) Alleged unjust treatment or discrimination
  - (ii) Alleged unfair working conditions
  - (iii) Any disciplinary action
  - (iv) Any concern regarding the interpretation, application, operation or alleged violation of this Agreement, including any questions as to the arbitral nature of the matter.
- (b) Grievances which involve a dismissal or a suspension will commence at level II and shall be initiated in writing within fourteen (14) days of the date when the Employee first became aware of the disciplinary action.
- (c) A Policy Grievance is a dispute between the parties which due to its nature is not properly subject of a Personal Grievance and concerns the interpretation, application, operation or alleged violation of this Agreement, including any questions as to the arbitral nature of the matter.
- (d) Formal personal grievances shall be initiated within fourteen (14) calendar days from the date of the incident prompting the grievance. Policy grievances shall be initiated either by the Union or the County in writing, within thirty (30) calendar days from the day that the matter giving rise to the grievance came to the attention of the Designated Officer of the Union or Official of the Employer.
- (d) Formal grievances shall be in writing (email will suffice) and must state the general nature of the grievance, identify the specific clauses in the collective agreement that are being violated, and provide details with respect to the redress or remedy being sought.
- (f) "Days" means calendar days.
- (g) A Union member submitting a grievance has the right to be assisted by an attending Union Representative or Union Steward.

#### 14.03 Grievance Process

A Personal Grievance shall be initiated by the Employee as follows:

#### Level I

The grievance shall be taken up with the Employee's manager or designate. an Employee wishes to pursue a Personal Grievance, the Employee and/or Union Representative shall advise the Employee's manager or designate that they wish to initiate a Level I grievance. The Employee and/or the Union representative will discuss the subject of the proposed grievance with the Employee's manager or designate within fourteen (14) calendar days of the receipt of the grievance. date upon which the subject of the grievance, in an attempt to resolve the matter. The manager or designate first level classified supervisor will provide a response to the Employee and/or Union Representative within seven (7) calendar days from when the proposed grievance was discussed.

#### Level II

If the Employee continues to be dissatisfied, he should submit the grievance in writing within fourteen (14) days of receiving the response from Level I, to the Department Director. The grievance shall state the general nature of the grievance, the Articles in the Collective Agreement alleged to have been breached and the redress or the remedy sought. The Director, in consultation with Human Resources, will hear the matter and will respond in writing to the Grievor within seven (7) calendar days.

## Level III

If the Union does not accept the decision of the Department Director, they may request in writing within seven (7) days of receiving the Department Directors decision, that the Chief Commissioner review the grievance. The Chief Commissioner shall hear the matter within ten (10) days and will advise the Union of his decision in writing within fourteen (14) days of the hearing;

The Chief Commissioner may designate an Associate Commissioner to hear and decide on Level III grievances;

If the grievance is not settled as a result of the Chief Commissioner's decision, the Union or the Employer may refer the matter to arbitration pursuant to the Labour Relations Code.

#### 14.04 Arbitration Board

In the event the matter being grieved is presented to an Arbitration Board, each Party shall bear the expense of its respective nominee and shall bear one-half (1/2) of the expenses of the Chairman of the Arbitration Board.

- 14.05 The Parties may mutually agree to have the matter heard by a single Arbitrator.
- 14.06 The Arbitration Board by its decision shall not alter, amend or otherwise change

the terms of this Collective Agreement.

14.07 All Union members submitting a grievance have the right to be assisted by an attending Union Representative.

## 14.08 <u>Procedure and Time Limits</u>

Extension of Time Limits: Time limitations set out in the above grievance procedure may be extended by mutual agreement between the parties and must be in writing.

Time Limits: Time limits and procedures contained in this grievance procedure are mandatory. Failure to pursue a grievance within the prescribed time limits and in accordance with the prescribed procedures shall result in abandonment of the grievance and all rights and recourse to the grievor.

Failure to reply to a grievance in a timely fashion shall advance the grievance to the next level. Grievances so advanced shall be subject to time limits as if a reply had been made on the last allowable day of the preceding level in the procedure.

14.09 Policy Grievances may be initiated either by the Union or the County in the following manner:

- (a) Policy Grievances shall be initiated either by the Union or the County in writing, within thirty (30) calendar days from the day that the matter giving rise to the grievance came to the attention of the Designated Officer of the Union or Official of the Employer. The written notice of the grievance shall be submitted to the President of the Union or the Director, Human Resources of Strathcona County and shall stipulate the general particulars of the grievance together with reference to the allegedly violated clause of the Agreement;
- (b) Grievances submitted in accordance with Clause 14.09 shall be referred to a Grievance Committee which shall endeavour to resolve the difference. This Committee will be composed of two representatives of the County and two members of the Union appointed by the Executive. This Committee shall meet within fifteen (15) calendar days of the submission of the written notice by either party;
- (c) After the parties have met in accordance with the preceding clause, each party will give any decision made in writing to the other party within fifteen (15) calendar days. In the event that the two parties reach a mutually acceptable agreement concerning the matter in question, a written and signed notice of mutual agreement will supersede the aforementioned notice of decision;
- (d) In the event that the parties do not reach a mutually acceptable agreement, or that either party does not accept the written notice of decision of the opposite party within thirty (30) calendar days of the Chief Commissioner's decision as per Clause 14.03 either party may notify the other party in writing of its intention to submit the grievance to arbitration pursuant to

#### ARTICLE 16 - HOURS OF WORK AND OVERTIME

The prescribed regular hours of work for an Employee, other than those referred to in Clause 16.04-5 following, shall be eight (8) hours in any one (1) day or forty (40) hours in any one (1) calendar week or five (5) working day period. Hours worked in excess of the above shall be calculated as overtime hours and shall be paid accordingly. Notwithstanding the above, when an established schedule provides a pattern that is irregular, the forty (40) hours in one (1) calendar week or five (5) working day period shall mean average over a two (2) week period.

## 16.02 Overtime

The overtime pay rates shall be calculated at the rate of two (2.0) times the regular pay rate for all hours worked beyond eight (8) hours in a day.

- An Employee may bank overtime hours at the rate it is earned; up to the maximum of forty (40) regular hours within a calendar year. An Employee may request to be paid out for banked hours up to the time equivalent of forty (40) regular hours per calendar year. Banked hours may be taken as time off by Employees after receiving written authorization from the County designated official who shall consider the wishes and application of the Employee. Up to forty (40) banked regular hours may be carried forward into the next calendar year. Banked hours cannot exceed the time equivalent of eighty (80) regular hours at any one time.
- 16.04 For those Employees who are engaged in road construction, road maintenance or the removal of snow from roads or any of those undertakings, for the purpose of calculating overtime, their regular day shall consist of a twenty-four (24) hour period beginning at midnight. The maximum number of hours of work at regular rates of pay shall be those prescribed or allowed under the Employment Standards Code and Regulations thereunder, but shall not exceed:
  - (a) Ninety (90) hours bi-weekly, except where permitted under the Employment Standard Code;
  - (b) Eighty (80) hours bi-weekly during the period November 01 to April 30, inclusive:
- 16.05 Only hours actually worked, paid vacation (including general holidays) and banked time used to supplement hours are used in accumulation of overtime. Sick leave or other paid or unpaid absences are not included.
- 16.06 For those Employees specified in Clause 16.04 from May 1 to October 31, overtime rates shall be calculated at the rate of two (2.0) times the regular pay rate for all hours worked beyond nine (9) hours in a day. This article only comes into effect when actually scheduled to work nine (9) hours a day.

## 16.06 <u>Posting of Shift Schedules</u>

Barring unforeseen circumstances and emergency situations, shift schedules shall be posted no less than thirty (30) days before the expiry of the current schedule.

## 16.07 <u>Shift Changes</u>

- (a) When an Employee is required to change from one shift to another shift, the Employer will provide at least twenty-four (24) hours written notice in advance of the start time of the originally scheduled shift and ten (10) hours off between shifts.
- (b) If the twenty-four (24) hours written notice and ten (10) hours off between shifts is not provided to the Employee, the Employee will be paid at the rate of two (2.0) times the regular rate of pay for the hours of the first shift worked.
- (c) If the Employee is on regular scheduled days off when the twenty-four (24) hours written notice is posted, the supervisor may contact the Employee directly to provide verbal twenty-four (24) hours notice change in shift.
- 16.08 An Employee shall be paid for or expected to work for a minimum of three (3) hours if the regular shift is cancelled or rescheduled and the Employee was not notified of such cancellation or rescheduling on or before the day twelve (12) hours prior to the cancelled or rescheduled shift.
- 16.09 When an Employee calls in ill for a work period affected by the preceding Clause 16.08, that Employee will be paid only for the minimum of three (3) hours that he would have been paid had he reported for work in the regular fashion. This clause shall not apply in cases in which the Employee provides proof of attendance at either medical, dental or optical appointments. In such cases, the Employee will be paid for either the half-day or full-day as appropriate to the circumstances and as provided in Clause 21.03 and 21.04.

## 16.10 Early Reporting

- (a) When an Employee is requested prior to the conclusion of a shift, to report or work more than two (2) or more hours earlier than his regularly scheduled shift, that Employee may (subject to operational requirements) be given a choice of leaving work prior to the end of their regularly scheduled shift or being paid the regular overtime rate for those hours worked prior to the time scheduled for the commencement of the regular shift. The total of the early reporting hours worked and regular shift hours worked will not be less than the number of regularly scheduled hours. Such early reporting arrangement may continue for two consecutive shifts.
  - If requested to report early on the second shift the early reporting hours will be at the overtime rate. If requested for a third consecutive shift, Clause 16.07 will apply.
- (b) When an Employee is requested prior to the conclusion of a shift to report up to two (2) hours earlier or up to two (2) hours later for the next regular shift, that Employee shall not receive the overtime rate for the earlier or

later reporting period. The total of the earlier reporting hours worked and regular shift hours worked will not be less than the number of regularly scheduled hours. If requested to report earlier or later on the second shift the earlier or later reporting hours will be at the overtime rate. If requested for a third consecutive shift, Clause 16.07 will apply.

## 16.11 Attendance

When an Employee is unable to report to work, the Employee shall follow the established protocol for their department to report such absence, and at a minimum shall report the absence one (1) hour prior to the start of the shift. within one hour prior to the scheduled shift start time. Failure to comply with this requirement shall result in loss of pay for the shift, and may result in discipline unless the Employee provides a reasonable explanation.

## 16.121 Work Break

All Employees shall be entitled to one fifteen (15) minute work break in each half day worked, or in any overtime work period in excess of two (2) hours. Such work break shall be taken at the work site and the time of such break shall be approved by the immediate supervisor of each Employee. The work break shall be considered part of a work hour for the purposes of the preceding paragraphs.

## 16.13 Sand Shift Workers and Sidewalk Crew

Notwithstanding Clause 16.01, when the needs of the operations require it, Sand Shift Workers and Sidewalk Crew may be required to work a modified shift schedule of up to twelve (12) hours. Modified hours of work shall commence annually on the first full pay period commencing on or after October 15 and shall continue until the first full pay period before or after April 15 based on operational need.

- (a) Overtime will be paid to Sand Shift Workers and Sidewalk Crew who exceed 80 hours of work in a bi-weekly pay period.
- (b) Employees working modified hours of work will have all benefits and entitlements which are expressed in terms of daily or weekly entitlement converted to produce the equivalent hours of benefits and entitlements as they would have had if the hours of work had not been modified. This will result in no loss or gain in Employee benefits and entitlements.

## ARTICLE 17 - SHIFT AND WEEKEND PREMIUMS

17.01 (a) A shift premium of one dollar and forty cents (\$1.40) one dollar and ninety cents (\$1.90) per hour shall be paid to all Union Employees for those hours worked between seventeen hundred (1700) hours and zero seven hundred

- (0700) hours Monday to Friday and for all hours worked on Saturday and Sunday and holidays.
- (b) At no time shall shift premium be included with the Employee's regular rate of pay for purposes of computing overtime payments, other premium payments, or any Employee benefits.

## ARTICLE 20 - RELIEF AND ACTING INCUMBENT PAY

## 20.01 Relief & Acting Pay

Any Employee who works in a classification (within the Bargaining Unit) with a higher rate of pay in any one (1) shift shall receive the higher rate of pay for each hour worked in the higher classification. If this time period exceeds ninety (90) days, effective the ninety-first (91st) day, the Employee will receive the higher rate of pay for all approved vacation, general holiday, illness, bereavement, and compassionate leave while working in the higher classification.

- 20.02 Notwithstanding the premium in Clause 20.09, An Employee appointed to Leadhand or Project Foreman shall not receive an hourly rate of pay that is less than the "County Department Union Employees" that he is responsible for.
- 20.03 Certified Technicians appointed to a seasonal Field Certified Technician assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- 20.04 Vehicle Serviceman II appointed to a seasonal Field Vehicle Serviceman II assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- The qualification or eligibility of an Employee to relieve or be assigned to a higher classification will be the responsibility of management.
- 20.03 An Employee who has become incapacitated by injury or illness may be employed in other work which he can do. Such Employee may not displace an Employee with more seniority.
- 20.04 An Employee who refuses without good and satisfactory reason, to accept the alternate classification or position, may be suspended immediately. The supervisor issuing the suspension will report the suspension to the Department Head (Human Resources) as soon as possible who will notify the Union Executive.
- 20.05 Any Employee affected by the preceding Clauses 20.03 and 20.04 and not satisfied with the action recommended in accordance with Clause 20.04, may appeal the reclassification or suspension in keeping with Clause 14.03 of this Agreement.
- 20.06 Should an Employee who has completed the probationary period be unable to perform his usual duties by reason of the loss or suspension of his motor vehicle operator's license or a similar happening, that Employee may be placed in some

other suitable position until the disability ends or is removed. The wages paid during the period of such disability shall be wages appropriate to the position in which the Employee is placed.

20.07 Notwithstanding the above clause, within fourteen (14) days of the suspension, an Employee may be allowed to book unused holidays at the wage prevailing prior to suspension of his operator's license.

## **Acting Incumbent**

- 20.08 Any member of the Union may be appointed for one (1) or more days as a Classified supervisor and such appointment will not cause the member to be disqualified from the Union.
- 20.09 If any Employee is appointed temporary Classified Supervisor, they shall receive a premium of two dollars and twenty-five cents (\$2.25) per hour in addition to their regular wage rate for one (1) complete shift or longer.
- 20.10 Any member of the Union may be appointed for a period of time to the position of temporary Classified Supervisor and such appointment will not cause the member to be disqualified from the Union.
- 20.11 Notwithstanding the premium in Clause 20.09, an Employee appointed to Leadhand or Project Foreman shall not receive an hourly rate of pay that is less than the "County Department Union Employees" that he is responsible for.
- 20.12 Certified Technicians appointed to a seasonal Field Certified Technician assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.
- 20.13 Vehicle Serviceman II appointed to a seasonal Field Vehicle Serviceman II assignment shall be paid a premium of one dollar and ten cents (\$1.10) per hour for one (1) complete shift or longer.

## ARTICLE 21 - PROVISION FOR ILLNESS AND NORMAL ABSENCE

- 21.01 Sick leave benefits as herein stated are sponsored by the County to protect the Employee in the event of an unavoidable illness or injury not covered by the Alberta Workers' Compensation Act.
- 21.02 When an Employee is unable to report to work, the Employee shall follow the established protocol for their department to report such absence within one hour prior to the scheduled shift start time. Failure to comply with this requirement shall result in loss of pay for the shift, and may result in discipline unless the Employee provides a reasonable explanation. Moved to Article 16
- 21.03 A Regular Employee, upon hire, shall be entitled to a maximum of eighty

- (80) hours ten (10) working days per calendar year of leave with pay for any casual illness, which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less. Up to sixteen (16) hours two (2) days of unused casual illness leave per calendar year shall be carried over and may be used in the next calendar year. The maximum paid casual illness leave in a calendar year is ninety-six (96) hours twelve (12) days including the carry over.
- A Temporary Employee after the completion of the standard probationary period, shall be entitled to a maximum of thirty-two (32) hours four (4) working days per calendar year of leave with pay for any casual illness, which may cause the Employee to be absent from work for a period of three (3) consecutive work days or less. Temporary Employees with more than six (6) months of service in the current calendar year shall be entitled to an additional sixteen (16) hours two (2) working days in the current calendar year, of leave with pay for any casual illness. Up to two (2) days of allotment can be used to attend to illness of immediate family members.
- 21.05 Each day or portion of a day of absence due to casual illness used within a calendar year, shall be deducted from the remaining casual illness leave entitlement for that calendar year.
- NEW Where an employee terminates and is rehired in the same calendar year, the casual illness allotment will be the amount that they had in their available bank at the time of departure.
- In any case, an Employee may be required to provide satisfactory proof of illness upon return to work, where it can be shown that reasonable doubt exists in respect to the purpose of an absence claimed to be due to illness. The employee will be provided the reasons for the request. Such proof may take the form of a medical certificate signed by a qualified physician, dental practitioner or other professional as deemed appropriate by the Disability Management Supervisor/Human Resources.
- 21.07 Upon approval (which will not be unreasonably denied), time off of two (2) hours or less for doctor appointments will result in no loss of casual illness or regular time. The Employer may require the Employee to submit proof of attendance at a medical, dental or optical appointment when time off work is granted to attend such appointments.
- 21.08 Each Regular Employee will be entitled to leave of absence with pay for General Illness in accord with the following scale. General Illness is defined as "any illness which causes an Employee to be absent from work for a period of more than three (3) consecutive work days".

To obtain General Illness benefits resulting from the inability of the Employee to perform the duties of his position, a proper medical certificate(s) signed by a qualified physician, dental practitioner or other professional as deemed appropriate by the Disability Management Supervisor/Human Resources is required.

## 21.09 <u>Family Illness Leave</u>

A Regular Employee shall be entitled to a maximum of twenty-four (24) hours three (3) working days per calendar year with pay to attend to illness of immediate family members as per defined in Clause 22.06.

- 21.10 A Temporary Employee may use up to sixteen (16) hours two (2) days of their casual illness allotment, as per Clause 21.04 can be used to attend to illness of immediate family members.
- 21.<del>09</del>11 Effective the calendar year following ratification, Sick leave entitlement for illness longer than three (3) working days shall be as follows:
  - (a) After ninety (90) days three (3) months five hundred (500) hours of continuous employment the Employee shall receive one hundred sixty (160) hours twenty (20) working days at the base rate of pay;
  - (b) On the first day worked, in the next calendar year sick leave entitlement shall be increased to a maximum **two hundred eighty** (280) hours thirty-five (35) working days at the base rate of pay
  - (c) On the first day worked, in subsequent consecutive calendar years sick leave entitlement shall be increased to a maximum of **five hundred twenty (520) hours** sixty-five (65) working days at the base rate of pay, plus any unused casual illness up to a maximum of **six hundred (600) hours** seventy five (75) working days.
- 21.<del>10</del>12 When a day designated as a Paid Holiday under Clause 23.01 falls within a period of General Illness, it will be considered as a paid holiday.
- 21.<del>12</del>13 Transfers, promotions, demotions and/or reclassification shall not affect sick leave entitlement of any Employee.

## 21.<del>11</del>14 Workers Compensation

- a) An Employee is not eligible to receive sick leave benefits if the absence is due to an injury while in the employ of any other Employer that is covered by Workers' Compensation Benefits.
- b) Employee's will be eligible to receive benefits in accordance with the County's policies.

- 21.12 Transfers, promotions, demotions and/or reclassification shall not affect sick leave entitlement of any Employee.
- 21.<del>13</del>15 Sick leave entitlement will not be affected while the Employee is in receipt of Alberta Workers' Compensation benefits.
- 21.4416 Regular Employees who are eligible to receive Workers' Compensation benefits shall receive full pay from the County for a period not to exceed six (6) months provided that all Workers' Compensation payments are assigned to the County for that period.
- 21.4517 Expenses for medical certificates will be the responsibility of the employee, including medical certificates required to support eligibility for casual or general illness. Where the Employer has requested the employee to attend a medical examination and/or obtain additional medical documentation, the Employer shall reimburse the employee for the cost of the documentation (if any).
- 21.4618 An Employee who has become incapacitated by injury or illness may be employed in other work which he can do. Such Employee may not displace an Employee with more seniority.
- An Employee who refuses without good and satisfactory reason, to accept the alternate classification or position, may be suspended immediately. The supervisor issuing the suspension will report the suspension to the Department Head (Human Resources) as soon as possible who will notify the Union Executive.
- 21.4820 Any Employee affected by the preceding Clauses 21.176 and 21.187 and not satisfied with the action recommended in accordance with Clause 20.04, may appeal the re-classification or suspension in keeping with Clause 14.03 of this Agreement.
- 21.1921 Should an Employee who has completed the probationary period be unable to perform his usual duties by reason of the loss or suspension of his motor vehicle operator's license or a similar happening, that Employee may be placed in some other suitable position until the disability ends or is removed. The wages paid during the period of such disability shall be wages appropriate to the position in which the Employee is placed.
- 21.<del>2022</del> Notwithstanding the above clause, within fourteen (14) days of the suspension *of* their operator's license, an Employee may be allowed to book unused holidays at the wage prevailing prior to suspension of his operator's license.

#### ARTICLE 22 -LEAVES OF ABSENCE

22.01 Other Leaves of Absence, without pay, may be granted at the sole discretion of the County, and in accordance with the Alberta Employment Standards Code, upon

written application by the affected Employee at least seven (7) days prior to the beginning date of the applied for Leave of Absence. To be eligible for such leave, an Employee shall have been employed with the County for ninety (90) days of continuous employment, with the exception of reservist leave which is twenty-six (26) consecutive weeks, and must have his application endorsed or approved by the Department Head. The application shall include the reason for the request as well as the dates of the requested Leave of Absence. Such approval should not be unreasonably withheld.

- In unusual circumstances or by mutual agreement, the requirement stated in Clause 22.01 for application to be made seven (7) days prior to the commencement of the requested leave of absence may be waived. Application for such waiver is to be made directly to the Department Head.
- In the case of an Employee who takes a leave of absence under the provisions of Clause 22.01, the benefits and conditions to which that Employee is entitled as a result of length of service in the County shall not be adversely affected as a result of having been granted such leave of absence.
- 22.04 An Employee on such approved leave may not engage in other employment for gain while on such leave without the express written consent of the County.

## 22.05 Court and Jury Duty

Leave of Absence shall be provided to an Employee who is required to serve as a juror, or is subpoenaed as a witness in a private capacity in any court proceeding. The Employee shall receive their regular rate of pay and shall remit to the Employer any allowance they may receive for such duty or service.

If the Employee is subpoenaed as a witness to appear in court in their official capacity to give evidence or to produce Strathcona County records, the Employee shall receive their regular rate of pay during regular working hours and is eligible to receive overtime pay to appear during non-work hours. The Employee shall remit to the Employer any allowance they may receive for such duty or service.

The Employee shall present proof of such duty or service and the amount of remuneration or allowance received in all cases.

Employees required to appear in court in a capacity other than subpoenaed as a witness or jurist will be granted leave without pay.

#### 22.06 Bereavement Leave

An Employee shall be granted Bereavement Leave of Absence with pay for not more than five (5) consecutive days because of the death of spouse (includes common-law), child (includes step, foster child or ward), parent, (includes step or foster parent), brother, sister, parent of spouse, guardian, son-in-law, daughter-in-law, grandchild or any other relative who is a member of the Employee's household, or the spouse of any of the foregoing. In addition to the above, the Employer may grant up to one (1) day travel to attend the funeral if the travel is

five hundred (500) kilometres or more away.

- 22.07 Leave of absence for one (1) full day, with pay, shall be granted at the time of to attend the funeral or death of a grand-parent, grandparent-in-law, brother-in-law or sister-in-law, aunt, uncle, niece or nephew, if the funeral is on a work day.
- 22.08 Upon written application, additional Compassionate or Bereavement Leave of Absence with pay, due to exceptional circumstances, may be granted an Employee at the discretion of the Chief Commissioner.
- One (1) day of leave may be granted without loss of pay or benefits to attend as a pallbearer or mourner for a deceased Employee. Where the family of a deceased Employee requests pallbearers from the Union, the County may grant the necessary leave with pay for up to six (6) pallbearers.

## 22.1122.10 Family Illness Leave

A Regular Employee shall be entitled to a maximum of twenty-four (24) hours three (3) working days per calendar year with pay to attend to illness of immediate family members as per defined in Clause 22.06.

22.11 A Temporary Employee may use up to sixteen (16) hours two (2) days of their casual illness allotment, as per Clause 21.04 can be used to attend to illness of immediate family members.

# 22.10 <u>Compassionate Leave</u>

An Employee shall be granted Compassionate Leave of Absence with pay for not more than five (5) consecutive days because of the critical illness of a member of his immediate family as per defined in Clause 22.06 or any other relative who is a member of the Employee's household. The Employee shall provide the County with a letter from the attending physician concerning the fact of the critical illness.

Any additional Compassionate Care Leave shall be provided in accordance with *Alberta Employment Standards Code*.

- 22.11 A Regular Employee shall be entitled to a maximum of three (3) working days per calendar year with pay to attend to illness of immediate family members as per Clause 22.06.
- 22.12 One (1) day leave with pay will be allowed for an Employee to be present at the birth or adoption proceeding of an Employee's child, providing such occurrence is on a scheduled work day.

#### 22.<del>13</del>11 Maternity Leave

Maternity Leave will be granted to Employees in accordance with the Alberta Employment Standards Code.

(a) Employees must have been employed with Strathcona County for ninety (90) days of continuous employment.

- (b) The maternity leave is not more than sixteen (16) weeks starting at any time during the thirteen (13) weeks immediately before the estimated date of delivery.
- (c) The Employee must give six (6) weeks' written notice of the date she intends to begin leave and four (4) weeks' notice of the date she intends to return to work.
- (d) If an Employee's pregnancy results in a miscarriage or stillbirth within sixteen (16) weeks of their due date, they are entitled to sixteen (16) weeks of maternity leave.

## 22.124 <u>Health Related Portion of the Maternity Leave</u>

During the one-week Employment Insurance waiting period, Strathcona County will top-up salary to ninety-five percent (95%) of the Employee's regular earnings. Verification that Employees are in receipt of employment insurance benefits is necessary to provide the top-up benefit. A medical certificate must be provided to Human Resources, confirming the date the child was born and the duration of the leave that the physician indicates is health related. During the health-related portion of the maternity leave Strathcona County will top-up salary to ninety-five percent (95%) of the Employees regular earnings. The benefit coverage that the Employee has at the time of her leave will continue during the health related portion of the absence. At the end of this period, parental leave will commence, and benefit coverage will continue if the Employee pays both Employer and Employee portions of the cost of the premiums.

When a pregnancy interferes with the performance of duties or the Employer has a concern for the Employee's health, the Employer may reassign the Employee's duties at no loss of pay, or by notice in writing to the Employee, require them to commence sick leave under Clause 21.08.

## 22.1416 Parental Leave

Employees who have been employed with Strathcona County for ninety (90) days of continuous employment are entitled to parental leave without pay for a maximum of sixty-two (62) weeks. The Employee must give at least six (6) weeks written notice of the date they intend to begin parental leave and four (4) weeks' notice of the date they intend to return to work. During that time they may receive benefit coverage by paying the Employee and Employer portions of the cost of the premiums.

22-1517 Employees wishing to return to work prior to the agreed return date may do so with two (2) weeks' notice.

## 22.1618 Adoption Leave

One (1) day leave with pay will be allowed for an Employee to be present at the

birth or adoption proceeding of an Employee's child, providing such occurrence is on a scheduled workday.

The Employer will grant adoption leave to Employees who are adopting a child. The Employee must have been employed with Strathcona County for ninety (90) days of continuous employment. Written notice of leave must be provided to the Employer at least two (2) weeks before Employees can reasonably expect to obtain custody of the child.

- 22.1719 If Employees cannot provide at least two (2) weeks written notice, they must notify the Employer immediately after receiving notice of the adoption.
- Adoption leave consists of not more than sixty-two (62) consecutive weeks of leave without pay within seventy-eight (78) weeks after the child is placed with the adoptive parents. During that time they may receive benefit coverage by paying the Employee and Employer portions of the cost of the premiums.
- 22.1921 Only one (1) parent of a child referred to in Clause 22.20 will be granted adoption leave under this clause.

## 22.2022 Shared Leave

If both parents are Employees of Strathcona County parental or adoption leave may be taken wholly by one of the Employees or be shared by both. Only one (1) parent will be granted parental or adoption leave at a time.

# 22.2123 Anniversary Date

Employees who have been on maternity, parental or adoption leave will have their anniversary date adjusted for evaluation and increment purposes only. They will not earn vacation leave during this period, the exception being for maternity health related reasons.

### 22.2224 Additional Leave

Leave of absence without pay beyond the approved maternity, parental or adoption leave may be granted upon written application up to a combined maximum of twelve (12) months at the discretion of the Employer.

### 22.23<del>25</del> Domestic Violence Leave

Domestic violence leave shall be granted in accordance with the *Alberta Employment Standards Code*.

# 22.**24**26 <u>Military Leave</u>

Military leave shall be granted in accordance with the *Alberta Employment Standards Code*.

# 22.25<del>27</del> <u>Citizenship Ceremony Leave</u>

Citizenship Ceremony Leave shall be granted in accordance with the Alberta

## Employment Standards Code.

#### **ARTICLE 23 - HOLIDAYS**

23.01 Employees shall be entitled to one (1) day's paid leave for each of the following specified or general holidays:

New Year's Day

Labour Day

Family Day

Thanksgiving Day

Good Friday

Remembrance Day

Easter Monday

Christmas Day

Victoria Day

Christmas Floater

Canada Day

Boxing Day

1st Monday in August

Truth and Reconciliation Day

And any other day designated as a General Holiday by the Lieutenant Governor in Council under the Employment Standards Code.

23.02

Employees shall receive the recognized specified or general holidays for which they are eligible, with pay; or other days with pay in lieu of such holidays, or pay in lieu as approved by the County, providing they are available for work immediately preceding and following the days or an otherwise approved leave of absence, or they provide a medical certificate attesting to their illness on the day prior or following the general holiday.

23.03

The provisions for pay in lieu of such specified or general holidays referred to in Clauses 23.01 and 23.02 shall be based on the number of regular hours which would have been worked had that day been a regular working day.

23.04

In the event a specified or general holiday falls on an Employee's regular day off, an additional day with pay shall be granted in lieu of that day.

23.05

If any of the above specified or general holidays should occur during an Employee's vacation, such vacation shall be extended by one day or an additional day with pay shall be granted in lieu of that day.

23.06

Temporary Employees, upon hire, shall be eligible for specified or general holidays as per Clause 23.02.

23.XX

Employees working on a paid holiday shall be paid double time (2X) for all hours worked plus holiday pay.

23.07

For Employees who are normally scheduled to work Monday to Friday, the The Christmas Floater holiday shall be observed to give five (5) consecutive days off including the weekend as follows:

- (a) On December 24 when Christmas Day falls on a Tuesday, a Thursday, a Friday or Saturday;
- (b) On December 27 when Christmas Day falls on a Monday or a Wednesday;
- (c) On December 28 when Christmas Day falls on a Sunday.

# 23.08 Employees scheduled on continuous operations shall receive:

- If some or all of the five consecutive days falls described in 23.07 falls on day(s) of rest, Employees shall receive up to five (5) paid days in lieu instead of Christmas Day, Boxing Day and the Christmas Floater as paid holidays. Such days in lieu shall be scheduled to be taken as paid time off at a mutually agreeable time.
- ii) If an Employee works on any of Christmas Day, Boxing Day, or the Christmas Floater, they shall be compensated in accordance with 23.XX (double time (2X) on hours worked, plus holiday pay.

### **ARTICLE 24 – VACATION LEAVE**

- 24.01 Annual vacation entitlement with pay shall be as follows:
  - (a) An Employee who has completed less than six (6) years of service shall earn vacation leave at the rate of four point sixty-two (4.62) hours per bi-weekly period of continuous service [fifteen (15) days per year] or five point two (5.20) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
  - (b) An Employee who has completed six (6) years but less than fourteen (14) years of service shall earn vacation leave at the rate of six point fifteen (6.15) hours per bi-weekly period of continuous service [twenty (20) days per year] or six point ninety-two (6.92) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
  - (c) An Employee who has completed fourteen (14) years of service but less than twenty two (22) years shall earn vacation leave at the rate of seven point seven (7.70) hours per bi-weekly period of continuous service [twenty-five (25) days per year] or eight point sixty-seven (8.67) hours per bi-weekly period of continuous service if based on the forty-five (45) hour work week;
  - (d) An Employee who has completed twenty two (22) or more years of service shall earn vacation leave at the rate of nine point twenty-three (9.23) hours per bi-weekly period of continuous service [thirty (30) days per year] or ten point thirty-nine (10.39) hours per bi-weekly period of continuous service

if based on the forty-five (45) hour work week.

24.02 If Employees become ill on an annual vacation leave for more than three (3) consecutive days paid vacation days, the applicable days will be deemed general illness, and they will have the applicable number of vacation days reinstated upon application and provision of a medical certificate.

24.03 The number of days of vacation leave which may be accumulated by an Employee is limited to the number he can earn in eighteen (18) months of continuous service.

The County will automatically pay out any excess vacation leave over the maximum accumulation at the beginning of each December.

When vacation pay is paid out rather than being taken as vacation, the payout shall be based on the Employee's rate of pay at the time of payout.

24.04 Vacation leave may be taken in the calendar year it is earned. However, vacation leave may not be taken until it has been earned. The vacation leave may be granted as follows:

- (a) In one continuous period;
- (b) As single days off or some other combination;
- (c) If an Employee earns vacation at a rate of four point sixty-two (4.62) hours per bi-weekly period of continuous service, one (1) week vacation leave may be taken as single days off or some other combination provided the total does not exceed five work days in any calendar year;
- (d) If an Employee earns vacation at a rate of six point fifteen (6.15) hours per bi-weekly period of continuous service or seven point seven (7.70) hours per bi-weekly period of continuous service, two (2) weeks' vacation leave may be taken as single days off or some other combination provided the total does not exceed ten (10) work days in any calendar year;
- (e) If an Employee earns vacation at a rate of nine point twenty-three (9.23) hours per bi-weekly period of continuous service, three (3) weeks' vacation leave may be taken as single days off or some other combination provided the total does not exceed fifteen (15) work days in any calendar year.

In addition to the automatic payout of excess vacation referred to in 24.03, an Employee may be permitted to receive a cash payment in lieu of vacation leave, provided any and all obligations have been met in accordance with Employment Standards legislation and the Employee has received written approval from the Department Head.

Temporary Employees shall have vacation pay added to their bi-weekly pay cheques and shall be entitled to unpaid vacation leave in accordance with the Employment Standards Code.

24.06 An Employee who does not complete one (1) full year of service shall be entitled

to vacation pay in accordance with the Employment Standards Code and regulations thereunder.

- (a) Temporary Employees will receive vacation pay at four percent (4%) of their gross regular earnings during their first four thousand (4000) regular standard hours.
- (b) Temporary Employees will receive vacation pay at six percent (6%) of their gross regular earnings after completion of four thousand (4000) regular standard hours.
- (c) If a Temporary Employee has a break in service greater than eighteen (18) months then vacation pay shall be paid at four percent (4%) of their gross regular earnings as per Clause 24.076 a) above.
- 24.07 Regular vacation leave shall be taken by Employees after receiving written authorization from the County or its designated official who shall consider the wishes and application of the Employee. Length of regular service Seniority with the County shall be the determining factor in resolving any dispute or conflict regarding preference for the time at which vacations are to be taken. Changes in vacation scheduling shall be authorized in writing.
- 24.08 Vacation hours will be paid out at the Employee's base rate in place at the time the vacation is taken.

## **Summer Vacation Schedules**

### 24.09 Summer Vacation Schedules

Employees must submit summer vacation applications to the Director or his designate by March 15th of each year. The Employer will post schedules, as approved by the Director, by May 1st of each year.

#### **ARTICLE 25 - BENEFIT AND PENSION PLANS**

### 25.01 Benefit Plans

The County shall contribute on behalf of all participating Employees in group insurance and benefit plans, which provide minimum benefit levels agreed upon by the County and the Union at rates outlined in the following clauses:

- (a) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Strathcona County Benefit Plan Life, Accidental Death and Dismemberment and Long-Term Disability Insurance. All members of the Union hired after July 1, 1978, shall be members of the said plan as a condition of employment;
- (b) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Extended Health Care option of

- the Strathcona County Employee Benefit Plan;
- (c) The County shall pay seventy five percent (75%) of the total premium for each Union member participating in the Dental option of the Strathcona County Employee Benefit Plan;
- (d) The County shall pay seventy five percent (75%) of the total premium for each eligible Union member enrolled in the Alberta Health Care Insurance Plan through the Strathcona County.

The County's liability shall be limited to the payment of premiums, where required, for the various plans and the County shall be relieved of any further liability with respect to the plans.

The benefit plan shall include a Flexible Health Spending Account (FHSA) of five hundred dollars (\$500) per year which can be carried forward for one additional year.

Effective Jan. 1, 2021, the FHSA shall be increased to one thousand dollars (\$1,000).

Retroactive FHSA amounts for 2021 and 2022 shall be accessible effective Jan. 1, 2023. In advance of Jan. 1, 2023, Employees shall have the option to divide the total amount between the Health Spending Account (non-taxable) and the Wellness Spending Account (taxable) in the proportion of their choosing.

25.02 In the event that the County should undertake to transfer group insurance and benefit plans from one carrier to another, it is agreed that the Union will be consulted prior to the negotiation of such coverage.

## **Benefits for Temporary Employees**

### 25.03 <u>Benefits for Temporary Employees</u>

To be eligible to participate in the Strathcona County Employee Benefit plans, Temporary Employees must meet the following conditions:

- (a) All Temporary Employees who have been employed for twelve (12) consecutive cumulative months or more, provided there has been no break in service for longer than six (6) months, shall receive the same benefits as Regular Employees, with the exception of pension and the LTD benefit entitlement will be limited to a maximum of twenty-four months.
  - For the purposes of calculating the twelve (12) cumulative months, leaves of absence of more than thirty (30) calendar days shall be excluded.
- (b) An exception to the above shall be made for Temporary Employees replacing Regular Employees on long term disability. A Temporary Employee in this situation will be eligible for benefits after six (6) consecutive months of employment.

#### Pension Plan

## 25.04 Pension Plan

The County participates in the Local Authorities Pension Plan. Participation in this plan is compulsory for "Regular Employees" after they have completed one (1) year of service or immediately if transferring from another Local Authority Board Employer.

25.05 The County shall make payroll deductions as follows:

- (a) The County shall deduct legislated Revenue Canada deductions and Local Authorities Pension Plan contributions as required from each bi-weekly pay cheque;
- (b) The current authorized deductions for Canada Savings Bonds will be taken each pay cheque;
- (c) The Employee contribution to the Strathcona County Employee Benefit Plan will be deducted from the first bi-weekly pay cheque. each month.

### **ARTICLE 29 - TEMPORARY EMPLOYEES**

- 29.01 Temporary Employees shall be entitled to the terms and conditions of this Agreement, except where Regular Employees are referenced. Seniority for employment periods of less than six (6) months shall not be accumulated.
- 29.02 A Temporary Employee shall be considered probationary as per Article 11.01 and shall not serve probation for subsequent years of employment unless a break of service of greater than one (1) year has occurred or the Employee is returning to a different position. Temporary Employees shall have access to the application of the grievance procedure except in connection with termination of employment as a result of either:
  - (a) Unsatisfactory performance, or;
  - (b) The abolishment of the position occupied by the Temporary Employee.
- 29.03 For the purpose of determination of Pay Grade and Step a Temporary Employee returning for subsequent employment within one (1) year may be required to undertake the appropriate probationary period unless he is returning to a position in which he has previously fulfilled the probationary requirements.
- 29.04 When Temporary Employees are recalled, it shall be based on:
  - (a) A satisfactory performance evaluation from the previous year of employment;

- (b) The Employee having qualifications to do the job for which he is recalled;
- (c) The Employee's accumulated service.

The opportunity for recall shall exist for six (6) months within the Branch.

This clause will not apply to Employees participating in the NAIT or other post-secondary Water and Wastewater Training Program.

### ARTICLE 30 - GENERAL WAGE ADMINISTRATION

30.01 The rates of pay for all Employee members of the Union excluding Apprentices, covered by the Agreement shall be as specified in the attached Schedule A.

Temporary Employees returning for subsequent employment within one (1) year shall have their prior service recognized for the purposes of determining Pay Grade and Step.

- 30.02 An Employee is entitled to the rates indicated in Schedule A in the following way:
  - (a) To Step One (1) upon hiring;
  - (b) To Step Two (2) upon completion of the five hundred (500) hours of work in the classification and recommendation of the appropriate supervisor;
  - (c) To Step Three (3) upon completion of an additional fifteen hundred (1,500) hours of work in the classification and the recommendation of the appropriate supervisor
  - (d) Water & Wastewater Operators step progression is as stated in the Water and Wastewater Job Summary attached to the Collective Agreement. Water and Wastewater Operators step progression shall be:
    - (i) To Step One (1) upon hiring;
    - (ii) To Step Two (2) upon completion of five hundred (500) hours of work in the classification, additional qualifications (in accordance with the job description), and recommendation of the appropriate supervisor;
    - (iii) To Step Three (3) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
    - (iv) To Step Four (4) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
    - (v) To Step Five (5) upon completion of an additional one thousand eight

hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;

(vi) To Step Six (6) upon completion of an additional one thousand eight hundred and sixty-four (1,864) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;

## (e) Construction Operator step progression shall be:

- (i) To Step One (1) upon hiring;
- (ii) To Step Two (2) upon completion of two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (iii) To Step Three (3) upon completion of an additional two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the job description), and the recommendation of the appropriate supervisor;
- (iv) To Step Four (4) upon completion of an additional two thousand (2,000) hours of work in the classification, additional qualifications (in accordance with the description), and the recommendation of the appropriate supervisor.

## (f) Water and Wastewater Leadhand step progression:

Water & Wastewater Operator,	Water & Wastewater Leadhand,
Steps 1 to 4	Step 1
Water & Wastewater Operator, Step 5	Water & Wastewater Leadhand, Step 2
Water & Wastewater Operator, Step	Water & Wastewater Leadhand,
6	Step 3

## (g) Water and Wastewater Infrastructure Repair Technician step progression:

Water & Wastewater Steps 1 to 4	Operator,	W&WW Infrastructure Repair Technician, Step 1				
Water & Wastewater Op 5	erator, Step	W&WW Techniciar	Infrastructure n, Step 2	Repair		

Wa	ter & Wastewater Operator, Step	1		Repair
6		Technician	, Step 3	

- (h) An appropriate form will be completed and placed In the Employee's personnel file when the wage rate is amended.
- 30.03 Should an Employee not be recommended by the appropriate supervisor for advancement as per Clause 30.02 b), c), d) or e), the Administration shall advise that Employee of the reasons in writing, and in addition:
  - (a) What action might be taken in order for the wage to be advanced.
  - (b) In the event that the wage will not be advanced, what alternate action the administration intended to take.
- 30.04 An Employee affected by Article 30.03 may file a grievance for a failure to be recommended for an increase.
- 30.05 The rate of pay established in Schedule A shall apply during the term of this Agreement. Employees shall be paid bi-weekly. On each pay day, each Employee shall be provided with an itemized statement of his wages, overtime and other earnings and deductions.

## 30.06 Flag Person

The Employer may hire Temporary Employees as Flagperson at a rate in accordance with Schedule A. Should such employees be assigned to any other work they shall be paid in accordance with Clause 20.01. No one shall be regularly assigned duties as Flagperson until they have successfully completed an approved Flagperson's training course. Employees taking the Flagperson training shall be paid the appropriate rate of pay while training.

### ARTICLE X - DIRECT RESPONSILBE PAY (DRP)

- X.01 When a Water and Wastewater Operator or Water and Wastewater Leadhand is assigned to the Utilities On-Call Schedule as On-Call #1, On-Call #2 or Transmission Operator (for emergent calls) and deemed to be in direct responsible charge (DRC), the Employee shall be paid an additional \$2.25 per hour on the base rate of pay, for all hours worked between sixteen hundred and thirty (1630) hours and zero eight hundred (0800) hours Monday to Friday and for all hours worked on Saturday and Sunday and Holidays.
  - (a) At no time shall Operators called out to assist the On-Call #1, On-Call #2 or Transmission Operator with emergency service or other similar callouts

## receive DRP.

- X.02 Where applicable, an Employee shall be eligible to receive DRP in addition to overtime, shift and weekend premiums.
- X.03 At no time shall DRP be included with the Employee's regular rate of pay for the purpose of calculating overtime payments, other premium payments, or any Employee benefits.

## Wage Increases

```
January 1, 2020 – 1%

<del>January 1, 2021 – 1%</del>

January 1, 2022 – 1%

January 1, 2023 – 1.5%
```

The following is included for clarity, but will not form part of the collective agreement:

- Retroactive increases are applicable on all hours worked and will be payable by April 12, 2023
- The new compensation structure and pay grid will be effective January 1, 2023.
- Temporary Employees will be eligible for benefits December 12, 2022 as long as ratification occurs by November 1, 2022.
- Shift Premiums are not retroactive and will be effective December 12, 2022 as long as ratification occurs by November 1, 2022.
- Unless otherwise specified, all changes are effective on the date of ratification.
- Premiums referenced in Article 17 will be included in pensionable earnings.

The following articles have no amendments and will remain as current, except for any consequential amendments that may arise as a result of other agreed to amendments:

Article 5 - Management Rights

**Article 7 – Time off for Union Business** 

Article 13 - Discipline

Article 18 – Call out & Standby

## Schedule A

Code	Position		Hour	y Rates	
		2020 - 0.5% increase	\$16.89	\$17.51	
3001	Flag Person	2022 - 1.0% increase			
		2023 - removed			
		2020 - 0.5% increase	\$20.88	\$21.25	
3002	Labourer	2022 - 1.0% increase			
		2023 - removed			
		2020 - 0.5% increase	\$23.53	\$24.30	
3003	Labourer II	2022 - 1.0% increase			
		2023 - removed			
		2020 - 0.5% increase	\$25.90	\$26.58	\$27.03
3004	Labourer III	2022 - 1.0% increase			
		2023 - removed			
TBD	PW Labourer	2023 - added	\$21.78	\$24.91	\$27.71
		2020 - 0.5% increase	\$24.97	\$25.53	
3005	Pickup/Courier Driver	2022 - 1.0% increase	\$25.22	\$25.79	
		2023 - 1.5% increase	\$25.60	\$26.18	
		2020 - 0.5% increase	\$24.64	\$25.36	
3006	Survey Aide I	2022 - 1.0% increase	-		
		2023 - 1.5% increase			
		2020 - 0.5% increase			\$27.88
3007	Survey Aide II	2022 - 1.0% increase	\$26.83	\$27.46	\$28.16
		2023 - 1.5% increase	\$27.23	\$27.87	\$28.58
		2020 - 0.5% increase	\$31.87	\$32.62	\$33.41
3008	PWTech1	2022 - 1.0% increase	\$32.19	\$32.95	\$33.74
		2023 - 1.5% increase	\$32.67	\$33.44	\$34.25
		2020 - 0.5% increase	\$36.78	\$37.27	\$37.65
3009	PW Tech II	2022 - 1.0% increase	\$37.15	\$37.64	\$38.03
		2023 - 1.5% increase			
			•		<del></del>
		2020 - 0.5% increase	\$29.25	\$29.91	\$30.38
3032	Gravel Checker	2022 - 1.0% increase	\$29.54	\$30.21	\$30.68
		2023 - 1.5% increase	\$29.98	\$30.66	\$31.14
		2020 - 0.5% increase	\$30.69	\$31.36	\$32.08
3010	Concrete Worker I	2022 - 1.0% increase	\$31.00	\$31.67	\$32.40
		2023 - 1.5% increase	\$31.47	\$32.15	\$32.89
		2020 - 0.5% increase	\$32.21	\$33.11	\$34.02
3011	Concrete Worker II	2022 - 1.0% increase	\$32.53	\$33.44	\$34.36
		2023 - 1.5% increase		\$33.94	
			<b>Y</b>	<b></b>	generalise productions are greatered an excellent products greatered to the extended only discount to consider

Code	Position		Hour	y Rates				
		2020 - 0.5% increase	\$28.09	\$28.69	\$29.29			
3012	Truck Driver I	2022 - 1.0% increase	\$28.37	\$28.98	\$29.58			
		2023 - removed						
	В примерения и примерения в примерения и примерения примерения и примерения и примерения и примерения и примерен	2020 - 0.5% increase	\$29.75	\$30.48	\$31.19		*	
3013	Truck Driver II	2022 - 1.0% increase	\$30.05	\$30.78	\$31.50			
		2023 - removed						
		2020 - 0.5% increase	\$32.48	\$33.13	\$33.93			
3014	Truck Driver III	2022 - 1.0% increase	\$32.80	\$33.46	\$34.27			
		2023 - removed						
TBD	Truck Driver - Class 1	2023 - added	\$33.29	\$33.96	\$34.78			
	<u> </u>	2020 - 0.5% increase	\$28.09	\$28.81	\$29.23		***************************************	
3015	Operator I	2022 - 1.0% increase		\$29.10				
		2023 - removed						
		2020 - 0.5% increase	\$31.63	\$32.36	\$33.00			
3016	Operator II	2022 - 1.0% increase	\$31.95	\$32.68	\$33.33			
		2023 - removed						
		2020 - 0.5% increase	\$33.69	\$34.54	\$35.43			
3017	Operator III	2022 - 1.0% increase	\$34.03	\$34.89	\$35.78			
		2023 - removed			9			
TBD	Equipment Operator I	2023-added \$	29.9	8 :	\$30.6	6	\$31.	14
TBD	Equipment Operator II	2023 - added	\$32.43	\$33.17	\$33.83			
TBD	Equipment Operator III	2023 - added	\$34.54	\$35.41	\$36.32			
				_				
		2020 - 0.5% increase	\$33.43	\$34.29	\$35.18			
3031	Sign Person	2022 - 1.0% increase	\$33.76	\$34.63	\$35.53			
		2023 - 1.5% increase	\$34.27	\$35.15	\$36.06			
		2020 - 0.5% increase	\$33.43	\$34.29	\$35.18			
3018	Yardperson	2022 - 1.0% increase	\$33.76	\$34.63	\$35.53			
		2023 - 1.5% increase	\$34.27	\$35.15	\$36.06			
mo	North Yardperson	2023 - added	\$37.87	\$38.78				

Code Position Hourly Rates

-	1 Caracti			y i succes				
		2020 - 0.5% increase	\$36.94	\$37.83				
3036	PW Leadhand	2022 - 1.0% increase	\$37.31	\$38.21				
		2023 - 1.5% increase	\$37.87	\$38.78				
	<u> </u>	2020 - 0.5% increase	\$38.51	\$39.56		***************************************		
3035	PW Project Foreperson	2022 - 1.0% increase	\$38.90	\$39.96				
		2023 - 1.5% increase	\$39.48	\$40.56				
		2020 - 0.5% increase	\$41.69	\$42.57	\$43.52			
3019	Operations Foreperson	2022 - 1.0% increase	\$42.11	\$43.00	\$43.96			
		2023 - 1.5% increase	\$42.74	\$43.65	\$44.62			
		2020 - 0.5% increase	\$20.88	\$24.30	\$27.03	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		
3034	W&WW Labourer	2022 - 1.0% increase	\$21.09	\$24.54	\$27.30			
		2023 - 1.5% increase	\$21.41	\$24.91	\$27.71			
	Wastewater Infrastructure	2020 - 0.5% increase	\$30.42	\$33.00	\$35.43			
3037		2022 - 1.0% increase	\$30.72	\$33.33	\$35.78			
	Operator	2023 - 1.5% increase	\$31.18	\$33.83	\$36.32			
	W&WW Operator (Collection &	2020 - 0.5% increase	\$30.42	\$32.93	\$34.43	\$36.90	\$38.64	\$41.11
3020	Distribution)	2022 - 1.0% increase	\$30.72	\$33.26	\$34.77	\$37.27	\$39.03	\$41.52
	Distribution)	2023 - 1.5% increase	\$31.18	\$33.76	\$35.29	\$37.83	\$39.62	\$42.14
		2020 - 0.5% increase	\$30.42	\$32.93	\$34.43	\$36.90	\$38.64	\$41.11
3049	W&WW Operator (Transmission)	2022 - 1.0% increase	\$30.72	\$33.26	\$34.77	\$37.27	\$39.03	\$41.52
		2023 - 1.5% increase	\$31.18	\$33.76	\$35.29	\$37.83	\$39.62	\$42.14
		2020 - 0.5% increase	\$38.46	\$40.20	\$42.67			
3038	W&WW Leadhand	2022 - 1.0% increase	\$38.84	\$40.60	\$43.10			
		2023 - 1.5% increase	\$39.42	\$41.21	\$43.75			
		2020 - 0.5% increase	\$39.86	\$41.57	\$44.09			
3039	W&WW Project Foreperson	2022 - 1.0% increase	\$40.26	\$41.99	\$44.53			
		2023 - removed						
TBD	W&WW Infrastructure Repair Technician	2023 - added	\$40.86	\$42.62	\$45.20			
		2020 - 0.5% increase	\$42.63	\$44.30	\$45.84			
3021	W&WW Foreperson	2022 - 1.0% increase	\$43.06	\$44.74	\$46.30			
		2023 - 1.5% increase	\$43.71	\$45.41	\$46.99			
		2020 - 0.5% increase	\$44.33	\$46.01	\$47.52			
3022	W&WW Electrician	2022 - 1.0% increase	\$44.77	\$46.47	\$48.00			
		2023 - 1.5% increase	\$45.44	\$47.17	\$48.72			
	I Miliaine Englishes & Anistonness	2020 - 0.5% increase	\$52.73	\$54.50	\$56.37			***************************************
3033	Utilities Facilities Maintenance Technician	2022 - 1.0% increase	\$53.26	\$55.05	\$56.93			
	I CU ( IIUdi )	2023 - 1.5% increase	\$54.06	\$55.88	\$57.78			
		2020-0.5% increase	\$33.23	\$33.69	\$34.14			
3023	Vehide Serviceperson I	2022 - 1.0% increase	\$33.56	\$34.03	\$34.48			
		2023 - 1.5% increase	\$34.06	\$34.54	\$35.00			
***************************************				•		,		,

Code	Position		Hourty i	Rates			
	·	2020 - 0.5% increase		\$37.10	\$37.80		•
3024	Vehicle Serviceperson II	2022 - 1.0% increase					
3024	TO BOX OUT TROUBLESON IN	2023 - 1.5% increase		\$38.03			
		2023-13/01/10/605	VV1.21	<del>400.00</del>	<del></del>	***************************************	Comment American Comment Comment
TBD	Shipper-Receiver, Materials Management	2023 - added	\$21.41	\$24.91	\$27.71		
		2020 - 0.5% increase	\$31.94	\$32.77	\$33.42		
3025	Parts Person I	2022 - 1.0% increase	\$32.26	\$33.10	\$33.75		
		2023 - 1.5% increase	\$32.74	\$33.60	\$34.26		
		2020 - 0.5% increase	\$34.31	\$35.26	\$36.13		
3026	Parts Person II	2022 - 1.0% increase	\$34.65	\$35.61	\$36.49		
		2023 - 1.5% increase	\$35.17	\$36.14	\$37.04		
		2020 - 0.5% increase	\$42.25	\$43.80	\$45.45	<b>К</b> ориция уканувания на отключения от Монеция от поставления	
3027	Welder	2022 - 1.0% increase	\$42,67	\$44.24	\$45.90		
		2023 - 1.5% increase	\$43.31	\$44.90	\$46.59		
		2020 - 0.5% increase	\$30.90	\$31.75	\$32.60		
3028	Small Engine Mechanic	2022 - 1.0% increase	\$31.21	\$32.07	\$32.93		
		2023 - 1.5% increase	\$31.68	\$32.55	\$33.42		
		2020 - 0.5% increase	\$47.26	\$48.66			
3040	V&E Leadhand	2022 - 1.0% increase	\$47.73	\$49.15			
-		2023 - 1.5% increase	\$48.45	\$49.89			
		2020 - 0.5% increase	\$46.14	\$46.57	\$48.66		
3030	V&E Chargehand	2022 - 1.0% increase	\$46.60	\$47.04	\$49.15		
		2023 - 1.5% increase	\$47.30	\$47.75	\$49.89		
		2020 - 0.5% increase	\$47.26	\$48.66			
3042	Transit Leadhand	2022 - 1.0% increase	\$47.73	\$49.15			
		2023 - 1.5% increase	\$48.45	\$49.89			
		2020 - 0.5% increase	\$42.25	\$43.80	\$45.45		
3062	Reet Services Technician	2022 - 1.0% increase	\$42.67	\$44.24	\$45.90		
		2023 - 1.5% increase	\$43.31	\$44.90	\$46.59		
**************************************							_
		2020-0.5% increase	\$28.09	\$28.81	\$29.23	<del></del>	
3043	Utility Operator I	2022 - 1.0% increase	\$28.37	\$29.10	\$29.52		
		2023 - removed					
		2020 - 0.5% increase	\$31.63	\$32.36	\$33.00		
3044	Utility Operator II	2022 - 1.0% increase	\$31.95	\$32.68	\$33.33		
		2023 - removed					
		2020 - 0.5% increase	\$33.79	\$35.90	\$38.00		
3045	Utility Operator III	2022 - 1.0% increase	\$34.13	\$36.26	\$38.38		
		2023 - removed					
TBD	Construction Operator	2023 - added	\$30.02	\$33.17	\$36.80	\$38.96	

Code	Position		Hourty R	ates		
		2020 - 0.5% increase	\$37.32	\$39.45	\$41.63	
3046	Utility Leadhand	2022 - 1.0% increase	\$37.69	\$39.84	\$42.05	
		2023 - 1.5% increase	\$38.26	\$40.44	\$42.68	
		2020 - 0.5% increase	\$41.91	\$43.53	\$45.12	
3047 Infrastr	Infrastructure Foreperson	2022 - 1.0% increase	\$42.33	\$43.97	\$45.57	
		2023 - 1.5% increase	\$42.96	\$44.63	\$46.25	
		2020 - 0.5% increase	\$20.88	\$24.30	\$27.03	
3048	Utility Labourer	2022 - 1.0% increase	\$21.09	\$24.54	\$27.30	
		2023 - 1.5% increase	\$21.41	\$24.91	\$27.71	

es Instrument nician/Electrician Dual es Electrician	2020 - 0.5% increase 2022 - 1.0% increase 2023 - 1.5% increase 2020 - 0.5% increase 2022 - 1.0% increase 2023 - 1.5% increase 2020 - 0.5% increase 2022 - 1.0% increase	\$47.00 \$47.71 \$44.33	\$48.29 \$48.77 \$49.50 \$46.01	\$33.00 \$33.33 \$33.83 \$49.91 \$50.41 \$51.17
es Instrument nician/Electrician Dual	2023 - 1.5% increase 2020 - 0.5% increase 2022 - 1.0% increase 2023 - 1.5% increase 2020 - 0.5% increase	\$47.00 \$47.71 \$44.33	\$48.77 \$49.50	\$33.83 \$49.91 \$50.41 \$51.17
nician/Electrician Dual	2020 - 0.5% increase 2022 - 1.0% increase 2023 - 1.5% increase 2020 - 0.5% increase	\$47.00 \$47.71 \$44.33	\$48.77 \$49.50	\$49.91 \$50.41 \$51.17
nician/Electrician Dual	2022 - 1.0% increase 2023 - 1.5% increase 2020 - 0.5% increase	\$47.00 \$47.71 \$44.33	\$48.77 \$49.50	\$50.41 \$51.17
nician/Electrician Dual	2023 - 1.5% increase 2020 - 0.5% increase	\$47.71 \$44.33	\$49.50	\$51.17
	2020 - 0.5% increase	\$44.33		
es Electrician			\$46.01	\$47.52
es Electrician	2022 - 1 096 increases	*		VTI with
	TOTE - TOWN HIT COSC	\$44.77	\$46.47	\$48.00
	2023 - 1.5% increase	\$45.44	\$47.17	\$48.72
	2020 - 0.5% increase	\$44.33	\$46.01	\$47.52
es Instrument Technician	2022 - 1.0% increase	\$44.77	\$46.47	\$48.00
	2023 - 1.5% increase	\$45.44	\$47.17	\$48.72
	2020 - 0.5% increase	\$44.33	\$46.01	\$47.52
ol System Technician	2022 - 1.0% increase	\$44.77	\$46.47	\$48.00
	2023 - 1.5% increase	\$45.44	\$47.17	\$48.72
		2023 - 1.5% increase 2020 - 0.5% increase of System Technician 2022 - 1.0% increase	2022 - 1.0% increase \$44.77 2023 - 1.5% increase \$45.44 2020 - 0.5% increase \$44.33 ol System Technician 2022 - 1.0% increase \$44.77	2022 - 1.0% increase \$44.77 \$46.47 \$45.44 \$47.17 \$2020 - 0.5% increase \$44.33 \$46.01 \$45.44 \$47.77 \$46.47 \$46.47 \$46.47 \$46.47

	Anneantics A. en Markanic	2020 - 0.5% increase	\$24.99
3101	Apprentice Auto Mechanic,	2022 - 1.0% increase	\$25.24
	1st year (55%)	2023 - 1.5% increase	\$25.62
	America A. en Markania	2020 - 0.5% increase	\$31.82
3102	Apprentice Auto Mechanic,	2022 - 1.0% increase	\$32.14
	2nd year (70%)	2023 - 1.5% increase	\$32.62
	Assessment Assessment	2020 - 0.5% increase	\$36.35
3103	Apprentice Auto Mechanic,	2022 - 1.0% increase	\$36.71
	3rd year (80%)	2023 - 1.5% increase	\$37.26
	America A. to Markania	2020 - 0.5% increase	\$40.89
3104	Apprentice Auto Mechanic, 4th year (90%)	2022 - 1.0% increase	\$41.30
	Hulycai (50/9)	2023 - 1.5% increase	\$41.92

Code	Position		Hourly F	lates		
	Apprentice Heavy Duty Mechanic,	2020 - 0.5% increase	\$27.27			
3111	1st year (60%)	2022 - 1.0% increase	\$27.54			
	Bryca (w/a)	2023 - 1.5% increase	\$27.95			
	Apprentice Heavy Duty Mechanic,	2020 - 0.5% increase	\$31.82			
3112	2nd year (70%)	2022 - 1.0% increase	\$32.14			
	Zici yeai (7076)	2023 - 1.5% increase	\$32.62			
	Apprentice Heavy Duty Mechanic,	2020 - 0.5% increase	\$36.35			
3113	• • • • • • • • • • • • • • • • • • • •	2022 - 1.0% increase	\$36.71			
	3rd year (80%)	2023 - 1.5% increase	\$37.26			
	Apprentice Heavy Duty Mechanic,	2020 - 0.5% increase	\$40.89			
3114	4th year (90%)	2022 - 1.0% increase	\$41.30			
	401 year (5076)	2023 - 1.5% increase	\$41.92			
3129	Certified Technician (HD)	2020 - 0.5% increase 2022 - 1.0% increase		\$43.80 \$44.24		
		2023 - 1.5% increase	\$43.31	\$44.90	\$46.59	
		2023 - 1.5% increase		\$44.90	\$46.59	
	Apprentice Transit Mechanic,	2020 - 0.5% increase	\$2727	\$44.90	\$46.59	
3131	Apprentice Transit Mechanic, 1st year (60%)	2020-0.5% increase 2022-1.0% increase	\$27.27 \$27.54	\$44.90	\$46.59	
3131	• •	2020 - 0.5% increase 2022 - 1.0% increase 2023 - 1.5% increase	\$27.27 \$27.54 \$27.95	\$44.90	\$46.59	
	• •	2020 - 0.5% increase 2022 - 1.0% increase 2023 - 1.5% increase 2020 - 0.5% increase	\$27.27 \$27.54 \$27.95 \$31.82	\$44.90	\$46.59	
3131	1st year (60%)	2020 - 0.5% increase 2022 - 1.0% increase 2023 - 1.5% increase 2020 - 0.5% increase 2022 - 1.0% increase	\$2727 \$2754 \$2795 \$3182 \$32.14	\$44.90	\$46.59	
	1st year (60%)  Apprentice Transit Mechanic,	2020 - 0.5% increase 2022 - 1.0% increase 2023 - 1.5% increase 2020 - 0.5% increase 2022 - 1.0% increase 2023 - 1.5% increase	\$2727 \$2754 \$2795 \$31.82 \$32.14 \$32.62	\$44.90	\$46.59	
3132	1st year (60%)  Apprentice Transit Mechanic,	2020 - 0.5% increase 2022 - 1.0% increase 2023 - 1.5% increase 2020 - 0.5% increase 2022 - 1.0% increase 2023 - 1.5% increase 2020 - 0.5% increase	\$27.27 \$27.54 \$27.95 \$31.82 \$32.14 \$32.62 \$36.35	\$44.90	\$46.59	
	1st year (60%)  Apprentice Transit Mechanic, 2nd year (70%)	2020-0.5% increase 2022-1.0% increase 2023-1.5% increase 2020-0.5% increase 2022-1.0% increase 2023-1.5% increase 2020-0.5% increase 2022-1.0% increase	\$27.27 \$27.54 \$27.95 \$31.82 \$32.14 \$32.62 \$36.35 \$36.71	\$44.90	\$46.59	
3132	1st year (60%)  Apprentice Transit Mechanic, 2nd year (70%)  Apprentice Transit Mechanic,	2020-0.5% increase 2022-1.0% increase 2023-1.5% increase 2020-0.5% increase 2022-1.0% increase 2023-1.5% increase 2022-1.0% increase 2022-1.0% increase 2023-1.5% increase	\$27.27 \$27.54 \$27.95 \$31.82 \$32.14 \$32.62 \$36.35 \$36.71 \$37.26	\$44.90	\$46.59	
3132	1st year (60%)  Apprentice Transit Mechanic, 2nd year (70%)  Apprentice Transit Mechanic,	2020-0.5% increase 2022-1.0% increase 2023-1.5% increase 2020-0.5% increase 2022-1.0% increase 2023-1.5% increase 2020-0.5% increase 2022-1.0% increase	\$27.27 \$27.54 \$27.95 \$31.82 \$32.14 \$32.62 \$36.35 \$36.71	\$44.90	\$46.59	

# **SCHEDULE B**

# **January 1, 2023**

EQUIPMENT	MINIMUM OF CLASSIFICATION RATE PAID
Non-Finishing Packers (Rubber Roller, Pad Foot Operator, and combination for non- finishing work)	Equipment Operator I – Step 1
Finishing Packer (Smooth Drum Rollers and combination of finishing work)	Equipment Operator I
Skid Steer / Sidewalk Machine	Operator II Equipment Operator I
Single Axle Truck	Truck Driver I Equipment Operator I
Sign Truck (Single Axle with Manlift)	Truck Driver II Equipment Operator I
<del>Tracked Dozer</del>	Equipment Operator II
Wheel Loader	Equipment Operator II
Tandem Axle Truck	Truck Driver II Equipment Operator II
Street Sweepers	Equipment Operator II
Truck Plow	Equipment Operator II
Sander	Truck Driver II Equipment Operator II
RPM Snow Blower	Equipment Operator II
Road Reclaimer	Equipment Operator II
PB Patcher Patch Truck	Equipment Operator II
Rubber Tire Backhoe	Equipment Operator II
Graders	Operator II Equipment Operator III
Track Hoe	Equipment Operator III + \$0.20 / hour
Paver	Equipment Operator III
Tractor-Trailer	Truck Driver ## Class 1
Directional Driller	Utility Operator III

## **Letter of Understanding**

### Between

## Strathcona County

and

The Alberta Union of Provincial Employees, Local 118, Chapter 009

Subject: Skid Steer/Sidewalk Machine Rate of Pay

- Employees who, at the date of ratification, have a class 3 license, will be paid the Equipment
  Operator II rate of pay, for hours worked on the Skid Steer/Sidewalk Machine (the "Old Rate").
- 2. Eligible employees will only be paid the Old Rate for a period of two years following the date of ratification. At the expiry of the two (2) years, the employees will then be paid in accordance with Schedule B.
- 3. Employees employed at the time of ratification, who have not yet obtained their class 3 licence, have until April 1, 2023 to obtain their class 3 licence. Employees who obtain their class 3 license by April 1, 2023, will be paid in accordance with paragraphs 1 & 2 above.
- 1. Employees employed in the Labourer I, II or III or Operator I classifications who have experience operating the Skid Steer/Sidewalk Machine who have a class 3 license at the date of ratification will be employed in the Operator II/Equipment Operator II classification and will be expected to operate all equipment associated with that classification as listed in Schedule B, subject to the provision of hands-on training.
- 2. Employees who have not yet worked a minimum of eighty (80) hours on an equipment type listed in Schedule B, shall be provided with hands-on training provided by an Employee who has completed at least one thousand five hundred (1,500) hours worked on such equipment, with a minimum of one hundred (100) hours of which has been worked within the last twelve (12) months. Such training shall not be less than six (6) twelve- (12) hour shifts, or ten (10) eight- (8) hour shifts. Requests for additional training shall not be unreasonably denied.
- 3. Employees will be eligible for refresher training of three (3) shifts if they have not used a piece of equipment within the last twelve (12) months or if a newer/different model of the same type of equipment is in use.
- 4. Employees employed in the Labourer I, II or III or Operator I classifications who have experience operating the Skid Steer/Sidewalk Machine who who do not have a class 3 license, at the date of ratification shall be reimbursed for the cost of such training upon completion and shall be provided with up to twenty-four (24) hours paid time off for the purposes of attending such training. Eligible Employees who obtain a class 3 license by December 31, 2023, will be employed in the Equipment Operator II classification

5.	Employees who, as of the date of ratification, are employed in the Labourer I, II or III or Operator I classifications who have experience operating the Skid Steer/Sidewalk Machine but do not have a class 3 license will continue to be paid at the Operator II/Equipment operator II rates of pay for hours worked on the Skid Steer/Sidewalk Machine until December 31, 2024.